Published for SISSA by 🖄 Springer

RECEIVED: May 18, 2021 REVISED: July 27, 2021 ACCEPTED: August 21, 2021 PUBLISHED: September 28, 2021

The EFT stringy viewpoint on large distances

Stefano Lanza,^a Fernando Marchesano,^b Luca Martucci^c and Irene Valenzuela^d

ABSTRACT: We observe a direct relation between the existence of fundamental axionic strings, dubbed EFT strings, and infinite distance limits in 4d $\mathcal{N} = 1$ EFTs coupled to gravity. The backreaction of EFT strings can be interpreted as RG flow of their couplings, and allows one to probe different regimes within the field space of the theory. We propose that any 4d EFT infinite distance limit can be realised as an EFT string flow. We show that along such limits the EFT string becomes asymptotically tensionless, and so the EFT eventually breaks down. This provides an upper bound for the maximal field range of an EFT with a finite cut-off, and reproduces the Swampland Distance Conjecture from a bottom-up perspective. Even if there are typically other towers of particles becoming light, we propose that the mass of the leading tower scales as $m^2 \sim \mathcal{T}^w$ in Planck units, with \mathcal{T} the EFT string tension and w a positive integer. Our results hold even in the presence of a non-trivial potential, as long as its energy scale remains well below the cut-off. We check both proposals for large classes of 4d $\mathcal{N} = 1$ string compactifications, finding that only the values w = 1, 2, 3 are realised.

KEYWORDS: D-branes, Flux compactifications, String theory and cosmic strings, Supergravity Models

ARXIV EPRINT: 2104.05726



^aInstitute for Theoretical Physics, Utrecht University, Princetonplein 5, 3584 CE Utrecht, The Netherlands

^bInstituto de Física Teórica UAM-CSIC,

Cantoblanco, 28049 Madrid, Spain

^cDipartimento di Fisica e Astronomia "Galileo Galilei", Università degli Studi di Padova,

[&]amp; I.N.F.N. Sezione di Padova, Via F. Marzolo 8, 35131 Padova, Italy

^d Jefferson Physical Laboratory, Harvard University,

Cambridge, MA 02138, U.S.A. E-mail: s.lanza@uu.nl, fernando.marchesano@csic.es, luca.martucci@pd.infn.it, ivalenzuela@g.harvard.edu

Contents

1	Introduction	1
2	Fundamental axionic strings in 4d $\mathcal{N} = 1$ EFTs 2.1 Axionic strings in $\mathcal{N} = 1$ EFTs 2.2 Dual formulation 2.3 Energy-momentum tensor and tension	4 4 7 8
	2.4 EFT strings and validity of the solution	9
3	 Strings and infinite field distance limits 3.1 Physics along the string solution 3.2 RG flow interpretation 3.3 String flows as paths in field space 3.4 EFT breakdown 	13 13 15 18 20
4	 Strings, instantons and asymptotic limits 4.1 Strings, instantons and cones 4.2 A simple example 4.3 Connection with the 5d viewpoint 4.4 Finiteness and conical structures 	22 22 28 33 33
5	 4d strings and swampland criteria 5.1 The distant axionic string conjecture 5.2 Some implications 	35 35 37
6	 String theory evidence 6.1 Heterotic on Calabi-Yau three-folds 6.2 Type I on Calabi-Yau three-folds 6.3 Type IIB/F-theory compactifications 6.4 Type IIA Calabi-Yau orientifolds 6.5 M-theory on G₂ spaces 	44 45 50 51 58 66
7	Conclusions	72
Α	Terminology A.1 Glossary of strings and their flows A.2 Algebraic geometry	76 76 77
в	Gravitational contributions to the energy density	79
С	Geodesic paths in field space and string flows	80
D	String RG flows in Field Theory	82

EEFT instantons87FEFT string flows in toroidal orbifolds89F.1Heterotic/type I complex structure flows in toroidal orbifolds89F.2F-theory Kähler flows on the resolved base $T^6/(\mathbb{Z}_2)^3$ 91F.3M-theory flows on the toroidal orbifold $T^7/(\mathbb{Z}_2)^3$ 95

1 Introduction

One pressing problem for Effective Field Theories (EFTs) of quantum gravity is to determine which kind of physics one can obtain at regions of weak coupling. Since these regimes appear along large field distance limits, one may be led to assume that the size of the EFT field space can be arbitrarily large, which is very suggestive for, e.g., embedding macroscopic models of large field inflation in a UV complete theory of quantum gravity. Concrete string theory realisations however show that asymptotic limits of infinite distance come with a mechanism that prevents a 4d EFT description beyond a certain point, as predicted by the Swampland Distance Conjecture (SDC) [1]. As a consequence of this conjecture, quantum gravity would impose a maximal size to any EFT field space. Determining what this size is and what is the precise mechanism that triggers the EFT breakdown is a central part of the Swampland Program [2–5], and substantial progress has been achieved through different studies and refinements of the SDC [6–24].

The standard strategy to test and improve our insight on the SDC is to consider a particular string theory compactification, and then attempt to classify and understand the physics of each of its infinite distance limits. While this has been a very successful approach, a more intrinsic EFT viewpoint on these asymptotic large distance limits and their associated phenomena is still lacking. Such a description would undoubtedly put the SDC and all the swampland conjectures connected to it on a firmer ground, even beyond the string theory realm. A promising avenue to achieve this goal is to consider large field variations induced by the presence of localised objects in our EFT, like black holes or bubbles of nothing, as done in [17, 25–31]. In this spirit, it was pointed out in [32] that for 4d EFTs most of the swampland conjectures are connected to the physics of low-codimension objects, namely strings and membranes. In particular, certain $\frac{1}{2}$ BPS strings are directly related to the presence of infinite distance limits in 4d $\mathcal{N} = 1$ EFTs.

A connection between strings and large field distances is already hinted by the 4d cosmic string solutions of [33]. Indeed, in this work the authors construct codimensiontwo profiles of an axio-dilaton τ , whose simplest solution asymptotes towards the infinite field distance limit $\tau \to i\infty$ at spatial infinity. It was later on realised in [34] that the point at which $\tau = i\infty$ can be interpreted as the location of a D7-brane. From this Ftheory perspective one may construct 4d string solutions by compactifying type IIB on a six-manifold X, wrapping several (p,q) 7-branes on X and placing them parallel to each other in 4d. When approaching a D7-brane location the axio-dilaton profile will draw an infinite distance path $\tau \to i\infty$ in field space. From the 4d EFT viewpoint, such profile is sourced by the presence of a localised operator, which is how D7-branes are treated in type IIB supergravity. Moreover, the solution around the D7-brane describes a monodromy $\operatorname{Re} \tau \to \operatorname{Re} \tau + 1$, which is promoted to an axionic shift symmetry as we approach the D7-brane core. In other words, the source of the profile $\tau \to i\infty$ is a 4d fundamental string magnetically coupled to the axion $\operatorname{Re} \tau$.

The purpose of this work is to characterise 4d EFT large field distance limits in terms of the key EFT ingredients of the above example. More precisely, we consider fundamental axionic strings, treated as codimension-two operators of the EFT, and study their backreaction profile in the vicinity of the string core. In the context of $\mathcal{N} = 1$ theories this backreaction describes a trajectory in a field space region with a perturbative axionic symmetry, with a logarithmic profile for the saxionic partner. Such a profile allows to map the physics of the backreacted solution to the physics of vacua up to the distance $1/\Lambda$ from the string core, with Λ the EFT cut-off scale. As we approach the core we asymptotically reach a regime in which those non-perturbative effects that break the axionic symmetry become negligible, pointing to the emergence of a continuous shift symmetry. By the no global symmetry conjecture [35, 36], one expects that for EFTs coupled to quantum gravity such exact shift symmetries are located at infinite distance points in field space, with some mechanism preventing the EFT to ever reach them, similarly to the setup in [37]. Therefore, these solutions are particularly suitable to study the physics of the SDC from an EFT perspective. In fact, the properties of these strings guarantee that the associated solutions can always be treated in an EFT weakly-coupled regime, which is why we refer to them as *EFT strings* throughout the paper.

Indeed, one difference with [33] is that in our case the fields that vary along a $\frac{1}{2}$ BPS string flow will generically be subject to a potential. However, we argue that if the Hubble and mass scales of the potential are low compared with the EFT cut-off scale Λ , so that near this scale it makes sense to talk about a field space, our approach to extract the physics of large field distances still applies. One way to see this is to apply the philosophy of [38, 39] to our context, and interpret the string backreaction as a renormalisation group (RG) flow of the string couplings. In this scheme, the backreaction details closer to the string core determine the string couplings at higher energies, and also the asymptotic large field behaviour of our theory. Therefore, in a sensible setup the interesting part of the EFT string solution will be at wavelengths much shorter than the effect induced by a potential. The RG flow picture also helps to understand the universal behaviour of the EFT string tension \mathcal{T} , which asymptotically vanishes either as we increase Λ or we approach an infinite distance point. This monotonic behaviour provides a rationale for the EFT breakdown that the SDC predicts. Indeed, if an EFT string is a genuinely fundamental object of our 4d EFT, then it cannot be resolved by it for any scale Λ , and so by consistency the EFT semiclassical description must break down before $\mathcal{T} < \Lambda^2$.

Another generalisation with respect to [33] is the arbitrary number of different perturbative axionic shift symmetries that may coexist in given region of a 4d $\mathcal{N} = 1$ EFT. We find that this richness is encoded in a lattice convex cone of EFT string charges, whose interplay with the saxionic field space of this region allows us to classify different kinds of string flows, which ultimately reflect the asymptotic properties of the Kähler potential. We have indeed checked that this structure is realised in a plethora of 4d $\mathcal{N} = 1$ examples built from string compactifications, some of which allow us to connect 4d EFT strings with the definition of *supergravity strings* recently formulated in [40] for 5d theories.

This set of universal properties that stem from the notion of EFT string prompt us to propose the *Distant Axionic String Conjecture*, which essentially claims that all 4d EFT infinite distance limits are realised by the RG flow associated to an EFT string. This proposal was already advanced in [32], and in the present paper is captured by Conjecture 1. Thereafter, we will discuss at some length several implications of this conjecture, like for instance how it constrains the behaviour of the field space curvature at infinity.

Obviously, the most direct consequences involve the SDC. As said before, EFT strings are asymptotically tensionless, and thus any consistent 4d EFT must break down along the infinite distance trajectories that they describe. This observation does not specify the precise breakdown mechanism: it could be via the tower of oscillation modes of the string itself, or via some other tower of states — e.g. Kaluza-Klein (KK) modes — that appear at a lower scale, as observed in the context of 4d $\mathcal{N}=2$ theories in [15]. We analyse a large set of asymptotic limits in different classes of 4d $\mathcal{N} = 1$ string compactifications, finding that in each case it is either a tower of KK-like modes or the oscillation modes of an EFT string that trigger the EFT breakdown, strengthening the Emergent String Conjecture [15]. Furthermore, whenever the leading tower is not given by the oscillation modes of the EFT string, the asymptotic behaviour of the scale m_* of this tower is tied up to the asymptotics of the EFT string tension that describes this limit. We believe this to be a general feature of 4d EFTs compatible with quantum gravity, with the precise relation between m_* and the EFT string tension as captured by Conjecture 2, which states that in Planck units $m_*^2 \sim \mathcal{T}^w$ for some scaling weight $w \in \mathbb{Z}_{>0}$, asymptotically along each EFT string flow. Finally, as already pointed out in [32], these two conjectures together with the Weak Gravity Conjecture (WGC) for EFT strings imply the SDC in 4d theories, with the exponential descent rate of m_* specified by the EFT string extremality factor and the scaling weight. Therefore, if this set of proposals is true, one could essentially estimate the maximal cut-off m_* of a 4d EFT theory simply using EFT data!

The paper is organised as follows. In section 2 we describe fundamental axionic strings and their properties from the viewpoint of 4d EFTs, which lead to the definition of EFT string. In section 3 we argue why EFT string solutions are directly related to trajectories of infinite distance in the EFT field space, giving them an RG flow interpretation. In section 4 we discuss the interplay between $\frac{1}{2}$ BPS strings and non-perturbative corrections, and capture them in terms of continuous and discrete conical structures. Section 5 contains two of the main results of the paper, namely conjectures 1 and 2. They characterise the asymptotic limits of infinite distance of a 4d EFT in terms of EFT string flows, and have interesting implications for several swampland criteria. Section 6 shows how these two conjectures and all the EFT structure developed to arrive to them are realised in several classes of string compactifications, providing abundant examples that support our claims. We finally draw our conclusions in section 7.

Several details have been relegated to the appendices. Appendix A contains a glossary of terms for string flows and for cones in algebraic geometry used throughout the paper.

Appendix B shows that the gravitational contribution to the EFT string energy density does not affect the results of section 2. Appendix C argues that EFT string flows correspond to asymptotically geodesic paths in field space. Appendix D elaborates on the correspondence between EFT string solutions and their RG flows, by computing the latter from a field theory perspective. Appendix E discusses the backreaction and flows associated to 4d $\frac{1}{2}$ BPS instantons. Appendix F examines in detail string flows in different toroidal orbifold compactifications and their resolutions.

2 Fundamental axionic strings in 4d $\mathcal{N} = 1$ EFTs

In this section we describe, from a 4d $\mathcal{N} = 1$ EFT perspective, solutions that correspond to $\frac{1}{2}$ BPS axionic fundamental strings. These strings are characterised by the following properties:

- They are fundamental localised objects, in the sense that they have a singular core that cannot be resolved within a quantum field theory approach. This is to distinguish them from other solitonic strings that can be described within the EFT.
- They are magnetically coupled to axions and enjoy an approximate continuous shift symmetry near their core. With *approximate* we mean that it should be preserved at perturbative level by the Kähler potential and only broken by exponentially suppressed corrections. Hence, the axion behaves as a 0-form gauge field.

This allows one to describe their solution by means of a dual formulation, where the axion is replaced by the 2-form gauge field \mathcal{B}_2 under which the strings are electrically charged. Preserving the $\mathcal{N} = 1$ supersymmetric requires that the chiral multiplets are replaced by linear multiplets. One of the vantage points of working in this dual picture is that we may easily compute the string tension \mathcal{T} , which should satisfy

$$\Lambda^2 < \mathcal{T} < M_{\rm P}^2 \,, \tag{2.1}$$

in order not to imply the EFT breakdown, where Λ is the EFT cut-off scale. Therefore, such strings must be included as localised operators in the theory, as they cannot be resolved within the 4d EFT regime of validity.

2.1 Axionic strings in $\mathcal{N} = 1$ EFTs

Let us consider a 4d $\mathcal{N} = 1$ EFT formulated in terms of chiral multiplets. Axionic string configurations naturally appear when the set of chiral fields can be split as $\{\phi^{\alpha}\} = \{t^i, \chi^{\kappa}\}$, with the superpotential $W \equiv W(\chi)$ only depending on a subset of them, and a Kähler potential K that for now we keep general. Under these circumstances, one may find a moduli space of supersymmetric vacua with a fixed value χ_* for the scalars $\{\chi^{\kappa}\}$ and arbitrary value for the fields t^i . Indeed, since the F-flatness conditions imply

$$D_{ti}W|_{\text{vac}} = (\partial_{ti}K)W|_{\text{vac}} = 0, \qquad (2.2)$$

then the supersymmetry conditions amount to impose $W(\chi_*) = \partial_{\chi^{\kappa}} W(\chi_*) = 0$, leaving the $\{t^i\}$ unconstrained. Moreover, the Cremmer et al. [41] F-term potential $V(t, \chi)$ evaluated at χ_* vanishes at least quadratically on the chiral fields, and so it can be ignored when considering string-like configurations with a varying profile on the $\{t^i\}$. A similar statement holds for no-scale models with a semidefinite positive potential and a moduli space of Minkowski vacua.

Ignoring the presence of V, the relevant terms of the effective action are

$$S = M_{\rm P}^2 \int \left(\frac{1}{2} R * 1 - K_{\alpha\bar{\beta}} \,\mathrm{d}\phi^\alpha \wedge * \mathrm{d}\bar{\phi}^{\bar{\beta}} \right) \,, \tag{2.3}$$

whose string-like solutions can be analysed following the discussion in [33]. For this one splits the 4d coordinates into (t, x, z, \overline{z}) , and imposes 2d Poincaré invariance on (t, x). That is, we allow the varying fields t^i to depend only on z, \overline{z} and choose a 4d metric of the form

$$ds^{2} = -dt^{2} + dx^{2} + e^{2D} dz d\bar{z}, \qquad (2.4)$$

where D depends only on the transverse coordinates (z, \bar{z}) . The equations of motion read

$$K_{\bar{\alpha}\beta}\partial\bar{\partial}\phi^{\beta} + K_{\bar{\alpha}\beta\gamma}\partial\phi^{\beta}\wedge\bar{\partial}\phi^{\gamma} = 0, \qquad (2.5)$$

where $\partial \equiv dz \wedge \partial_z$. The simplest class of solutions corresponds to holomorphic

$$\bar{\partial}\phi^{\alpha} = 0, \qquad (2.6)$$

or anti-holomorphic profiles: $\partial \phi^{\alpha} = 0$. In our setup, the choice (2.6) amounts to set $\chi \equiv \chi_*$ and

$$\bar{\partial}t^i = 0. (2.7)$$

We can then regard $t^i(z)$ as defining a holomorphic map from \mathbb{C} to the moduli space \mathcal{M} . From the Einstein equations it also follows that

$$e^{2D} = |f(z)|^2 e^{-K}, (2.8)$$

with f(z) a holomorphic non-vanishing function [33]. One may also derive condition (2.6) by writing the energy per unit length of a solution in a BPS form:

$$\mathcal{E} = M_{\rm P}^2 \int K_{\alpha\bar{\beta}} \left(\partial \phi^\alpha \wedge *_2 \bar{\partial} \bar{\phi}^{\bar{\beta}} + \bar{\partial} \phi^\alpha \wedge *_2 \partial \bar{\phi}^{\bar{\beta}} \right) = M_{\rm P}^2 \int J_{\mathcal{M}} + 2M_{\rm P}^2 \int K_{\alpha\bar{\beta}} \,\bar{\partial} \phi^\alpha \wedge *_2 \partial \bar{\phi}^{\bar{\beta}} \,, \tag{2.9}$$

with the integral performed over the (z, \bar{z}) -plane, where the Hodge star operator $*_2$ acts, and $J_{\mathcal{M}} \equiv i K_{\alpha\bar{\beta}} d\phi^{\alpha} \wedge d\bar{\phi}^{\bar{\beta}}$. For fixed boundary conditions the first term is topological, while the second is positive definite. We can then obtain the BPS bound

$$\mathcal{E} \ge M_{\rm P}^2 \int J_{\mathcal{M}} \,, \tag{2.10}$$

which is saturated for holomorphic profiles.

Local solutions to (2.7) corresponding to strings located at z = 0 and associated to a monodromy of the form

$$t^i \to t^i + e^i, \qquad e^i \in \mathbb{Z},$$

$$(2.11)$$

are given by

$$t^{i} = t_{0}^{i} + \frac{1}{2\pi i} e^{i} \log\left(\frac{z}{z_{0}}\right) , \qquad (2.12)$$

where t_0^i, z_0 are constant. Writing $t^i = a^i + is^i$ and $z = re^{i\theta}$ we can split this solution as

$$s^{i} = s_{0}^{i} - \frac{1}{2\pi} e^{i} \log\left(\frac{r}{r_{0}}\right),$$
 (2.13a)

$$a^{i} = \frac{\theta}{2\pi} e^{i} + \text{const}, \qquad (2.13b)$$

where, later on, a^i is to be interpreted as an axion and s^i as its saxionic partner. We see that the latter develop a harmonic profile with a logarithmic singularity at the location of the string. More precisely,

$$s^i \to e^i \cdot \infty$$
 for $r \to 0$. (2.14)

This implies that when approaching the strings location certain scalars within $\{t^i\}$ are driven towards the boundary of the moduli space patch that they describe.

Let us for simplicity assume that $e^i \ge 0$, $\forall i$ — the more general case works analogously and will be discussed in detail in the following sections. By considering $s_0^i \gg 1$ and $M_{\rm P}r_0 \gg 1$, the string solution will describe a large saxion profile s^i along a range of rmuch larger than the four-dimensional Planck length, namely $r \le r_0$. Within this region, by appropriately redefining z one may set the non-vanishing holomorphic function in (2.8) to a constant:

$$e^{2D} = e^{K_0 - K(t,\chi_*)}, \qquad (2.15)$$

which we choose as $K_0 \equiv K(t_0, \chi_*)$. In this way the normalisation of the warp factor is fixed as $e^{2D}|_{z_0} = 1$, and only t_0^i and z_0 remain as free adjustable parameters of the solution. Moreover, let us assume that the trajectory (2.14) takes the EFT to a region of field space in which K exhibits an approximate continuous shift symmetries of the form $\operatorname{Re} t^i \to \operatorname{Re} t^i + \operatorname{const}$, so that the $a^i = \operatorname{Re} t^i$ involved in (2.13b) can be considered to be axions in this limit. As a consequence, the solution for the metric will display a radial symmetry, and the solution (2.12) then corresponds to a 4d $\frac{1}{2}$ BPS axionic string.

Finally, the single string solution can be easily generalised to a multi-string solution:

$$t^{i} = t_{0}^{i} + \frac{1}{2\pi i} \sum_{a} e_{a}^{i} \log\left(\frac{z - z_{a}}{z_{0} - z_{a}}\right), \qquad (2.16)$$

where e_a^i are the charges of the string located at z_a in the z-plane, and t_0^i is the value of t^i at z_0 . If $s_0^i \equiv \text{Im} t_0^i$ is large and charges $e^i \geq 0$, then s^i will remain large in the domain $\bigcap_a D_a$, where $D_a = \{|z - z_a| \leq |z_0 - z_a|\}$, which is non-vanishing if the strings are sufficiently close to each other. This shows that they are in equilibrium and do not exert any force among them, as expected for mutually $\frac{1}{2}$ BPS objects.

2.2 Dual formulation

The above description of $\frac{1}{2}$ BPS axionic strings involves a Kähler potential which, near their core, is invariant under axionic shifts $a^i \to a^i + c^i$, and then depends on the chiral field t^i only through its saxionic component $s^i = \text{Im} t^i$. In this case, the local string solution admits a dual description in terms of *dual saxions* ℓ_i and two-form potentials \mathcal{B}_{2i} defined as

$$\ell_i = -\frac{1}{2} \frac{\partial K}{\partial s^i}, \qquad \mathcal{H}_{3i} = \mathrm{d}\mathcal{B}_{2i} = -M_\mathrm{P}^2 \mathcal{G}_{ij} * \mathrm{d}a^j, \qquad (2.17)$$

where

$$\mathcal{G}_{ij} \equiv \frac{1}{2} \frac{\partial^2 K}{\partial s^i \partial s^j} . \tag{2.18}$$

More precisely, by using a metric of the form (2.4), the BPS condition $\bar{\partial}t^i = 0$ implies that

$$\mathcal{H}_{3i} = -M_{\mathrm{P}}^2 \,\mathrm{d}t \wedge \mathrm{d}x \wedge \left(\mathcal{G}_{ij} \,\mathrm{d}s^j\right) = M_{\mathrm{P}}^2 \,\mathrm{d}t \wedge \mathrm{d}x \wedge \mathrm{d}\ell_i \,, \tag{2.19}$$

so that we can identify

$$\mathcal{B}_{2i} = M_{\rm P}^2 \ell_i \mathrm{d}t \wedge \mathrm{d}x\,,\tag{2.20}$$

up to a closed contribution. Therefore, as expected, strings transverse to the (z, \bar{z}) -plane are electrically charged under the two-form potentials \mathcal{B}_{2i} . The dual action describing a single string with charges e^i is given by [42, 43]

$$-\frac{1}{2}\int \mathcal{G}^{ij}\left(M_{\rm P}^2\,\mathrm{d}\ell_i\wedge\ast\mathrm{d}\ell_j+\frac{1}{M_{\rm P}^2}\mathcal{H}_{3\,i}\wedge\ast\mathcal{H}_{3\,j}\right)+S_{\rm string}\,,\qquad(2.21)$$

with

$$S_{\text{string}} = -\int_{\mathcal{S}} \mathrm{d}^2 \xi \, \mathcal{T}_{\mathbf{e}} \sqrt{-h} + e^i \int_{\mathcal{S}} \mathcal{B}_{2\,i} \,, \qquad (2.22)$$

where ξ^a are coordinates along the world-sheet S, h_{ab} is the pulled-back world-sheet metric and

$$\mathcal{T}_{\mathbf{e}} \equiv M_{\mathbf{P}}^2 \, e^i \ell_i \tag{2.23}$$

is the field-dependent string tension.

By using (2.4), the corresponding equations of motions are

$$d\left(\mathcal{G}^{ij} * d\ell_{j}\right) = e^{i}dt \wedge dx \wedge \delta_{2}(\mathcal{S}),$$

$$d\left(\mathcal{G}^{ij} * \mathcal{H}_{3j}\right) = -M_{\mathrm{P}}^{2}e^{i}\delta_{2}(\mathcal{S}).$$
(2.24)

From (2.19) we get $*\mathcal{H}_{3j} = -M_{\rm P}^2 *_2 d\ell_j$ and then both equations reduce to

$$d\left(\mathcal{G}^{ij} *_{2} d\ell_{j}\right) = -d *_{2} ds^{i} = -2i\partial\bar{\partial}s^{i} = e^{i}\delta_{2}(\mathcal{S}), \qquad (2.25)$$

which is indeed satisfied by (2.13a).

In this formulation it is clear that the physical consistency condition $e^i \ell_i > 0$ must be satisfied at each point in field space. In fact, for each $\frac{1}{2}$ BPS string of charges e^i satisfying $e^i \ell_i > 0$, there is an anti-string of charges $\bar{e}^i = -e^i$, which preserves the opposite fraction of supersymmetry. Its contribution to the EFT is obtained by replacing $e^i \to \bar{e}^i$ and $e^i \ell_i \to -\bar{e}^i \ell_i = e^i \ell_i > 0$ in (2.22). The corresponding flow solution is given by the antiholomorphic counterpart of (2.12): $t^i = t_0^i - \frac{1}{2\pi i} \bar{e}^i \log\left(\frac{\bar{z}}{\bar{z}_0}\right)$.

It is important to realise that a correct physical interpretation of the localised string terms appearing in (2.21) and (2.24) strongly depends on the metric structure of the moduli space around the point $s^i = e^i \times \infty$. Take for instance the flat Kähler potential $K = C\phi\bar{\phi}$ for $\phi = e^{2\pi i t}$. In the t chiral coordinate the metric looks degenerate at $s = \text{Im } t = \infty$, but this is just a coordinate singularity. In this case the localised terms in (2.21) and (2.24) are just the dual manifestation of this degenerate parametrisation of field space. Indeed, by using the more natural ϕ coordinate, the local string flow solution (2.12) corresponding to an elementary charge e = 1 is just given by the perfectly smooth local solution $\phi = \phi_0 \frac{z}{z_0}$, in which no localised source appears. This local solution can then be considered 'solitonic'.¹ A similar conclusion holds for any solution around a smooth finite distance point in field space. Instead in this paper we will focus on fundamental localised strings, which induce a flow of the saxions s^i to infinite field space distance.

2.3 Energy-momentum tensor and tension

We now compute the components of the energy-momentum tensor for a single local string flow solution in both dual pictures. We start with the chiral picture where, by imposing the holomorphic condition (2.7), one obtains the following energy-momentum tensor associated with the scalar sector

$$T_{tt} = -T_{xx} = 2M_{\rm P}^2 K_{i\bar{\jmath}} e^{-2D} \partial_z t^i \partial_{\bar{z}} \bar{t}^{\bar{\jmath}},$$

$$T_{z\bar{z}} = T_{zz} = T_{\bar{z}\bar{z}} = 0.$$
(2.26)

Then, by integrating it on a certain domain we get the associated linear energy density:

$$\mathcal{E}_{\text{back}}(r) = \frac{\mathrm{i}}{2} \int_{D(r)} \mathrm{d}z \wedge \mathrm{d}\bar{z} \, e^{2D} T_{tt} = \mathrm{i} M_{\mathrm{P}}^2 \int_{D(r)} K_{i\bar{j}} \, \partial t^i \wedge \bar{\partial} \bar{t}^{\bar{j}} = M_{\mathrm{P}}^2 \int_{D(r)} J_{\mathcal{M}} \,, \qquad (2.27)$$

where in the last equality we have omitted terms containing $d\chi^{\kappa}$, $d\bar{\chi}^{\bar{\kappa}}$, as they vanish on the string solution. As in (2.9), $\mathcal{E}_{\text{back}}$ is related to the kinetic energy density of the string solution. Note that here we have only performed the integral on a disc D(r) of radius rin the (z, \bar{z}) -plane, where non-perturbative corrections are suppressed. In this region one may also assume an axionic symmetry, which further simplifies this result because then $K_{i\bar{j}} = \frac{1}{2}\mathcal{G}_{ij}$ and so

$$J_{\mathcal{M}}|_{\mathrm{sol}} = \mathrm{i}K_{i\bar{j}}\,\mathrm{d}t^{i}\wedge\mathrm{d}\bar{t}^{\bar{j}} = \mathcal{G}_{ij}\,\mathrm{d}a^{i}\wedge\mathrm{d}s^{j} = \mathrm{d}\ell_{i}\wedge\mathrm{d}a^{i} = \frac{1}{2\pi}\,e^{i}\mathrm{d}\ell_{i}\wedge\mathrm{d}\theta\,.$$
(2.28)

The linear energy density associated with a disk of radius r can then be written as

$$\mathcal{E}_{\text{back}}(r) = M_{\text{P}}^2 \int_{D(r)} J_{\mathcal{M}} = M_{\text{P}}^2 e^i \int_0^r \mathrm{d}\ell_i = M_{\text{P}}^2 e^i [\ell_i(r) - \ell_i(0)].$$
(2.29)

¹This solution may be easily completed into a finite energy configuration by considering ϕ as a local coordinate for \mathbb{P}^1 and $K = C |\phi|^2$ as an approximation for $|\phi| \ll 1$ of the Fubini-Study Kähler potential $K = C \log(1 + |\phi|^2)$ on \mathbb{P}^1 . By using (2.10) one can easily see that the tension of this BPS solitonic string is $2\pi C M_{\rm P}^2$.

There is, however, a second contribution to the energy-momentum tensor coming from the localised fundamental string. This contribution is more easily computed using the dual linear multiplet Lagrangian (2.21). In general, the contribution of the localised fundamental string is

$$T_{\text{string}}^{mn} = -M_{\text{P}}^2 e^i \ell_i h^{ab} \partial_a X^m \partial_b X^n \frac{\sqrt{-h}}{\sqrt{-g}} \delta^2(\mathcal{S}) \,. \tag{2.30}$$

In our solution this amounts to

$$T_{tt}^{\text{string}} = -T_{xx}^{\text{string}} = M_{\text{P}}^2 e^{-2D} e^i \ell_i \,\delta^2(r) \,, \qquad (2.31)$$

which contributes to the linear energy density as

$$\mathcal{E}_{\text{string}} = \frac{\mathrm{i}}{2} \int \mathrm{d}z \wedge \mathrm{d}\bar{z} \, e^{2D} T_{tt}^{\text{string}} = M_{\mathrm{P}}^2 e^i \ell_i(0) \,. \tag{2.32}$$

Adding up both contributions we obtain

$$\mathcal{E}(r) \equiv \mathcal{E}_{\text{back}} + \mathcal{E}_{\text{string}} = M_{\text{P}}^2 e^i \ell_i(r) \,. \tag{2.33}$$

In obtaining this result we have implicitly neglected the gravitational contribution to $\mathcal{E}(r)$. In appendix B we argue that such a contribution leaves (2.33) unchanged.

Note that the expression for \mathcal{E} is identical to the tension $\mathcal{T}_{\mathbf{e}}$ that one typically assigns to a BPS string in an unperturbed vacuum, cf. (2.23), for some choice of saxion vevs. In the string solution (2.13), the value of the saxions are determined by the choice of radius as $\ell_i = \ell_i(r)$. Therefore changing the radius corresponds to a flow in the value of the moduli ℓ_i , and so to a flow in the string tension $\mathcal{T}_{\mathbf{e}}$. The result (2.33) then implies that this tension flow is captured by the radial dependence of the linear energy density of the solution. As discussed in [32] and in the next section, this fits nicely with the 4d EFT description of string-like objects, if one interprets 1/r as the mass scale Λ at which the effective string tension is computed.

2.4 EFT strings and validity of the solution

Given the local string solution (2.12) and (2.15), one may reconsider its validity when some of the assumptions taken to derive it no longer apply. First of all, it is important to stress that the solution (2.12) is not unique. In principle one may add an arbitrary linear combination of terms of the form $e^{2\pi i m_i t^i}$, $m_i \in \mathbb{Z}$ without spoiling the monodromy (2.11). Hence the solution (2.12) is physically relevant if such terms can be neglected within a sufficiently large distance from the core of the string. In general this property depends on the specific form of the EFT Kähler potential and on the choice of string charges e^i .

Besides these intrinsic limitations of the above string solution, one may consider the effect of a superpotential that does depend on all the fields ϕ^{α} , and in particular on the fields t^i . This induces a scalar potential V that cannot be neglected in (2.3), and which modifies the equations of motion and their solution. The question is under which circumstances the modification is small enough such that the above solution can still be considered to be a good approximation. In general, any potential V can be easily associated with two

kinds of field dependent energy scales: the mass $m \sim \sqrt{|\partial^2 V|}$ and Hubble $H \sim \sqrt{V}/M_{\rm P}$ scales. Roughly speaking, we can then trust the string solution (2.12) at length scales L well below 1/m and 1/H, within an error of order mL or HL. Hence, as long as m and H remain below the EFT cut-off Λ , there is still a finite energy regime where we can trust the solution. Quantifying the effect of a non-trivial potential in general is quite involved, although one can be more specific for the string solutions that we will consider in this paper.

In this work we will restrict to fundamental axionic strings, which imply the following two assumptions from an EFT perspective. First, the string singular core cannot be resolved with a 4d quantum field theory approach, so the string corresponds to a fundamental localised object in the theory. Second, the Kähler potential should preserve the continuous axionic shift symmetries at perturbative level in an expansion in 1/s, where *s* is the linear combination of saxions that diverges at the string core. In other words, the axionic shift symmetries should be broken only by those non-perturbative corrections of the form $e^{2\pi i m_i t^i}$ that are exponentially suppressed in a sufficiently large disk around the string core. In terms of the above string solutions this implies that we will focus on those of the form

$$\mathcal{C}_{S}^{EFT} = \left\{ \begin{array}{c}
\text{Flows (2.12) along which } K \text{ displays a perturbative ax-} \\
\text{ionic shift symmetry and non-perturbative effects are sup-} \\
\text{pressed} \end{array} \right\}.$$
(2.34)

We will dub the corresponding strings as EFT strings, for reasons to become clear shortly. Note that, by definition, along an EFT string flow solution we can neglect the non-perturbative corrections which break the axionic symmetries and then we are legitimated to pass to the dual formulation of subsection 2.2. This means that the EFT flows can be interpreted as sourced by fundamental localised strings charged under the \mathcal{B}_{2i} gauge fields.

Both assumptions behind the definition of an EFT string are actually correlated. The first one implies that the axion can be interpreted as a *fundamental* axion (a 0-form gauge field) rather than an ordinary pseudo-Goldstone boson, and it is then the gauge invariance of the 0-form gauge field which protects the theory from perturbative corrections spoiling the shift symmetry. If the axion is not fundamental, the dual description in terms of the \mathcal{B}_2 -field will stop being valid at some energy scale $\Lambda_{\mathcal{B}_2}$ associated with a finite number of new degrees of freedom. Above this energy scale $\Lambda_{\mathcal{B}_2}$ the EFT gets UV completed into a different 4d EFT. In this case, the strings are solitonic in the sense that their core can be fully resolved within the 4d EFT above $\Lambda_{\mathcal{B}_2}$. Furthermore, the tension of strings charged under pseudo-Goldstone bosons is typically determined by the symmetry breaking scale. Contrary, the \mathcal{B}_2 -field description of a fundamental axion is always valid at least up to a scale m_* at which the 4d EFT description breaks down because of an *infinite* number of new degrees of freedom. So the string cannot be resolved within a 4d EFT approach, in which case we denote them as fundamental strings. Their tension is a priori arbitrary, although if we want to keep a semiclassical description of these objects we need to impose (2.1).

For fundamental strings, the gauge invariance of the axion should be preserved at any energy scale below m_* , implying that the 0-form global symmetry can only be broken in some specific ways that preserve the (-1)-form gauge invariance associated to a 0-form

gauge field. This negative degree might sound strange, but notice that large transformations of this gauge symmetry are simply the familiar discrete shifts of the axion, while the continuous shift correspond to the 0-form global symmetry. The 0-form global symmetry can be either broken by instantons (electrically charged states) or by coupling it to a (-1)-form gauge field [44, 45] with a flux-induced potential à la axion monodromy. This (-1)-form gauge field strength is simply a dynamical parameter (typically an internal flux) labelling different vacua of a multi-branched potential (see [46–48]). This is the analogous axionic version of a Stückelberg-like coupling providing a mass for a gauge field: the axionic symmetries get spontaneously broken and the axion gets massive by 'eating up' the fluxes. These two possible mechanisms lead to a superpotential of the following form,

$$W = W_{\text{flux}}(\phi) + \mathcal{O}(e^{2\pi i m_i t^i}), \qquad (2.35)$$

where $W_{\text{flux}}(\phi) = M_{\text{P}}^3 f_A \Pi^A(\phi)$ with $\Pi^A(\phi)$ some functions of the chiral fields usually dubbed periods and $f_A \in \mathbb{Z}$ the so-called flux quanta. As above, the exponentially contributions are usually associated to non-perturbative effects due to the presence of BPS instantons charged under the axions, and are exponentially suppressed near the axionic string core by assumption.

Indeed, let us first consider the case where $W_{\text{flux}} = 0$ and the superpotential is a sum of terms of the form $\mathcal{O}(e^{2\pi i m_i t^i})$, $m_i \in \mathbb{Z}$. Such non-perturbative corrections are always expected to be present for at least some values of m_i , modifying both the superpotential and the Kähler potential, and breaking any continuous shift symmetry that the latter may develop in some field space regions. In this sense, assuming that (2.14) takes the EFT to an axionic shift symmetry region amounts to require that $e^i m_i \geq 0$ for all the instanton charges m_i that are relevant in the EFT region, and it essentially identifies the solution near the string core with a perturbative EFT limit, in which all corrections of the form $\mathcal{O}(e^{2\pi i m_i t^i})$ can be neglected. In particular, the second derivative of the scalar potential will be suppressed like $\mathcal{O}(e^{-4\pi m_i s^i})$ which is negligible in the region (2.14). String charges satisfying these positivity constraints correspond to the EFT strings in (2.34). As it will be discussed in section 4, using these observations one can characterise (2.34) as a discrete cone of string charges, directly related with an EFT regime of perturbative axionic symmetries.

Let us now consider the case where the superpotential is of the form $W = W_{\text{flux}}(\phi)$. The axionic string may become anomalous as the 0-form global symmetry may be spontaneously broken to a discrete remnant requiring the simultaneous shift of the axions and the fluxes. This anomaly is cured by attaching the string to membranes, whose charge m_A modifies the flux quanta like $f_A \to f_A + m_A$ as we cross them. Hence, the membranes mediate non-perturbative transitions between the different branches of flux vacua. As pointed out in [42] and elaborated in [32], not all of these transitions can be described dynamically at the level of the EFT, but only a sublattice Γ_{EFT} of them with a tension \mathcal{T}_{mem} satisfying $\frac{T_{\text{mem}}}{M_p^2} < \Lambda$. Those flux quanta not connected by Γ_{EFT} should be considered as separate sectors from the EFT viewpoint, each corresponding to a different choice of non-dynamical flux quanta. The question is then for which of these choices the string solution (2.12) is still a good approximation. By our discussion above, we may evaluate the effect of W_{flux} near the string core, where non-perturbative corrections are assumed to die off. Intuitively, a first necessary condition for the string solution not to be spoiled is that the Hubble scale vanishes asymptotically along (2.12), which is the kind of potentials analysed in [49]. The associated membranes are selected by requiring $\mathcal{T}_{\text{mem}} < M_P^3$ in the perturbative regime, and the induced potential exhibits a runaway behaviour in the limit (2.14). Additionally, one should require that $(K_{i\bar{j}}^{-1}\partial_{ti}\partial_{\bar{l}\bar{j}}V)^{1/2}$ is small compared to the EFT cut-off Λ . Following the discussion in [32, section 4.3],² one may see that this will be the case if W_{flux} involves periods $\Pi^A(\phi)$ that either do not depend on the direction (2.14) or, if they do, they correspond to flux quanta within Γ_{EFT} . All remaining fluxes will generate a potential which, asymptotically along (2.14), generates a mass for this field direction above Λ , so (2.14) cannot actually be considered as an EFT field direction. Hence, we get a self-consistent picture in which EFT strings get attached to EFT membranes associated to fluxes inducing a potential that dies fast enough in the perturbative regime. For those choices of fluxes in which the limit (2.14) is not obstructed and makes sense in the EFT, the string solution (2.12) should be a good approximation in the said limit.

To sum up, for EFT strings all non-perturbative effects will die along the trajectory (2.14), and a sensible choice of flux quanta guarantees that the BPS string solution (2.12) can be trusted in a large region near the string core, even in the presence of a potential. For this reason, these EFT string solutions will always make sense from the viewpoint of an $\mathcal{N} = 1$ EFT. In fact, as will be discussed in section 3, their profiles admit a compelling RG flow interpretation, as expected for localised operators of an EFT. This interpretation will support that, for our analysis to make sense, the fields entering the string solution need not be strictly massless, but only light compared to the EFT cut-off scale Λ .

Finally, let us consider the case of axions coupled à la Stückelberg to one or more U(1) gauge fields. That is, the axion may shift as $a^i \to a^i + \lambda p^i$ under a U(1) gauge transformation $A \to A + d\lambda$, with $p^i \in \mathbb{Z}$. One can still dualise the axions a^i into \mathcal{B}_{2i} , which couple to the gauge field A via a term $p^i \int F \wedge \mathcal{B}_{2i}$. This coupling makes massive the \mathcal{B}_2 -field, so that a string of charges p^i can break by nucleation of a pair of a magnetic monopoles. The mass-squared is of order $m^2 \sim g^2 M_{\rm P}^2 \mathcal{G}_{ij} p^i p^j$, with g being the U(1) gauge coupling which is typically of order $g \sim 1/\sqrt{s}$, where s is the linear combination of saxions that couples linearly to $F \wedge *F$. Following the reasoning above, as long as this mass is below the cut-off, there will exist some energy regime in which the EFT string solution will be a good approximation, and the string instability can be neglected. This seems a reasonable assumption in the perturbative regime; as it will be shown in the next section, the 2-form gauge coupling \mathcal{G}_{ij} vanishes asymptotically. Therefore the mass will be small as long as the 1-form gauge coupling g does not blow up. From the perspective of the axion, the 0-form global axionic symmetry gets gauged as the gauge field A gets massive by eating up the axion. Hence, instanton-breaking terms can only appear if the theory contains electrically charged particles. Here we can also make the distinction between fundamental axions or those coming from a sort of Abelian Higgs model. Recall that only the first case is of

 $^{^{2}}$ This analysis assumes a particular class of field metrics which we will encounter in the next section, cf. (3.8).

interest for us, as the associated strings are fundamental rather than solitonic. Finally, Stückelberg couplings generate D-term potentials with saxion-dependent Fayet-Iliopoulos terms in $\mathcal{N} = 1$ EFTs, which could in principle distort the corresponding string flow. In those cases we will however assume the presence of matter ϕ^{α} charged under the U(1)fields, such that there is always a flat D-term direction along (t^i, ϕ^{α}) which defines the actual string flow.

3 Strings and infinite field distance limits

As we have seen above, 4d EFT string solutions feature a logarithmic backreaction that describes a flow in the field space \mathcal{M} of our EFT, and more precisely on the saxionic fields s^i . Therefore, following the same philosophy as in [25, 29] (see also [17, 26–28, 30, 31, 44, 50]), one may use these string solutions to study large variations in the field space of the EFT. In particular, as we will argue, EFT string flows are naturally associated to field space trajectories of infinite distance, and they provide a unique tool to test the physics of constant-field configurations. In addition, the string solution has a natural RG flow interpretation, which allows one to link variations along \mathcal{M} with changes of the EFT cut-off Λ . This dictionary between string backreacted solutions, RG flows and field space variations yields several important lessons. In particular, just like Λ has a finite range and above some scale one must drop the 4d EFT description, the same occurs after some distance along flows in \mathcal{M} generated by EFT string solutions.

3.1 Physics along the string solution

If we consider a certain point in field space $\mathbf{t}_0 = \{t_0^i\} \in \mathcal{M}$ and an EFT string of charge \mathbf{e} , then the solution (2.13) will realise a flow on the saxions $\mathbf{s} = \{s^i\}$ from an initial value \mathbf{s}_0 at r_0 to the boundary of field space (2.14), as we approach the string location at r = 0. Going in the opposite direction, some s^i will become of order one and expected corrections of order $\mathcal{O}(e^{2\pi i t^i})$ will become significant, a point at which the local solution (2.12) can no longer be trusted. Therefore, even if a general 4d string solution can be seen as a map from \mathbb{C} to the EFT field space \mathcal{M} , an EFT string solution selects a disc $D(r_0) \subset \mathbb{C}$, which is mapped to a perturbative region of \mathcal{M} in which all non-perturbative effects can be neglected. In particular, non-perturbative corrections of the form $\mathcal{O}(e^{2\pi i m_i t^i})$ with $m_i e^i > 0$ will be more and more suppressed as we proceed along the flow (2.13a) towards $r \to 0$, until we reach an exact axionic shift symmetry in the limit (2.14). By standard quantum gravity arguments [35, 36], we only expect to realise such global continuous symmetries at infinite distance in field space. As a result, consistency with quantum gravity relates EFT string locations to infinite distance points on the boundary of \mathcal{M} , and the backreacted saxionic trajectory (2.13a) to infinite distance paths in \mathcal{M} .

This relation can be quantified by parametrising the radial saxionic flow (2.13a) as

$$s^i(\sigma) = s^i_0 + \sigma e^i \,, \tag{3.1}$$

where we have defined $2\pi\sigma = \log(r_0/r)$. For a given s_0^i , let us denote by σ_* the value of σ at which the flow reaches the boundary of the saxionic field space. The field space distance

travelled by the radial flow is given by

$$d_* = \int_{\text{flow}} \sqrt{\mathcal{G}_{ij} ds^i ds^j} = \int_0^{\sigma_*} d\sigma \sqrt{e^i e^j \mathcal{G}_{ij}(\sigma)} = \frac{1}{M_{\text{P}}} \int_0^{\sigma_*} \mathcal{Q}_{\mathbf{e}}(\sigma) d\sigma , \qquad (3.2)$$

where in the last equality we have introduced the physical string charge Q_e , defined as

$$\mathcal{Q}_{\mathbf{e}}^2 = M_{\mathbf{P}}^2 \mathcal{G}_{ij} e^i e^j \,. \tag{3.3}$$

This definition of $\mathcal{Q}_{\mathbf{e}}$ is well-motivated in the field space region probed by the EFT flow along $D(r_0)$, since it lies within a weakly-coupled region where axionic shift symmetries are a good approximation. One can then resort to the dual formulation described in section 2.2 in terms of two-form potentials \mathcal{B}_{i2} to which the strings couple electrically. Since the kinetic matrix for the \mathcal{B}_{2i} is $M_{\rm P}^{-2}\mathcal{G}^{ij}$, see (2.21), the physical charge is then defined as in (3.3). Notice that along the flow (3.1) $\mathcal{Q}_{\mathbf{e}}(\sigma)$ is simply given by

$$\mathcal{Q}_{\mathbf{e}}(\sigma) = M_{\mathrm{P}} \sqrt{\frac{\mathrm{d}^2 K(s(\sigma))}{2\mathrm{d}\sigma^2}} \,. \tag{3.4}$$

Quantum gravity arguments constrain the possible asymptotic behaviour of $\mathcal{Q}_{\mathbf{e}}$ along the EFT flow. First, one can argue that necessarily $\sigma_* = \infty$, which is what we indeed expect for an EFT string flow of the form (2.13a), since there the axionic shift symmetry gets restored at the string core r = 0. For this, notice that $\frac{1}{2\pi}M_{\rm P}\mathcal{Q}_{\mathbf{e}}$ can also be interpreted as axion decay constant f_{ϑ} of the axionic direction ϑ that shifts by 2π around the EFT string: $a^i = \frac{1}{2\pi}e^i\,\vartheta,\,\vartheta \simeq \vartheta + 2\pi$. The axion ϑ has the kinetic term

$$-\frac{1}{2}\int f_{\vartheta}^{2}\,\mathrm{d}\vartheta\wedge\ast\mathrm{d}\vartheta = -\frac{1}{8\pi^{2}}M_{\mathrm{P}}^{2}\int \mathcal{Q}_{\mathbf{e}}^{2}\,\mathrm{d}\vartheta\wedge\ast\mathrm{d}\vartheta\,.$$
(3.5)

An instanton of charges m_i has action $S_{\mathbf{m}} = 2\pi m_i s^i$, and the axionic version of the WGC [51] applied to these instantons reads

$$f_{\vartheta} S_{\mathbf{m}} \le \gamma_{\mathbf{m}} M_{\mathrm{P}} \qquad \Leftrightarrow \qquad \mathcal{Q}_{\mathbf{e}}(m_i s^i) \le \gamma_{\mathbf{m}} \,, \tag{3.6}$$

where $\gamma_{\mathbf{m}}$ is some constant depending only on the instanton charges m_i . The positive quantity $m_i s^i$ does not decrease along the EFT flow for allowed instanton charges m_i , and so (3.6) implies that $\mathcal{Q}_{\mathbf{e}}$ must remain finite along the corresponding EFT flow. As said, the absence of global symmetries requires that $d_* = \infty$, which can then only happen if $\sigma_* = \infty$.

Second, since $m_i s^i \sim (m_i e^i) \sigma$ for $\sigma \to \infty$ with $m_i e^i \ge 0$, then (3.6) implies that $\mathcal{Q}_{\mathbf{e}}$ vanishes asymptotically along the flow: $\mathcal{Q}_{\mathbf{e}} \to 0$. Notice however that (3.2) and $d_* = \infty$ imply that $\mathcal{Q}_{\mathbf{e}}$ cannot go to zero too quickly, although all conditions are satisfied if $\mathcal{Q}_{\mathbf{e}} \sim \sigma^{-\eta}$, with $\eta \in (0, 1]$. This range can be reduced if we take into account that the asymptotic behaviour of the string tension $\mathcal{T}_{\mathbf{e}}(\sigma) = M_{\mathrm{P}}^2 e^i \ell_i(\sigma)$ is governed by

$$\frac{\mathrm{d}\mathcal{T}_{\mathbf{e}}(\sigma)}{\mathrm{d}\sigma} = -\mathcal{Q}_{\mathbf{e}}^2 < 0\,,\tag{3.7}$$

where we have used (2.17) and (2.18). Requiring that $\mathcal{T}_{\mathbf{e}}$ does not become negative at $\sigma_* = \infty$ further restricts the range to $\eta \in (1/2, 1]$. Moreover, for EFT strings we expect

that a WGC-like bound is saturated asymptotically for $\sigma \to \infty$, so that an asymptotic relation of the form $M_{\rm P} \mathcal{Q}_{\mathbf{e}} \sim \gamma \mathcal{T}_{\mathbf{e}}$ with a constant γ is satisfied. It is easy to see that this only happens if $\mathcal{Q}_{\mathbf{e}} \sim \sigma^{-1}$.

Therefore, the only reasonable option for EFT strings seems to be a physical charge $Q_{\mathbf{e}}$ that vanishes asymptotically like σ^{-1} , with a similar behaviour for the tension $\mathcal{T}_{\mathbf{e}}$. As such, the proper distance diverges logarithmically as $\sigma \to \sigma_* = \infty$, and using (3.4) one obtains a Kähler potential which asymptotically takes the form $K \sim -n \log \sigma$, for $n \in \mathbb{R}^+$. This is indeed the case for all the string theory examples analysed in section 6, where we have that asymptotically

$$K \simeq -\log P(s), \qquad (3.8)$$

with P(s) some homogeneous function of positive integral degree on the saxions. Hence, we can use the WGC and the existence of EFT strings as a physical motivation for the behaviour of the Kähler potential observed in string theory compactifications. Notice that then the dual saxions ℓ_i decrease as σ^{-1} along the trajectory (3.1), just like (3.4) and the tension $\mathcal{T}_{\mathbf{e}}$. Finally, using (3.8) one may argue that EFT string flows are asymptotically geodesic, see appendix C.

Let us remark that, in practice, not any choice of charges e^i corresponds to an EFT string, nor to a point at infinite distance. Indeed, as we will discuss in section 4, some choices of e^i correspond to saxionic flows along which some non-perturbative effects become relevant and destroy the axionic shift symmetry. Interestingly, for such cases the string flow reaches the field space boundary for a finite value of the parameter σ in (3.1), at least at the classical level.

To sum up, as we proceed along the string flow (2.13a) towards the core of an EFT string, we probe regions of the EFT field space along which the string becomes tensionless and weakly coupled, and the associated axionic symmetry becomes more and more exact. This picture may provide the wrong impression that at the string core we reach a regime in which strings are tensionless and the axionic symmetry exact. This is however not so, because (2.12) is not the actual flow realised by an EFT with a finite cut-off Λ . Instead, an approximation of (2.12) is realised, which tends towards (2.12) as we increase Λ . Taking this observation into account leads in fact to an interesting RG flow interpretation of the string solution, which we now turn to discuss.

3.2 RG flow interpretation

In an EFT with a given cut-off Λ the string backreaction profile will not look like (2.13), but rather like a coarse-grained profile in which only 4d Fourier modes up to momentum Λ can enter. From this kind of observation stems the EFT interpretation of the backreaction of extended objects, used in [38, 39] to interpret *p*-brane backreaction in string theory. Following the same philosophy, a 4d string backreaction can be interpreted in terms of a classical β -function for the 4d string couplings to the bulk fields, and in particular for its effective charge and tension.

To apply this philosophy to our context, let us consider the linear energy density $\mathcal{E}(r)$ of the EFT string solution contained in a disk of radius r, as computed in section 2.3. As

follows from (2.33), $\mathcal{E}(r)$ is a sum of two contributions, namely the energy contained in the string backreaction $\mathcal{E}_{\text{back}}$ and the localised contribution $\mathcal{E}_{\text{string}}$. While the sum $\mathcal{E}(r)$ is a physical, fixed quantity that only depends on the radius r, the contribution of each term depends on the energy cut-off Λ of the EFT describing the axionic string.

From this observation one can derive interesting consequences. On the one hand, if we take the formal limit $\Lambda \to \infty$, then $\mathcal{E}_{\text{back}}$ captures the full energy of the profile (2.13), and $\mathcal{E}_{\text{string}}$ reaches its minimum. On the other hand, if we take $\Lambda \sim 1/r$ the EFT essentially sees a flat backreaction profile and no energy can be stored in the term $\mathcal{E}_{\text{back}}$. Equivalently, for a given cut-off Λ the choice of radius $r = 1/\Lambda$ implies that $\mathcal{E}_{\text{back}}(r) = 0$, and the whole contribution to the linear energy is stored in $\mathcal{E}_{\text{string}}$. Notice that this radius is the minimal distance from the string that can be resolved by the EFT and, by the previous observation, the point from where we can read the string couplings (2.30) at the cut-off-scale Λ , directly from the string backreaction (see figure 1).

In other words, the string backreaction (2.12) realises the RG flow of its couplings, with the cut-off-scale setting the distance $1/\Lambda$ at which the couplings are defined. The bulk profile along the radial direction r can then be interpreted as an RG flow of the brane couplings as one changes $\Lambda \sim 1/r$. In particular, we may identify the effective string tension at the scale Λ with

$$\mathcal{T}_{\rm str}(\Lambda) = \mathcal{E}(r_{\Lambda}) = M_{\rm P}^2 e^i \ell_i(r_{\Lambda}), \qquad (3.9)$$

where $r_{\Lambda} \equiv 1/\Lambda$ is the radius where the coarse-grained flow stops.³ In this sense, approaching the string location along the profile (2.13) can be related to increasing the EFT cut-off, with the corresponding changes in the effective string tension and charge. This variation can be easily linked to our above discussion if we express the flow parameter as $\sigma = \frac{1}{2\pi} \log(\Lambda r_0)$, with Λ the cutoff at which the effective string tension is defined. Then (3.7) becomes

$$2\pi\Lambda \frac{\mathrm{d}\mathcal{T}_{\mathbf{e}}(\Lambda)}{\mathrm{d}\Lambda} = -\mathcal{Q}_{\mathbf{e}}^2 < 0.$$
(3.10)

It follows from our previous discussion that for EFT strings $\mathcal{T}_{\text{str}}(\Lambda)$ decreases as $\Lambda \to \infty$. In particular, for Kähler potentials of the form (3.8) this monotonically decreasing behaviour is such that $\mathcal{T}_{\text{str}}(\Lambda) \to 0$ as $\Lambda \to \infty$. This can either be seen by looking at the asymptotic behaviour of the dual saxions ℓ_i coupling to the string, or by performing a direct 4d field theory computation of the string RG flow along the lines of [52], see appendix D.

In light of these considerations, some comments are in order:

- Near the EFT cut-off scale Λ all the effects at a much lower scale can be neglected. In this way we may reinterpret our results of section 2.4, in which the different potentials that appear in sensible setups do not modify the string solution near the string core. In this sense, the relevant EFT field space \mathcal{M} for our analysis can either contain flat directions or fields whose mass is much smaller than Λ . In the following, a point in

³More precisely, one should use the backreacted metric to compute the distance from the string core. By using (3.8), (2.15) gives a warping $e^{2D} \sim P(s)$. Applied along the saxionic flow (2.13a), this produces a logarithmic correction to the flat-space identification $r_{\Lambda} \sim 1/\Lambda$. In the following we will ignore such a logarithmic correction.



Figure 1. Axionic string backreaction (2.13a) and (2.13b). In gray, we highlighted the region $r \leq r_{\Lambda}$ that cannot be resolved within the EFT.

 \mathcal{M} will denote either a vacuum or a constant-field configuration in which the Hubble and mass scales of the potential are negligible compared to Λ .

- Even if the string solution of section 2 is obtained in a supersymmetric setup, our conclusions should also hold for EFTs in which supersymmetry is spontaneously broken. Indeed, SUSY-breaking effects will modify the string solution, but if supersymmetry is restored at some scale Λ_{SUSY} then such a solution will look like the above $\frac{1}{2}$ BPS string solution at distances shorter than $r_{SUSY} = 1/\Lambda_{SUSY}$. Notice that our reasoning only depends on the asymptotic behaviour of the string solution near the string core, and is therefore insensitive to SUSY-breaking effects or any mass deformation well below the EFT cut-off Λ . In particular, for the sake of the argument that leads to the behaviour $Q_e \sim \sigma^{-1}$, the saturation of the WGC-bound only needs to be imposed asymptotically. In terms of the RG flow interpretation, this means that the reasoning also applies to EFT strings that may not be seen as BPS in the IR, but become so in the UV.
- An analogous reasoning can be applied in cases where the axionic shift symmetry is broken by a perturbative superpotential. Indeed, as already discussed in section 2.4, we may consider a superpotential of the form (2.35) with W_{flux} breaking the axionic shift symmetry. However, if the flow still makes sense as a field space direction, the string solution of section 2 should be a good approximation near the string core. Then, since our reasoning in this section only depends on the shift symmetries of the Kähler potential and the subsequent properties of the string charge $Q_{\mathbf{e}}$, having $W_{\text{flux}} \neq 0$ should not affect the fact that the string should be located at infinite field distance, as well as all the consequences that follow. Pictorially, 4d EFT string solutions whose axionic symmetry is broken by W_{flux} correspond to solutions with a radial branch cut, in which a membrane is placed to render the string operator gauge

invariant. Away from the membrane location the radial flow of the saxion should be less and less distorted as we approach the string core, and the discussion of this section should follow through.

3.3 String flows as paths in field space

One of the main applications of EFT string solutions and their RG flow interpretation is to extract physical information about similar paths in the EFT field space \mathcal{M} . That is, we would like to map the EFT string solution (2.12) to a family of constant-field configurations that describe a trajectory in field space

$$\gamma_{\mathbf{e}} \equiv \{ \bar{\boldsymbol{t}} = \bar{\boldsymbol{a}}_0 + \mathrm{i}\bar{\boldsymbol{s}}(\sigma) \} \subset \mathcal{M}, \qquad \bar{\boldsymbol{s}}(\sigma) = \bar{\boldsymbol{s}}_0 + \sigma \mathbf{e}, \qquad \bar{\boldsymbol{a}}_0, \bar{\boldsymbol{s}}_0 = \mathrm{const.}$$
(3.11)

now in the absence of any backreacting string and for a fixed EFT cut-off Λ . Here we use barred quantities to denote constant-field configurations parametrised by σ , as opposed to e.g. the local value of the saxion s(r) along the string solution (2.13a).

A first question is if the physics of a constant-field configuration $\bar{t} \in \gamma_{\mathbf{e}} \subset \mathcal{M}$ is captured by a local space-time patch of the backreacted EFT string solution (2.12), around the point $z \in \mathbb{C}$ such that $t(z) = \bar{t}$. The difference between these two cases is that in the string solution the fields are not constant. Because of the logarithmic profile, the derivatives of the EFT string solution set a length scale of order $1/r \equiv 1/|z|$, after which we start seeing the field variations. Accordingly, we expect that a local patch of the string solution captures the physics of a constant-field configuration provided that we consider energies above 1/r. Then, because our EFT only describes energies up to Λ , we are restricted to the energy range $1/r < E < \Lambda$. Notice that this is consistent with the fact that the coarse-grained string solution stops at a distance $r_{\Lambda} \sim 1/\Lambda$ from the string core, so that at shorter distances we have a constant profile.

Therefore, if we want to understand the physics of constant-field configuration at energies near Λ , one may use the EFT string solutions to probe a region near $\bar{t} \in \mathcal{M}$. Notice that the paths covered by a solution at finite Λ are large but of finite distance, so we never reach infinite distance points in this way. However, we may still use such finite distance paths to see the asymptotic behaviour of the theory along large field distances in \mathcal{M} . This will be the philosophy underlying the following sections of the paper.

As emphasised above, of particular interest to us is to test the SDC along large distance paths $\gamma_{\mathbf{e}} \subset \mathcal{M}$. This involves considering the mass scale m_* of a tower of states that lies above Λ , and therefore that it has been integrated out in our EFT description, so it is quite hard to guess the moduli dependence of m_* purely from EFT data. Nevertheless, for regions of field space \mathcal{M} probed by EFT string solutions, it is natural to assume that any microscopic completion of our EFT contains an energy threshold $E_{\mathbf{e}} > \Lambda$, for any fundamental axionic string charge \mathbf{e} that exists in the theory. This can be argued by means of the Completeness Conjecture [53], which would predict the existence of the corresponding string state, and from associating $E_{\mathbf{e}}$ with the mass of the lightest string oscillation modes. When the charge \mathbf{e} corresponds to an EFT string that is $\frac{1}{2}$ BPS in the UV, and we are in a region of \mathcal{M} where the corresponding axionic shift symmetry is a good approximation, this energy threshold should be simply given by $E_{\mathbf{e}}^2(\bar{s}) \equiv T_{\mathbf{e}}(\bar{s}) = M_{\mathbf{P}}^2 e^i \bar{\ell}_i$. That is, the *probe* string tension $T_{\mathbf{e}}$ that is computed from dimensional reduction in string compactifications to 4d, see section 6 for examples.

Both quantities $T_{\mathbf{e}}$ and $\mathcal{T}_{\mathbf{e}}$ have the same functional dependence on the dual saxions (2.23), as dictated by supersymmetry and axionic shift symmetry. In fact, when we evaluate $\mathcal{T}_{\mathbf{e}}$ along its string solution as in section 3.1, what we are doing is to estimate $T_{\mathbf{e}}$ along the corresponding saxionic path (3.11) in field space. The connection of $T_{\mathbf{e}}$ with the RG flow interpretation of the string solution is however less obvious, because $\mathcal{T}_{\mathbf{e}}(\Lambda)$ depends on the cut-off while $T_{\mathbf{e}}$ does not; $\mathcal{T}_{\mathbf{e}}(\Lambda)$ appears as a coupling of an EFT operator, while $T_{\mathbf{e}}^{1/2}$ corresponds to the mass of a closed oscillating string of radius $T_{\mathbf{e}}^{-1/2}$, and so it is beyond the EFT resolution scale.

To clarify their relation let us consider a closed string on a loop of radius $L > 1/\Lambda$, that we add 'on top of' a constant field configuration $\bar{t} \in \mathcal{M}$. This closed string will backreact as in (2.12) with and UV cut-off at Λ , and now also with an IR cutoff at approximately 1/L, coming from the fact that the string forms a closed loop. Indeed, beyond a distance Lfrom the string its backreaction will quickly die off, and we will just see an approximately constant field configuration such that $\text{Im } t \simeq \bar{s}$, which we identify with a point in \mathcal{M} . Microscopically, we associate a total energy of order $T_{\mathbf{e}}(\bar{s})L$ to this string loop. Then, similarly to (2.33), we can split the total energy of the system as

$$T_{\mathbf{e}}(\bar{s})L = \mathcal{T}_{\mathbf{e}}(\Lambda, \bar{s})L + E_{\text{back}}(\Lambda, L, \bar{s}) , \qquad (3.12)$$

where the l.h.s. is independent of the EFT cut-off Λ . In the limiting case in which $\Lambda \sim 1/L$ there is no backreaction, and we recover the microscopic result $\mathcal{T}_{\mathbf{e}}(1/L, \bar{s}) = T_{\mathbf{e}}(\bar{s})$. If we now increase Λ a backreaction will be developed, such that the fields t will start flowing towards the string core. More precisely we will have a coarse-grained version of the flow (2.13), with $r_0 = L$ and $s_0 = \bar{s}$. As described above, this flow will probe the physics of different constant-field configurations, until the region that corresponds to $\bar{s}_{\Lambda} \equiv s(r_{\Lambda} \equiv 1/\Lambda) = \bar{s} + \frac{\mathbf{e}}{2\pi} \log(\Lambda L)$. By using the symmetries of the solution (2.12), or directly from the similarities between (3.7) and (3.10) it follows that⁴

$$\mathcal{T}_{\mathbf{e}}(\Lambda e^{2\pi\sigma}, \bar{s}) = \mathcal{T}_{\mathbf{e}}(\Lambda, \bar{s} + \mathbf{e}\sigma).$$
(3.13)

From here we have that $\mathcal{T}_{\mathbf{e}}(\Lambda, \bar{s}) = \mathcal{T}_{\mathbf{e}}(1/L, \bar{s}_{\Lambda}) = T_{\mathbf{e}}(\bar{s}_{\Lambda})$. That is, the renormalised tension $\mathcal{T}_{\mathbf{e}}(\Lambda, \bar{s})$ describes $T_{\mathbf{e}}$ in the vicinity of the string core. Equivalently, one can see $\mathcal{T}_{\mathbf{e}}^{1/2}(\Lambda, \bar{s})$ as the smallest amount of energy needed to create a string of size $T_{\mathbf{e}}^{-1/2}$ in this whole configuration.

To sum up, by changing the cut-off Λ in the above setting generates a saxionic profile that interpolates between \bar{s} and $\bar{s}_{\Lambda} \equiv s(r_{\Lambda})$. In this way, we probe the physics of constantfield configurations that follow the same path in \mathcal{M} , with $\mathcal{T}_{\mathbf{e}}(\Lambda, \bar{s})$ measuring how $T_{\mathbf{e}}$

$$\mathcal{T}^{-1}(\Lambda, s) = \mathcal{T}^{-1}(\Lambda_0, s_0) + \frac{2}{neM_{\mathrm{P}}^2} \left(s - s_0 + \frac{e}{2\pi}\log\frac{\Lambda}{\Lambda_0}\right) \,.$$

⁴One can illustrate this symmetry explicitly by considering a single-field model with a Kähler potential of the form $K = -n \log s$. Then the EFT string flow yields the expression

changes along such paths. The larger ΛL is, the larger the path in \mathcal{M} , although some limitations will be in place. For instance, if $\mathcal{T}_{\mathbf{e}}(\Lambda, \bar{s}) \leq \Lambda^2$, we would be describing a region where our assumption (2.1) does not hold, and so we would not expect to capture the physics of a constant-field configurations in our 4d EFT. In fact, as we argue below, for $\bar{t} \in \mathcal{M}$ such that $T_{\mathbf{e}}(\bar{s}) \leq \Lambda^2$ the semiclassical description of our EFT should break down.

It follows that those flows that are useful to probe the physics of EFT constant-field configurations must satisfy $\Lambda^2 < \mathcal{T}_{\mathbf{e}}$. In terms of EFT string solutions, this condition translates into a restriction of the parameters (\mathbf{s}_0, r_0) that enter (2.13a) or, in the above closed string configuration, of the parameters $(\mathbf{\bar{s}}, L)$. By requiring that $\Lambda^2 < \mathcal{T}_{\mathbf{e}}(1/r, \mathbf{\bar{s}})$ for any $r \in (r_{\Lambda}, L)$, one in particular constrains the allowed values for $\mathbf{\bar{s}}_{\Lambda}$, and therefore sets a bound on the choice of constant-field configuration $\mathbf{\bar{t}} \in \mathcal{M}$ probed by EFT string flows. This is consistent with the fact that an upper bound for Λ translates, via (3.13), into a maximal field range along saxionic directions generated by EFT strings. In fact, the correspondence (3.13) suggests that, just as the 4d EFT stops being valid at some scale above Λ , the same should happen along large distances in \mathcal{M} generated by EFT string flows. This observation is at the origin of why along large field space trajectories corresponding to EFT strings one recovers the physics predicted by the SDC, a link that will be made more precise in section 5.

3.4 EFT breakdown

Our results regarding the effective string tensions $\mathcal{T}_{\mathbf{e}}$ indicate how the mass of the lightest oscillation modes of the different EFT strings $T_{\mathbf{e}}^{1/2}(\bar{s})$ vary as we move in the EFT field space \mathcal{M} along the paths (3.11). By assumption, such modes lie above the EFT cut-off scale Λ , and so we must impose $T_{\mathbf{e}}(\bar{s}) > \Lambda^2$. This is reminiscent of the first inequality in (2.1), which follows from assuming that EFT strings are fundamental objects. On general grounds, we expect that fundamental objects lie above the EFT cut-off scale, since the EFT semiclassical description should break down whenever we are able to resolve one of them.

What the results of this section show is that, as we move in field space \mathcal{M} for fixed cut-off Λ along the path (3.11), the corresponding threshold $E_{\mathbf{e}} \equiv T_{\mathbf{e}}^{1/2}$ will decrease monotonically towards zero, and so at some point s_{break} we will have that $T_{\mathbf{e}}(s_{\text{break}}) = \Lambda^2$. Beyond such a point, the EFT string should trigger the breakdown of the EFT semiclassical description. In this way any point of infinite distance that corresponds to an EFT string flow endpoint will present at least one natural candidate for a tower of states that satisfy the Swampland Distance Conjecture, as such string will become tensionless as we proceed along the path, and therefore its oscillation modes massless. One may even argue that the SDC will be satisfied whenever the said EFT string satisfies the WGC, deriving this way the exponential behaviour of the cut-off, see section 5.2 for details.

Let us discuss in some detail why a string threshold $T_{\mathbf{e}}^{1/2}$ below the cut-off implies the breakdown of the EFT and whether this is really due to an infinite tower of states as the SDC requires. First of all, we can argue for the existence of new degrees of freedom signaling the EFT breakdown when $T_{\mathbf{e}} < \Lambda^2$ for weakly coupled strings. For this, consider some classical fluctuating string solution, like the closed string loop in eq. (3.12). As in there, this classical state has center-of-mass energy $E = T_{\mathbf{e}}L$, where L can be identified with either the oscillation time-scale or the length of the string. On the one hand, the local low-energy EFT regime seems legitimate for $L \gtrsim r_{\Lambda} \equiv 1/\Lambda$, so the classical estimate for the mass of the lightest states will be $E \sim T_{\mathbf{e}}/\Lambda$. On the other hand, the self-consistency of our semiclassical description of these string states requires the characteristic string length scale L to be bigger than its Compton wavelength 1/E. Hence, the semiclassical description of the lightest states with $L \sim r_{\Lambda}$ is only justified if $T_{\mathbf{e}} > \Lambda^2$. In other words, the semiclassical description of this continuous family of states breaks down if $T_{\mathbf{e}} < \Lambda^2$. Rather, they should be treated as quantum states to be included in the EFT spectrum. Taking into account that the string is weakly coupled, we expect that the continuum spectrum of fluctuations gets discretised and yields an infinite tower upon quantisation, as it happens for critical strings. However, this will remain an assumption for us.

Even if an asymptotically tensionless EFT string indicates the eventual breakdown of the EFT, it does not specify the nature of the breakdown. As said, the oscillation modes of the string provide a natural candidate for a tower of states realising the SDC, but there could be other towers of lighter states that force the EFT breakdown before the string becomes tensionless. Indeed, this effect has already been observed in [15] (see also [16]) where these lighter towers of states were interpreted as decompactification limits. In fact, on general grounds we would expect that a tower of modes of mass m_* like KK modes becomes light along an EFT string flow at least as fast or even faster than the string tension. Otherwise we would be able to decouple both scales and engineer 4d string theories with an infinite number of oscillation modes in a space which is approximately Minkowski. In section 5 we will propose a precise relation between the behaviour of the scale m_* in terms of the string tension which, to the best of our knowledge, is satisfied in all string theory compactifications, and which fits nicely with the Emergent String Conjecture of [15].

We have seen that any EFT string backreaction drives the scalars along large field space distances. In other words, EFT string flows dynamically explore different perturbative vacuum sectors of the theory. In fact, as we will see in explicit examples, by taking different string charges one can explore all the asymptotic limits of a given perturbative regime via EFT string flows. Again, the vanishing asymptotic tension is guaranteed by a Kähler potential of the form (3.8), found near weakly-coupled infinite distance points in string theory compactifications. It is thus tempting to speculate that all points of this sort should correspond to EFT strings, an idea upon which we will elaborate in section 5. Notice that this proposal fits nicely with the results of [6, 8, 9] which classify infinite distance limits in terms of monodromies of particle charges. In our $\mathcal{N} = 1$ setup this monodromy is still there, and it is physically realised by the discrete shift (2.11). As discussed in [32], in our more general setup it does not act on BPS particles as in the $\mathcal{N} = 2$ setups of [6, 8, 9], but instead on membranes.

Incidentally, this picture provides an interesting purely classical realisation of the Emergence Proposal [6, 54–56]. For a fixed value of the EFT cut-off Λ , the EFT field space consistent with our description, which we dub \mathcal{M}_{Λ} , can only contain finite distances. Indeed, as explained above, when proceeding along the saxionic paths (3.11) of infinite



Figure 2. The moduli space \mathcal{M}_{Λ} accessible within an EFT characterised by a cutoff Λ .

distance in \mathcal{M} generated by EFT strings, we will reach a set of points $\{t_{\text{break}}\}\$ at which $T_{\mathbf{e}}(s_{\text{break}}) = \Lambda^2$. At these points the EFT will break down, defining a boundary for \mathcal{M}_{Λ} . Hence, at finite cut-off our EFT is associated with a finite scale-dependent field space \mathcal{M}_{Λ} . As we lower Λ and we integrate out 4d high energy modes, the paths $\gamma_{\mathbf{e}}$ will be allowed to reach smaller values for $e^i \ell_i$, which correspond to points at a larger field space distance and smaller string tensions. So \mathcal{M}_{Λ} will grow and, in the limit $\Lambda \to 0$, it will asymptote to the naive field space \mathcal{M} , and points at infinite distance will emerge (see figure 2). Around these points, the string couplings and the kinetic terms for the saxions coupling to the string will be connected, via (3.13), with the string RG flow towards the UV, which dictates the asymptotic behaviour of the system.

4 Strings, instantons and asymptotic limits

The definition of EFT string implies a powerful statement about non-perturbative corrections in regions of the EFT field space \mathcal{M} with approximate axionic shift symmetries. In this section we analyse the interplay between instanton effects and string flows, specially in those cases where several axionic symmetries coexist. The underlying $\mathcal{N} = 1$ framework reveals a conical structure for the set of asymptotic limits of infinite distance, together with a discrete cone of EFT string charges generating them. This strengthens the correspondence between infinite distance asymptotic limits in \mathcal{M} and EFT string solutions, and motivates some of the swampland criteria that will be proposed in the next section. The reader not interested in the general structure of EFT string charges may skip this material and jump to section 5. Most of the definitions that emerge from the following analysis are summarised by table 1 and appendix A.1.

4.1 Strings, instantons and cones

In a region of \mathcal{M} with approximate axionic symmetries in the field space metric, we may reach perturbative asymptotic regimes by taking limits in which linear combinations of saxions s^i become very large. Indeed, notice that non-perturbative contributions coming from $\frac{1}{2}$ BPS instantons charged under the axionic symmetries are a sum of terms of the form

$$\mathcal{O}(e^{2\pi \mathrm{i} m_i t^i}), \qquad m_i \in \mathbb{Z}.$$
(4.1)

Therefore, along paths of infinite distance like (3.1) with $\sigma \to \infty$, several of these effects become negligible. Because these are typically the leading non-perturbative corrections, we expect the strength of other non-perturbative effects to become negligible as well. In particular, an approximate shift symmetry that corresponds to the discrete shift (2.11) should become exact if all non-perturbative effects such that

$$\langle \mathbf{m}, \mathbf{e} \rangle = m_i e^i \neq 0 \tag{4.2}$$

asymptotically vanish in that limit. In other words, if e^i correspond to the charges of an EFT string, all non-perturbative effects satisfying (4.2) should die off along the saxionic string flow (2.13a) towards $r \to 0$, as otherwise the axionic shift symmetry will be broken.

In general, we can consider the instanton charges m^i as components of an element \mathbf{m} of a lattice $M_{\mathbb{Z}}$ of rank n = #(s) axions, dual to the lattice $N_{\mathbb{Z}}$ in which the string charges \mathbf{e} take values. Similarly, the chiral fields t^i can be identified with a vector $\mathbf{t} \in N_{\mathbb{C}} \equiv N_{\mathbb{Z}} \otimes \mathbb{C}$, subject to periodic identifications $\mathbf{t} \simeq \mathbf{t} + \mathbf{n}$ with $\mathbf{n} \in N_{\mathbb{Z}}$. One then expects to be able to identify, for each perturbative regime associated with a chiral sector \mathbf{t} , a set of chiral operators $\mathcal{O}_{\mathbf{m}} \equiv e^{2\pi i \langle \mathbf{m}, t \rangle}$ which respect the axionic periodicities and are exponentially suppressed as we proceed along the EFT string flows. Such operators are natural chiral observables on the asymptotic region and in concrete string theory models they typically enter the EFT as $\frac{1}{2}$ BPS instanton corrections. In this sense \mathbf{m} represents the corresponding 'instanton charges'.

From a purely EFT viewpoint, we can make the above ideas more precise by specifying a perturbative regime in the following way. For each region with perturbative axionic symmetries in the field space metric, we assume that the breaking of the said symmetry is measured by a set of instanton charges $C_{\rm I} \subset M_{\mathbb{Z}}$, which specifies the following set of asymptotic chiral observables:⁵

$$\mathcal{O}_{\mathbf{m}}(x) \equiv e^{2\pi i \langle \mathbf{m}, t(x) \rangle} \qquad \mathbf{m} \in \mathcal{C}_{\mathbf{I}}.$$
(4.3)

The perturbative region is then identified by requiring that $\mathcal{O}_{\mathbf{m}}(x)$ is exponentially suppressed for any $\mathbf{m} \in \mathcal{C}_{\mathbf{I}}$. Since $|\mathcal{O}_{\mathbf{m}}| = e^{-2\pi \langle \mathbf{m}, s \rangle}$, we get the identification

perturbative region:
$$\langle \mathbf{m}, \mathbf{s} \rangle \gg 1 \qquad \forall \mathbf{m} \in \mathcal{C}_{\mathrm{I}},$$
 (4.4)

where $s \equiv \text{Im} t \in N_{\mathbb{R}}$ collectively denotes the saxions s^i . This means that the asymptotic region (4.4) can be identified as the *deep interior* of a saxionic cone Δ , which is defined as follows:

$$\Delta \equiv \{ \boldsymbol{s} \in N_{\mathbb{R}} \mid \langle \mathbf{m}, \boldsymbol{s} \rangle > 0, \ \forall \mathbf{m} \in \mathcal{C}_{\mathrm{I}} \}.$$
(4.5)

That is, we characterise Δ as the interior of the cone $C_{\mathbf{I}}^{\vee}$.⁶ It is easy to see that Δ is convex: if s belongs to Δ , then λs also does for any $\lambda > 0$, and if $s_1, s_2 \in \Delta$ then $s_1 + s_2 \in \Delta$.

 $^{{}^{5}}$ The operators (4.3) may transform under an additional duality group, which should be taken into account in order to construct duality invariant observables. We will come back to this point in section 4.4.

⁶We recall that in general, given a vector space V and its dual V^* , for any set $S \subset V$ one can define the (closed convex) cone $S^{\vee} \equiv \{\eta \in V^* | \langle \eta, v \rangle \ge 0, \forall v \in S\} \subset V^*$.

Another necessary property of this perturbative region is that any asymptotic limit inside Δ such that $\langle \mathbf{m}, s \rangle$ diverges to $+\infty$ for some choices of $\mathbf{m} \in C_{\mathrm{I}}$, while it remains constant for the remaining ones, is an infinite distance limit. This condition clearly regards the Kähler structure of the field space, and not just the holomorphic one, and is necessary to identify (4.4) with a proper physical perturbative region in which the axionic symmetries are perturbatively preserved.

Reciprocally, in terms of Δ we can define C_I as the discretisation of the cone Δ^{\vee} dual to Δ :

$$\mathcal{C}_{\mathrm{I}} = \Delta^{\vee} \cap M_{\mathbb{Z}} \,. \tag{4.6}$$

If $\mathbf{m} \in \mathcal{C}_{I}$ then, necessarily, $-\mathbf{m} \notin \mathcal{C}_{I}$. Hence, \mathcal{C}_{I} has the structure of a discrete convex cone.

We will see that this characterisation of a given perturbative regime is quite general and applies to all the string theory examples that will be analysed in section 6. In particular, the elements of $C_{\rm I}$ are typically associated with $\frac{1}{2}$ BPS instantonic brane configurations appearing in the UV completion of the EFT. As a prototypical example, the reader can keep in mind the heterotic Calabi-Yau compactifications that will be considered in subsections 4.2 and 6.1. In the large volume regime, the saxionic cone Δ contains the Kähler cone, whose corresponding set $C_{\rm I}$ can be identified with the Mori cone of effective curves, wrapped by world-sheet instantons. As one approaches the boundary of the saxionic cone one expects non-perturbative corrections to become relevant and then the perturbative description to break down, as it clearly happens in the heterotic Kähler cone example.

Notice, however, that the correspondence between points in $C_{\rm I}$ and microscopic BPS instantons is in general more subtle. For instance, there may be walls of marginal stability, across which certain instantons cease to be BPS. In such cases, we will require that microscopic instantons in $C_{\rm I}$ should be at least asymptotically BPS, along any asymptotic limit within Δ . Furthermore, the existence of at least one microscopic BPS instanton (or multi-instanton) for *each* point in $C_{\rm I}$ — the "BPS completeness" of the instanton sector is expected but in general not obvious. We will not find any explicit counterexample to this expectation, but a more systematic study of this important question would be worthwhile. Keeping these possible caveats in mind, we will refer to $C_{\rm I}$ as the set of BPS instanton charges relevant for Δ .

The next step is to compare Δ with the lattice of string charges $N_{\mathbb{Z}}$. In each perturbative regime associated with a saxionic cone Δ , for any charge $\mathbf{e} \in N_{\mathbb{Z}}$ we may formally write down a supersymmetric EFT contribution of the form (2.22), with tension $\mathcal{T}_{\mathbf{e}} = M_{\mathrm{P}}^2 |\langle \mathbf{e}, \boldsymbol{\ell} \rangle| \equiv M_{\mathrm{P}}^2 |e^i \ell_i|$, where $\boldsymbol{\ell} \in M_{\mathbb{R}}$ is the vector of dual saxions ℓ_i , and which is a good approximation of the string tension in the perturbative regime (4.4). In this description, two charges \mathbf{e} and \mathbf{e}' preserve the same $\frac{1}{2}$ of supersymmetry if and only if $\langle \mathbf{e}, \boldsymbol{\ell} \rangle$ and $\langle \mathbf{e}', \boldsymbol{\ell} \rangle$ have the same sign. In the following we will dub as $\frac{1}{2}$ BPS strings those that correspond to the choice $\langle \mathbf{e}, \boldsymbol{\ell} \rangle > 0$, for which $\mathcal{T}_{\mathbf{e}} = M_{\mathrm{P}}^2 \langle \mathbf{e}, \boldsymbol{\ell} \rangle$, while the opposite choice will be dubbed as anti-BPS strings.

Now, in our 4d EFT description, a string charge **e** is associated to a flow in the saxionic variables. If the string solution is $\frac{1}{2}$ BPS one expects that $\langle \mathbf{e}, \boldsymbol{\ell} \rangle > 0$ all along an EFT string flow, as this corresponds to a positive linear energy density (2.33) at different

scales $\Lambda = 1/r$ and with different flow parameters (\mathbf{s}_0, r_0) . Since by varying the latter one can cover the saxionic cone Δ with each EFT string flow, one would expect that a set of EFT string charges that are mutually BPS have a positive tension for each charge and at each point of Δ .

This observation motivates the definition of a distinguished set C_S of string charges **e** for which $\langle \mathbf{e}, \boldsymbol{\ell} \rangle$ is positive for any point of Δ . More precisely, one may express the saxionic cone in terms of dual variables as

$$\mathcal{P} \equiv \left\{ \boldsymbol{\ell} \in M_{\mathbb{R}} \mid \ell_i = -\frac{1}{2} \left. \frac{\partial K}{\partial s^i} \right|_{\boldsymbol{s} \in \Delta} \right\} \,, \tag{4.7}$$

which is nothing but the image of Δ under a Legendre transform, and is well-defined for a perturbative Kähler potential K displaying the axionic shift symmetries associated with Δ . Then $C_{\rm S}$ can be identified with the discretisation of the cone dual to \mathcal{P} :

$$\mathcal{C}_{\mathrm{S}} \equiv \mathcal{P}^{\vee} \cap N_{\mathbb{Z}} \,. \tag{4.8}$$

The charges in $C_{\rm S}$ correspond to potential BPS strings preserving the same $\frac{1}{2}$ of supersymmetry all over Δ or, dually, \mathcal{P} . On the other hand, the charges in $-C_{\rm S}$ would correspond to anti-BPS strings preserving the opposite $\frac{1}{2}$ of supersymmetry.

Besides being BPS, our definition of EFT string requires that the string backreaction remains within the perturbative region (4.4) as we approach the core of the string. In general, this is not guaranteed for any $\mathbf{e} \in C_{\mathrm{S}}$. To see this, let us first rewrite the saxionic flow (2.13a) as

$$\mathbf{s} = \mathbf{s}_0 - \frac{1}{2\pi} \mathbf{e} \log \lambda \equiv \mathbf{s}_0 + \sigma \mathbf{e}$$
 with $\lambda \equiv \frac{r}{r_0} \equiv e^{-2\pi\sigma}$. (4.9)

Consider a flow that starts in the perturbative region (4.4), that is $\langle \mathbf{m}, \mathbf{s}_0 \rangle \gg 1$ for any $\mathbf{m} \in \mathcal{C}_{\mathrm{I}}$. The behaviour of chiral operators (4.3) along the flow (4.9) is

$$|\mathcal{O}_{\mathbf{m}}| = e^{-2\pi \langle \mathbf{m}, \mathbf{s} \rangle} = e^{-2\pi \langle \mathbf{m}, \mathbf{s}_0 \rangle} \times \lambda^{\langle \mathbf{m}, \mathbf{e} \rangle} .$$
(4.10)

If $\mathbf{e} \in \overline{\Delta} \equiv \text{closure of } \Delta$, then $\langle \mathbf{m}, \mathbf{e} \rangle \geq 0$ for any $\mathbf{m} \in C_1$. If $\langle \mathbf{m}, \mathbf{e} \rangle > 0$, then all instanton effects in C_1 asymptotically die off as we take the limit $\lambda \to 0$, and if $\langle \mathbf{m}, \mathbf{e} \rangle = 0$, they remain constant but still arbitrarily small for $\langle \mathbf{m}, \mathbf{s}_0 \rangle$ large enough. In both cases the flow remains in the region (4.4) for any $\lambda \in (0, 1]$ and by assumption reaches the infinite distance boundary at the string core $\lambda = 0$. Such a charge \mathbf{e} represents an EFT string, as defined in (2.34).

If instead $\mathbf{e} \in \mathcal{C}_{\mathrm{S}} - \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$ then there must be an instanton charge $\mathbf{m} \in \mathcal{C}_{\mathrm{I}}$ such that $\langle \mathbf{m}, \mathbf{e} \rangle < 0$. The associated non-perturbative effects grow along the string flow of charges \mathbf{e} , until they reach $\mathcal{O}(1)$ contributions at a finite radial distance

$$\lambda_* = \frac{r_*}{r_0} = e^{-\frac{2\pi \langle \mathbf{m}, \mathbf{s}_0 \rangle}{|\langle \mathbf{m}, \mathbf{e} \rangle|}}.$$
(4.11)

At this point we have that $\langle \mathbf{m}, \mathbf{s}(r_*) \rangle = 0$, which means that $\mathbf{s}(r_*)$ necessarily belongs to the finite field distance boundary of $\overline{\Delta}$. Hence, for charges $\mathbf{e} \in \mathcal{C}_{\mathrm{S}} - \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$, the weaklycoupled EFT description breaks down along the string flow, due to some BPS instanton operator (4.3) that becomes relevant and signals a non-perturbative regime. In this sense, we refer to strings corresponding to $\mathbf{e} \in C_{\mathrm{S}} - C_{\mathrm{S}}^{\mathrm{EFT}}$ either as strongly coupled or non-EFT.

Therefore, we may identify the set of EFT strings associated with a given perturbative regime with the following lattice cone

$$\mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}} = \overline{\Delta} \cap N_{\mathbb{Z}} \,. \tag{4.12}$$

Notice that to arrive at this definition we have not imposed that $\mathcal{T}_{\mathbf{e}} > 0$ along the flow, although as already mentioned in section 3 we expect this to be the case for any EFT string. In other words, we are lead to conclude that for a healthy EFT we must also have

$$\mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}} \subset \mathcal{C}_{\mathrm{S}} \quad \Leftrightarrow \quad \overline{\Delta} \subset \mathcal{P}^{\vee} \,. \tag{4.13}$$

As we will see, this condition is indeed satisfied for any string model considered in section 6.

Similarly to the case of instantons, these EFT definitions could be challenged by their microscopic realisation. Indeed, in all string models that we will encounter the potential $\frac{1}{2}$ BPS string charges $\mathbf{e} \in C_{\mathrm{S}}$ admit a microscopic realisation in terms of wrapped branes. The BPSness of such brane configurations can in fact depend on additional UV information invisible to the EFT and experience the presence of walls of marginal stability, that could be crossed along a string flow. Needless to say, these general issues are crucial for understanding the possible BPS completeness of C_{S} , and we will not attempt to exhaustively address them in the present paper. However, all the examples that we will consider neatly suggest that these UV issues are in fact absent for the subsector of EFT string charges $C_{\mathrm{S}}^{\mathrm{EFT}}$.

One can see that if K takes the form (3.8) with P homogeneous in the saxions, then \mathcal{P} is itself a cone. This is what will happen in all the examples that we will consider. We will also see that \mathcal{P} can take different shapes, but by using (3.8) it is easy to see that the infinite distance limit obtained by homogeneously rescaling all saxions always converge to the tip of \mathcal{P} . This means that all BPS string tensions vanish in this limit. More generically, EFT string flows converge to infinite distant points on the boundary \mathcal{P} . In particular, we have seen above that for a general Kähler potential of the form (3.8) the tension $\mathcal{T}_{\mathbf{e}} = M_{\mathrm{P}}^2 \langle \mathbf{e}, \boldsymbol{\ell} \rangle$ vanishes asymptotically as

$$\mathcal{T}_{\mathbf{e}} \sim \frac{M_{\mathrm{P}}^2}{\sigma}, \qquad \text{for } \sigma \to \infty,$$

$$(4.14)$$

along the flow (4.9). Hence, it will end on the subset

$$\mathcal{F}_{\mathbf{e}} = \overline{\mathcal{P}} \cap H_{\mathbf{e}}, \quad \text{with} \quad H_{\mathbf{e}} \equiv \{ \boldsymbol{\alpha} \in M_{\mathbb{R}} \mid \langle \boldsymbol{\alpha}, \mathbf{e} \rangle = 0 \}.$$
 (4.15)

In all the examples that we will consider, if $\mathbf{e} \in \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$ then $\mathcal{F}_{\mathbf{e}}$ will be an infinite distance boundary face of \mathcal{P} . $\mathcal{F}_{\mathbf{e}}$ can be codimension-one, but can also have higher codimension, up to the maximal one, corresponding to $\mathcal{F}_{\mathbf{e}}$ being just the tip of the cone \mathcal{P} . The codimension of $\mathcal{F}_{\mathbf{e}}$ counts the number of linearly independent string charges \mathbf{e}' whose (probe) tensions $T_{\mathbf{e}'} = M_{\mathrm{P}}^2 \langle \mathbf{e}', \boldsymbol{\ell} \rangle$ vanish asymptotically along the flow. We will say that the string flow

EFT data	Cone
Saxionic cone	$\Delta = \{ \boldsymbol{s} \in N_{\mathbb{R}} \mid \langle \mathbf{m}, \boldsymbol{s} \rangle > 0 , \forall \mathbf{m} \in \mathcal{C}_{\mathrm{I}} \}$
EFT strings	$\mathcal{C}^{\scriptscriptstyle\mathrm{EFT}}_{\mathrm{S}} = \overline{\Delta} \cap N_{\mathbb{Z}}$
BPS instantons	$\mathcal{C}_{\mathrm{I}} = \Delta^{\vee} \cap M_{\mathbb{Z}}$
Dual saxionic cone	$\mathcal{P} = \left\{ \boldsymbol{\ell} \in M_{\mathbb{R}} \mid \ell_i = -\frac{1}{2} \left. \frac{\partial K}{\partial s^i} \right _{\boldsymbol{s} \in \Delta} \right\}$
EFT instantons	$\mathcal{C}_{\mathrm{I}}^{\scriptscriptstyle\mathrm{EFT}}=\overline{\mathcal{P}}\cap M_{\mathbb{Z}}$
BPS strings	$\mathcal{C}_{\mathrm{S}} = \mathcal{P}^{ee} \cap N_{\mathbb{Z}}$

Table 1. EFT conical structures induced by strings and instantons.

degeneracy is of order p if it ends on a codimension-p face $\mathcal{F}_{\mathbf{e}}$. We will also refer to orderone flows as *non-degenerate*, while flows of higher degeneracy will be dubbed *degenerate*.⁷

Finally, we can also define the instantonic analogue of (4.12) as follows:

$$\mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}} \equiv \overline{\mathcal{P}} \cap M_{\mathbb{Z}} \quad \subset \quad \mathcal{C}_{\mathrm{I}} \,, \tag{4.16}$$

where the inclusion in $C_{\rm I}$ follows from (4.6). In appendix E it is shown how, analogously to what happens for EFT strings, the backreaction of instantons of charges $\mathbf{m} \in C_{\rm I}^{\rm EFT}$ generate acceptable EFT flow solutions, while the BPS instantons of charges $\mathbf{m} \in C_{\rm I} - C_{\rm I}^{\rm EFT}$ do not. We will dub these two classes as EFT and non-EFT instantons, respectively. Interestingly, only non-EFT BPS instantons can generate the finite distance strong coupling effects along the flow of non-EFT strings discussed around (4.11). This follows immediately from (4.8) and $C_{\rm I}^{\rm EFT} \subset \mathcal{P}$, which imply that $C_{\rm S} \subset (C_{\rm I}^{\rm EFT})^{\vee}$. That is, if $\mathbf{m} \in C_{\rm I}^{\rm EFT}$, then $\langle \mathbf{m}, \mathbf{e} \rangle \geq 0$ for any $\mathbf{e} \in C_{\rm S}$.

We have summarised all the relevant definitions of the conical structures introduced above in table 1.

Notice that, in terms of the natural paring between string and instanton charges, the definition (4.12) of EFT string could be stated as

$$\mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}} = \{ \mathbf{e} \in N_{\mathbb{Z}} | \langle \mathbf{m}, \mathbf{e} \rangle \ge 0 \quad \forall \mathbf{m} \in \mathcal{C}_{\mathrm{I}} \}.$$

$$(4.17)$$

Physically, given a string of charge **e** and instanton of charge **m**, the pairing $\langle \mathbf{m}, \mathbf{e} \rangle$ gives the magnetic charge of the instanton under the two-form gauge field $\mathcal{B}_2^{\mathbf{e}} \equiv e^i \mathcal{B}_{2,i}$ elec-

⁷If the Kähler potential takes the form (3.8), we can also associate the degeneracy of the flow to the properties of the homogeneous function P(s). Following [8], one can divide the perturbative region into different growth sectors corresponding to a different ordering $s_1 \gg s_2 \gg \ldots$ characterising what saxions grow faster. Within each growth sector, the function P(s) will be approximated by a single monomial dominating asymptotically, which can be parametrised as $P(s) \propto s_1^{n_1} s_2^{n_2-n_1} \ldots s_i^{n_i-n_{i-1}} + \ldots$ If we send for instance $s_i \to \infty$, $i = 1, 2, \ldots j$, then n_j is the singularity type which is equivalent to sum over all the exponents of the saxions taken to the large field limit. If P(s) is approximated by the same monomial for all the growth sectors, the perturbative regime only contains non-degenerate flows. Otherwise, it will contain degenerate flows. This latter case is associated to infinite distance limits in which the singularity type does not increase in the enhancement chain, so $n_i - n_{i-1} = 0$ and some saxions are absent in some of the monomials.

trically sourced by the string. Indeed, as discussed in more detail in appendix E — see equation (E.3) — this magnetic charge is measured by the flux

$$\frac{1}{2\pi} \int_{S^3} \mathcal{H}_3^{\mathbf{e}} = \frac{1}{2\pi} e^i \int_{S^3} \mathcal{H}_{3,i} = \langle \mathbf{m}, \mathbf{e} \rangle, \qquad (4.18)$$

with $\mathcal{H}_3^{\mathbf{e}} = \mathrm{d}\mathcal{B}_2^{\mathbf{e}}$ and S^3 a three-sphere surrounding the instanton. One may then reformulate (4.17) as follows:

A BPS string belongs to $C_{\rm S}^{\rm EFT}$ whenever all instantons in $C_{\rm I}$ carry non-negative magnetic charge under the two-form field $\mathcal{B}_2^{\rm e}$ that couples to the string.

Dually, we can also interpret the pairing $\langle \mathbf{m}, \mathbf{e} \rangle$ as the axionic charge of the string corresponding to the axion that couples 'electrically' to the instanton: if $a^{\mathbf{m}} \equiv m_i a^i$ denotes this axion, then $a^{\mathbf{m}} \to a^{\mathbf{m}} + \langle \mathbf{m}, \mathbf{e} \rangle$ around the string. We can then analogously say that a BPS instanton belongs to $C_{\mathrm{I}}^{\mathrm{EFT}}$ if and only if all strings in C_{S} carry non-negative axionic charge under the axion $a^{\mathbf{m}}$ that couples to the instanton.

4.2 A simple example

Let us consider heterotic strings compactified in a Calabi-Yau three-fold X and focus on the large-volume, weak string coupling perturbative regime and the axionic symmetries that arise in this region of moduli space. The relevant $\frac{1}{2}$ BPS instantons arise from worldsheet instantons wrapping holomorphic curves on X, and from NS5-branes wrapping X. Reversing their orientation, one obtains the corresponding set of anti-instantons. The cone of $\frac{1}{2}$ BPS instanton charges $C_{\rm I}$ is thus generated by effective curve classes, together with the NS5-brane charge. Similarly, mutually $\frac{1}{2}$ BPS 4d strings arise from NS5-branes wrapped on effective divisors, together with fundamental strings which are point-like in X. Their corresponding cohomology classes generate the discrete cone $C_{\rm S}$.⁸ $C_{\rm I}$ and $C_{\rm S}$ are subsets of the dual lattices $M_{\mathbb{Z}} \simeq H_0(X, \mathbb{Z}) \oplus H_4(X, \mathbb{Z})/(\text{torsion})$ and $N_{\mathbb{Z}} \simeq H^6(X, \mathbb{Z}) \oplus$ $H^2(X, \mathbb{Z})/(\text{torsion})$, respectively.

The saxionic cone Δ corresponding to this perturbative regime is parametrised by the 4d dilaton and the Kähler moduli. More precisely, the saxions are given by $s^i = (s^0, s^a)$. The Kähler saxions s^a arise from the decomposition of the (string frame) Kähler form

$$J = s^a[D_a], (4.19)$$

in an integral basis $[D_a]$ Poincaré dual to a set of divisors D_a . J takes value in the Kähler cone $\mathcal{K}(X)$, while the universal saxion s^0 is given by

$$s^{0} = \frac{1}{3!} e^{-2\phi} \kappa_{abc} s^{a} s^{b} s^{c} = e^{-2\phi} V_{X} , \qquad (4.20)$$

where ϕ is the 10d dilaton, $\kappa_{abc} \equiv D_a \cdot D_b \cdot D_c$ are the triple intersection numbers of X in this basis, and V_X is the volume of X in string units. So, the relevant saxionic cone is

$$\Delta = \mathbb{R}_{>0} \oplus \mathcal{K}(X) \,, \tag{4.21}$$

where $\mathbb{R}_{>0}$ is parametrised by s^0 .

⁸To be precise, one should take into account F1 charges induced by world-volume flux and curvature corrections on the NS5-brane, which may translate into slightly different generators for the charge lattices. In the following we will ignore this effect, which plays no significant role in the discussion.

The closure of the Kähler cone is the cone $\operatorname{Nef}^1(X)$ generated by nef divisors — see appendix A.2 for a summary of the relevant terminology. Applying our general definition (4.12) to the present example we then see that $\mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$ is generated by the nef divisor classes and $H_0(X, \mathbb{Z})_+ \simeq \mathbb{Z}_{>0}$. So, to sum up, we have the following sets of BPS charges

$$\mathcal{C}_{S} = \{F1 + NS5\text{-branes on effective divisors}\} \simeq \mathbb{Z}_{>0} \oplus Eff^{1}(X)_{\mathbb{Z}}, \qquad (4.22a)$$

$$\mathcal{C}_{\mathrm{I}} = \{ \mathrm{NS5\text{-}branes on } X + \mathrm{F1 on effective curves} \} \simeq \mathbb{Z}_{\geq 0} \oplus \mathrm{Eff}_1(X)_{\mathbb{Z}}, \qquad (4.22\mathrm{b})$$

$$\mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}} = \{\mathrm{F1} + \mathrm{NS5}\text{-branes on nef divisors}\} \simeq \mathbb{Z}_{\geq 0} \oplus \mathrm{Nef}^{1}(X)_{\mathbb{Z}}.$$
(4.22c)

The pairing $\langle \mathbf{m}, \mathbf{e} \rangle$ between C_{I} and C_{S} just corresponds to the geometrical intersection number. Notice that indeed, by definition of nef divisors, we can identify $\overline{\Delta}$ as the cone dual to the cone generated by C_{I} . As will be discussed in the next section an EFT string completeness hypothesis follows as a direct consequence of Conjecture 1. In this setup, such a hypothesis implies that every nef divisor admits an effective representative. This mathematical property has already been conjectured in [40], for similar underlying reasons. See subsection 4.3 for the connection between our setup and the one in [40].

In our perturbative regime, the leading contribution to the Kähler potential for the above saxions reads

$$K = -\log s^0 - \log \left(\frac{1}{3!}\kappa_{abc}s^as^bs^c\right) + \dots, \qquad (4.23)$$

which has indeed the form (3.8). So the dual saxions $\ell_i = (\ell_0, \ell_a)$ are given by

$$\ell_0 = \frac{1}{2s^0} = \frac{e^{2\phi}}{2V_X}, \qquad \ell_a = \frac{1}{4V_X} \kappa_{abc} s^b s^c.$$
(4.24)

It is not so simple to describe the structure of \mathcal{P} and $\mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}}$ in general terms, see (4.7) and (4.16). So let us for now focus on a specific example, which will allow us to better illustrate some features of these class of models. We will come back to the general discussion in subsection 4.4.

Let us take the Calabi-Yau studied in [57]. This three-fold can be regarded as an elliptic fibration over \mathbb{P}^2 , and is such that $b_2(X) = 2$. The cone of effective divisors $\text{Eff}^1(X)$ is generated by two divisors B and L: B is the image of the \mathbb{P}^2 base under the global section of the elliptic fibration, while L is the 'vertical' divisor obtained by pulling back the base $\mathbb{P}^1 \subset \mathbb{P}^2$ by using the fibration projection. The nef cone $\text{Nef}^1(X)$ is instead generated by L and the 'horizontal' divisor $H \equiv B + 3L$. In the nef basis, the intersection numbers are summarised by the formal polynomial

$$\mathcal{I}_X = 9H^3 + 3H^2L + HL^2, \qquad (4.25)$$

where the coefficient of each monomial gives the value of the corresponding triple intersection. The dual Mori cone $\overline{\text{Eff}}_1(X)$ is generated by the effective curves

$$h = L^2$$
, $l = BL = HL - 3L^2$. (4.26)

The curve h can be identified with elliptic fibre, while l can be identified with the pushforward of a \mathbb{P}^1 in the base through the global section of the elliptic fibration.



Figure 3. Saxionic cones and their duals $\Delta \subset \mathcal{P}^{\vee}$ and $\mathcal{P} \subset \Delta^{\vee}$ of the two-Kähler-moduli heterotic model. Here we are suppressing the $\mathbb{R}_{>0}$ piece of the moduli space that corresponds to s^0 . The darker shaded regions correspond to the perturbative regime (4.4).

The Kähler cone is simplicial and is generated by the Poincaré dual classes [H] and [L]. We can then expand the (string frame) Kähler form as follows

$$J = s^{1}[H] + s^{2}[L] \in H^{2}(X, \mathbb{R}), \qquad (4.27)$$

where the Kähler saxions $s^1, s^2 \in \mathbb{R}_{>0}$ measure the volumes of the curves h and l respectively and, together with (4.20), parametrise Δ . The Kähler potential (4.23) takes the form

$$K = -\log s^0 - \log \left[3(s^1)^3 + 3(s^1)^2 s^2 + s^1 (s^2)^2 \right] + \dots , \qquad (4.28)$$

and the dual saxions are

$$\ell_{0} = \frac{1}{2s^{0}},$$

$$\ell_{1} = \frac{9(s^{1})^{2} + 6s^{1}s^{2} + (s^{2})^{2}}{6(s^{1})^{3} + 6(s^{1})^{2}s^{2} + 2s^{1}(s^{2})^{2}},$$

$$\ell_{2} = \frac{3(s^{1})^{2} + 2s^{1}s^{2}}{6(s^{1})^{3} + 6(s^{1})^{2}s^{2} + 2s^{1}(s^{2})^{2}}.$$
(4.29)

With these data, one can easily describe each of the continuous cones defined above:

$$\Delta = \{ (s^0, s^1, s^2) \in \mathbb{R}^3 | s^0, s^1, s^2 > 0 \},$$
(4.30a)

$$\mathcal{P} = \left\{ (\ell_0, \ell_1, \ell_2) \in \mathbb{R}^3 | \ell_0 > 0, \ell_1 > 3\ell_2 > 0 \right\},$$
(4.30b)

see figure 3, which also shows in darker colour the perturbative region (4.4) both in Δ and in \mathcal{P} . The corresponding discrete cones of charges are:

$$\mathcal{C}_{\rm I} = \{ (m_0, m_1, m_2) \in \mathbb{Z}^3 | m_0, m_1, m_2 \ge 0 \} \simeq \langle \text{NS5}, h, l \rangle , \qquad (4.31a)$$

$$\mathcal{C}_{\rm S} = \{ (e^0, e^1, e^2) \in \mathbb{Z}^3 | e^0, e^1 \ge 0, 3e^1 + e^2 \ge 0 \} \simeq \langle {\rm F1}, B, L \rangle , \qquad (4.31b)$$

$$\mathcal{C}_{\rm S}^{\rm EFT} = \{ (e^0, e^1, e^2) \in \mathbb{Z}^3 | e^0, e^1, e^2 \ge 0 \} \simeq \langle {\rm F1}, H, L \rangle , \qquad (4.31c)$$

$$\mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}} = \{ (m_0, m_1, m_2) \in \mathbb{Z}^3 | m_0 \ge 0, m_1 \ge 3m_2 \ge 0 \} \simeq \langle \mathrm{NS5}, h, l+3h \rangle, \qquad (4.31\mathrm{d})$$



Figure 4. BPS and EFT string and instanton lattice cones $C_{\rm S}^{\rm EFT} \subset C_{\rm S}$ and $C_{\rm I}^{\rm EFT} \subset C_{\rm I}$ of the two-moduli heterotic model.

as illustrated in figure 4. Note that C_S^{EFT} is dual to C_I , as it happens in general, and that also C_I^{EFT} is the dual of C_S . In particular, C_S contains NS5-instantons wrapping effective divisors and, dually [58], C_I^{EFT} contains F1-instantons wrapping 'movable' curves — see appendix A.2. The general differences between the EFT and non-EFT strings and instantons have been discussed in general terms in subsection 4.1 and in appendix E, respectively. We now illustrate them in this specific example, focusing on the string sector.

Consider first the EFT sector $C_{\rm S}^{\rm EFT}$, which is generated by the charges $\mathbf{e}_{\rm F1} = (1, 0, 0)$, $\mathbf{e}_H = (0, 1, 0)$ and $\mathbf{e}_L = (0, 0, 1)$. The corresponding strings are *elementary*, in the sense that they cannot be decomposed as superposition of other BPS strings. The flows (4.9) corresponding to $\mathbf{e}_{\rm F1}$ describes a zero coupling limit, in which the string coupling goes to zero, while the string frame internal metric remains constant. From the viewpoint of \mathcal{P} , these flows end on the codimension-one face of \mathcal{P} defined by $\ell_0 = 0$, which is precisely $\mathcal{F}_{\mathbf{e}_{\rm F1}}$ as defined in (4.15). On this face, precisely the strings with charges multiple of $\mathbf{e}_{\rm F1}$ become tensionless and the flow is non-degenerate. Also the string flows associated with \mathbf{e}_L are non-degenerate, as they end on the codimension-one face $\mathcal{F}_{\mathbf{e}_L} \equiv \overline{\mathcal{P}} \cap \{\ell_2 = 0\}$, at which only the strings of charge proportional to \mathbf{e}_L become tensionless. Instead, something different happens for the EFT flows generated by a string of charge \mathbf{e}_H . Indeed, in this case $s^1 \sim \sigma \to \infty$ and then from (4.29) it is clear that both ℓ_1 and ℓ_2 vanish asymptotically along the flow. Hence, both elementary strings become tensionless at the same rate:

$$\mathcal{T}_{\mathbf{e}_H} \simeq \frac{3}{2\sigma} M_P^2 \,, \quad \mathcal{T}_{\mathbf{e}_L} \simeq \frac{1}{2\sigma} M_P^2 \,, \tag{4.32}$$

as we approach $\mathcal{F}_{\mathbf{e}_H}$ along the flow. In other words, these flows end on the codimension-2 face $\mathcal{F}_{\mathbf{e}_H}$ and are degenerate of order-two.

Let us now consider the flow generated by the non-EFT charge $\mathbf{e}_B = (0, 1, -3)$. The string flow (2.13a) reads

$$s^{1} = s_{0}^{1} - \frac{1}{2\pi} \log \frac{r}{r_{0}}, \qquad s^{2} = s_{0}^{2} + \frac{3}{2\pi} \log \frac{r}{r_{0}}.$$
 (4.33)



Figure 5. Examples of saxionic string flows (4.9). On the left, string flows in the saxionic cone Δ (the s^0 direction is suppressed) and, on the right, their maps in the dual saxionic cone \mathcal{P} (the ℓ_0 direction is suppressed). The charges $\mathbf{e}_H, \mathbf{e}_L, \mathbf{e}_B$ are elementary, while $\hat{\mathbf{e}} = (0, 1, 3)$ is not. The picture also shows that the \mathbf{e}_B flow approaches the finite distance boundary $s^2 = 0$ of Δ and, correspondingly, $\mathcal{F}_{\mathbf{e}_B}$ lies in the strongly coupled light shaded region of $\overline{\mathcal{P}}$.

We then see that $s^2(r)$ flows towards strong coupling as $r \to 0$ and reaches a strongly coupled regime at $r_* = r_0 \exp\left(-\frac{2\pi s_0^2}{3}\right)$, at which $s^2 = 0$ and $s^1 = s_0^1 + \frac{1}{3}s_0^2$. Geometrically, in this limit the base \mathbb{P}^2 collapses to zero size while the elliptic fibre volume remains finite. Even though r_*/r_0 can be made exponentially small by choosing large enough s_0^2 , this behaviour is clearly opposite to the one of strings belonging to $C_{\rm S}^{\rm EFT}$. In the case of (4.33) one must fine-tune the initial value s_2^0 to be large enough, the corresponding RG flow does not correspond to a marginally relevant operator, and it does not flow to weak coupling as we approach the string. Furthermore, differently from strings with charges in $C_{\rm S}^{\rm EFT}$, the field-space distance travelled by the flow from r_0 to r_* is classically finite, as one can explicitly check by using (3.4) and (3.2), see also figure 5.

Finally, one can compute the behaviour of the \mathbf{e}_B -string tension around $r \simeq r_*$:

$$\mathcal{T}_{\mathbf{e}_B} = M_{\rm P}^2(\ell_1 - 3\ell_2) \simeq \frac{3M_{\rm P}^2 \left(s_0^2 - \frac{3}{2\pi}\log\frac{r_0}{r}\right)^2}{27(s_0^1)^3 + 27(s_0^1)^2 s_0^2 + 9s_0^1(s_0^2)^2 + (s_0^2)^3},\tag{4.34}$$

and conclude that it vanishes at $r = r_*$. In other words, we see that also the non-EFT \mathbf{e}_B -flow reaches a face $\mathcal{F}_{\mathbf{e}_B}$ of \mathcal{P} defined as in (4.15). However, $\mathcal{F}_{\mathbf{e}_B}$ is quite different from the faces corresponding to EFT strings, for two main reasons. First, $\mathcal{F}_{\mathbf{e}_B}$ is at *finite* field space distance. Second, according to the above arguments as we approach $\mathcal{F}_{\mathbf{e}_B}$ we expect strong non-perturbative corrections to modify (4.28) and break the axionic symmetry. In fact, the BPS instanton of charges $\mathbf{m}_l = (0, 0, 1)$ becomes unsuppressed along the flow, since $\langle \mathbf{m}_l, \mathbf{e}_B \rangle < 0$. This matches with the fact that the \mathbf{e}_B -string is non-EFT and therefore does not satisfy (4.17), showing a direct relation between non-EFT strings and non-EFT instantons in $\mathcal{C}_{\mathrm{I}} - \mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}}$. Let us also notice that such non-perturbative effects should in principle modify the flow (4.33) significantly, and also obstruct the possibility to use the

dual linear multiplet description in terms of two-form potentials \mathcal{B}_{2i} . As a consequence, the \mathbf{e}_B -string flow cannot be straightforwardly described within our perturbative EFT description. In particular, near $r = r_*$ the formula (4.34) is no longer reliable and the string may actually remain tensionful. Figure 5 provides a pictorial illustrations of the above flows, and also of a non-elementary EFT one.

4.3 Connection with the 5d viewpoint

Even though the physical properties of strings in four- and five-dimensional theories are drastically different, the characterisation (4.17) of the EFT string charges provides a direct link between our EFT strings and the *supergravity strings* of five-dimensional $\mathcal{N} = 1$ theories given in [40] — see eq. (2.31) therein. The connection between the two settings can be made more concrete by compactifying the 5d M-theory models of [40] to four dimensions on a segment à la Hořava-Witten [59, 60]. In this way one gets a strongly coupled $E_8 \times E_8$ four-dimensional heterotic model, in which supersymmetry is broken by the E_8 sectors supported at the end-points of the Hořava-Witten segment. In this strong string coupling limit, the heterotic NS5 strings uplift to M5-strings in M-theory. In particular, the 4d EFT NS5 strings in (4.22c) uplift to the supergravity strings of [40], which are indeed associated with nef divisors. Correspondingly, the heterotic F1-instantons in (4.22b) uplift to BPS particles in the 5d M-theory models, which by definition must have non-negative pairing with the supergravity strings, as in (4.17).

Notice that, on the other hand, the heterotic F1 strings in $C_{\rm S}^{\rm EFT}$ do not uplift to M-theory strings. Rather, they uplift to M2-branes stretched along the Hořava-Witten segment, which then become membranes in 5d in the extreme decompactification limit. Correspondingly, the heterotic NS5-instantons do not uplift to 5d particles, but rather to M5-brane instantons. The correspondence between the 4d cones ($C_{\rm S}^{\rm EFT}$, $C_{\rm I}$) and 5d supergravity string and BPS particle charges is therefore not one-to-one, but their relation in terms of a non-negative pairing remains in both pictures.

In [40] it is argued that supergravity strings are strings which cannot survive a rigid limit. This is clear for the above heterotic/M-theory strings associated with nef effective divisors, which are then 'movable' along the entire internal space — see appendix A.2. Hence, these divisors are not shrinkable and they do not survive a decoupling of gravity, that is, a decompactification of the internal space. Remarkably, we will see that this property actually holds for all the EFT strings of the string/M-theory models that we will discuss in section 6. For instance, also the F1 strings in (4.22c) clearly satisfy this property, since in a decompactification limit they can be moved away from any local 4d physical sector surviving the decoupling of gravity.

4.4 Finiteness and conical structures

The heterotic model considered in subsection 4.2 exhibits some key features of the broad class of models covered by subsection 4.1. However, some of its particular features are not realised in general. Therefore we end up this section by discussing some general features of the conical structures relevant for string flows.

First of all, the saxionic cone of our heterotic example is not only rational polyhedral (i.e. generated by a finite number of elements of $N_{\mathbb{Z}}$), but also simplicial, in the sense that its integral generators form a basis of $N_{\mathbb{R}}$. This property does not generically hold, since Δ may be rational polyhedral but not simplicial, or it may even not be rational polyhedral at all. For instance, let us consider the saxionic cone of a general heterotic compactification, which is given by (4.21) and therefore it includes the Kähler cone of a general Calabi-Yau. On the one hand, this Kähler cone can have a quite complicated structure [61] and in general does not satisfy these properties. On the other hand, the presence of an underlying rational polyhedral structure would appear natural from a physical viewpoint, as it would correspond to existence a finite number of appropriately quantised elementary EFT string charges generating all the other ones. In general, we dub the corresponding EFT flows as *elementary* flows.

In fact, this intuition can still be compatible with a non-rational polyhedral Δ , if there exists a 'perturbative' duality group $\mathcal{G}_{\mathbb{Z}}$ acting linearly on $N_{\mathbb{Z}}$ and preserving Δ , with a rational polyhedral fundamental domain. This property is indeed expected to hold for the nef cone of Calabi-Yau three-folds, as conjectured by Morrison in [62], see [63] for an explicit example. Morrison and Kawamata later extended this *cone conjecture* to the cone $\operatorname{Mov}^1(X)$ of movable divisors, see [64, 65] for the precise statements and [66] for a recent review.⁹ These conjectures admit a natural generalisation to a *saxionic cone conjecture* for our physical setting. Following [62, 64] we define Δ_+ as the convex hull of $\overline{\Delta} \cap N_{\mathbb{Q}}$, which is basically the cone obtained by eliminating from $\overline{\Delta}$ the irrational parts of its boundary. The saxionic cone conjecture can then be stated as follows:

Saxionic Cone Conjecture. for any saxionic cone Δ associated with the asymptotic region of an EFT compatible with quantum gravity, there exists a rational polyhedral cone $\Pi \subset \Delta_+$ such that the union of all its images under the duality group $\mathcal{G}_{\mathbb{Z}}$ covers Δ_+ . The polyhedral cone Π can be chosen to be the closure of a fundamental saxionic domain.

A non-trivial duality group $\mathcal{G}_{\mathbb{Z}}$ implies that the asymptotic sector of the moduli space should be identified with $\mathcal{M}_{pert} \equiv \mathcal{D}/\hat{\mathcal{G}}_{\mathbb{Z}}$, where $\mathcal{D} \equiv \{\mathbf{t} \in N_{\mathbb{R}} + i\Delta\}$ is the tube domain parametrised by the chiral fields t^i , and $\hat{\mathcal{G}}_{\mathbb{Z}} \equiv \mathcal{G}_{\mathbb{Z}} \ltimes N_{\mathbb{Z}}$ is the complete perturbative duality group, combining $\mathcal{G}_{\mathbb{Z}}$ and the group of axionic integral shifts, which can be identified with $N_{\mathbb{Z}}$. Correspondingly, the chiral observables should actually be given by $\mathcal{G}_{\mathbb{Z}}$ -invariant combinations of (possibly infinitely many) operators of the form (4.3), hence corresponding to multi-instanton effects. Adapting the arguments of [62, 64], the validity of the saxionic cone conjecture would also allow one to partially compactify the infinite distance sector of \mathcal{M}_{pert} by applying the general scheme of semi-toric local compactifications described in [68]. This would be the starting point to extend beyond the simplicial case the analysis made in [32] for the structure of the flux superpotential generated by EFT membranes [42] along the asymptotic limits associated with EFT strings.

In the heterotic example of section 4.2 the dual saxionic cone \mathcal{P} is simplicial, and then rational polyhedral, too. Again, this is not generically true, and actually \mathcal{P} can

⁹Some implications of the Morrison-Kawamata conjecture to the SDC have been recently discussed in [67], in the context of M-theory compactifications to five dimensions.
be non-polyhedral also in cases in which Δ is polyhedral. For instance, this happens in heterotic compactifications on Calabi-Yau spaces X in which at least a codimension-one face of the Kähler cone $\mathcal{K}(X)$ corresponds to a flopping contraction, according to the classification of [61]. In flopping contractions only curves collapse, while divisors do not. This implies that no NS5-strings become (classically) tensionless as we approach a flopping boundary wall of $\mathcal{K}(X)$, and then there is no corresponding codimension-one face (4.15) in the boundary of \mathcal{P} .

5 4d strings and swampland criteria

In the previous sections we have discussed how, in a given perturbative regime, saxions are driven towards infinite field distance trajectories along flows generated by EFT strings. As a result, each EFT string can be associated with an asymptotic perturbative limit in the moduli space, selected by the string charges. In this section, we will propose that the reverse is also true, namely that *all* infinite field distance limits can be realised as an EFT string flow. Moreover, building on this conjecture we will propose a specific asymptotic behaviour of the EFT maximal possible cut-off m_* , associated with the appearance of an infinite tower of new light states, along such infinite field distance trajectories. The evidence for this second conjecture arises from a plethora of string theory examples, as will be shown in section 6, and provides a very reduced set of possible values for the exponential rate of the SDC.

5.1 The distant axionic string conjecture

Consider a 4d EFT compatible with a consistent theory of quantum gravity. We conjecture that the following property is universally realised:

Conjecture 1 (Distant Axionic String Conjecture).

Every infinite field distance limit of a 4d EFT consistent with quantum gravity can be realised as an RG flow UV endpoint of an EFT string.

This statement should be understood in terms of the triple correspondence between EFT string RG flows, backreacted solutions and paths in field space \mathcal{M} described in section 3. By this correspondence, an EFT string of charge \mathbf{e} generates paths in \mathcal{M} of the form (3.11) when the cut-off Λ of the EFT is varied. The infinite distance point is approached as $\Lambda \to \infty$, which must be seen as a formal limit, since necessarily at some scale the 4d EFT description will stop being valid. Clearly, EFT string RG flows define specific paths in field space. In this sense, the above conjecture claims that they are sufficient to reach any infinite distance singularity.

Given the discussion carried in the previous sections, this proposal is quite natural in a minimally supersymmetric setting. Indeed, recall that an EFT string is a fundamental string magnetically charged under an axion satisfying (2.1) and exhibiting a perturbative continuous shift symmetry near its core. Moreover, all non-perturbative corrections breaking the shift symmetry should be suppressed near its core as defined in (2.34), which in turns implies a non-trivial positivity constraint on the allowed set of string charges, cf. eqs. (4.12) and (4.17). Therefore, an equivalent way to phrase Conjecture 1 is that all possible infinite distance regimes are associated with fundamental axionic symmetries. In such a case, by the Completeness Hypothesis¹⁰ we would expect that fundamental string operators with the corresponding magnetic charges appear in that regime of our EFT, and everything else follows from this.

The extension of the conjecture to non-supersymmetric setups is also quite natural if supersymmetry is spontaneously broken and restored at some scale Λ_{SUSY} below the EFT cut-off, following the observations at the end of section 3.2. If that is not the case, and if we nevertheless have an infinite-distance trajectory along which the quantum-corrected scalar potential is sufficiently flat, then Conjecture 1 becomes quite non-trivial. For instance, it would require the existence of an axion-like scalar that pairs up with the field describing the said trajectory, under which the string is magnetically charged. Notice that, in this sense, our conjecture is a sort of 4d reciprocal of Conjecture 4 of [1], for which every axion must come together with a radial mode (a saxion) to guarantee that the moduli space first homotopy group vanishes.

The Swampland Distance Conjecture (SDC) [1] predicts that along any infinite distance limit there appears an infinite tower of new light states whose lightest mass m_* decreases exponentially as $e^{-\alpha\Delta\phi}$ with the geodesic field distance $\Delta\phi$, for some constant α . From a purely EFT viewpoint, there is typically no way to identify neither such tower scale m_* nor the corresponding constant α without knowing the UV completion of the EFT. On the other hand, Conjecture 1 implies a distinctive physical way of classifying the infinite distance limits as flows (4.9) generated by EFT strings. The corresponding tensions, whose renormalisation is also dictated by the flow, then provide a natural cut-off scale which can be computed within the EFT. In certain cases, the strings themselves will provide the leading tower signaling the EFT breakdown, but in others, there will be additional towers of states getting light faster than the string. It is then natural to wonder what is the relation between \mathcal{T} and the leading tower scale m_* . The following proposal significantly restricts the range of possible answers to this question.

Conjecture 2 (Cut-off asymptotics).

Along an asymptotic limit specified by the RG flow of an EFT string, its tension \mathcal{T} goes to zero. The maximal EFT cut-off m_* then scales like

$$m_*^2 \simeq M_{\rm P}^2 A \left(\frac{\mathcal{T}}{M_{\rm P}^2}\right)^w$$
 for some positive integer $w = 1, 2, \dots$ (5.1)

with a coefficient A depending on the non-flowing chiral fields.

We will refer to the integer w as the *scaling weight* of the EFT string flow. In terms of the correspondence between RG flows and paths in field space \mathcal{M} , Conjecture 2 should

¹⁰More precisely, we have assumed the string to be BPS in the UV, so we are actually assuming a BPS version of the Completeness Hypothesis, as in [40]. It would be interesting to check if the BPS condition can be dropped.

be understood as follows. An EFT string of charge \mathbf{e} generates paths $\gamma_{\mathbf{e}} \subset \mathcal{M}$ through the EFT RG flow. As discussed in section 3.3, the variation of the effective tension $\mathcal{T}_{\mathbf{e}}$ with Λ in (3.9) describes how the probe string tension $\mathcal{T}_{\mathbf{e}}$ varies along $\gamma_{\mathbf{e}}$. The relation (5.1) then provides the asymptotic variation of the tower scale m_* along this same path in field space. While the conjecture could have been phrased in terms of the probe tension, and indeed in section 6 we will check the relation between the probe tension and m_* in different string theory setups, let us recall that it is $\mathcal{T}_{\mathbf{e}}$ the quantity that more directly enters the EFT, and so it makes more sense to describe our swampland criteria in terms of it.

Notice that one can always compute, within the EFT, the relation between the flow parameter σ and the field space distance travelled by the saxionic flow (4.9). As discussed in appendix C, in our scheme EFT string flows can be argued to be asymptotically geodesic. Hence, knowing the scaling weight of the string flow would allow one to compute the exponential rate α appearing in the SDC. Remarkably, in the string theory examples of section 6 we find that $A \simeq \mathcal{O}(1)$ for generic values of the non-flowing fields and only three possible values for the scaling weight, w = 1, 2, 3. Furthermore in all those examples we have observed the following intriguing convexity of the scaling weight

$$w_{\mathbf{e}_1+\mathbf{e}_2} \le w_{\mathbf{e}_1} + w_{\mathbf{e}_2} \qquad \forall \mathbf{e}_1, \mathbf{e}_2 \in \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}, \tag{5.2}$$

which could provide an interesting organising principle of the allowed scaling degrees.

5.2 Some implications

The above proposals have interesting implications for the landscape of 4d EFTs consistent with quantum gravity.

Existence of EFT strings

The first obvious consequence of our Conjecture 1 is the existence of axionic strings in any EFT with a moduli space containing infinite distance points. As emphasised in subsection 2.4, our conjectures are based on the behaviour of EFT string tensions evaluated at the EFT cut-off Λ . As a result, they are insensitive to whether the asymptotic limits under consideration correspond to actual flat directions of the theory or to directions subject to some potential with scales below Λ . The only thing that matters is whether there is some geodesic in field space that can be taken arbitrarily large while staying below a finite energy scale. Since our conjecture characterises infinite field distances in terms of EFT strings, it thus implies the universal presence of such strings in EFTs consistent with quantum gravity describing large distances in field space.

Furthermore, we are not predicting a single EFT string but rather a complete spectrum of EFT strings associated to each asymptotic regime. This can be formulated in terms of the saxionic cone in (4.5) and the cone of EFT string charges in (4.12) as follows:

EFT string completeness: for any EFT consistent with quantum gravity in a given asymptotic regime associated with a saxionic cone Δ , any string charge in $C_{\rm S}^{\rm EFT}$ is represented by an EFT string or by a superposition of them.

Notice that this statement is conceptually different from the one that the standard Completeness Hypothesis [53] would suggest. Indeed, in our context the latter would imply the existence of a set of string operators in a particular EFT asymptotic regime. However, the string EFT description not only involves the existence of an operator with the appropriate charges, but also that the corresponding string flow is realised, along which the said operator is well-defined. As we have discussed in section 4.1, this last ingredient is quite restrictive, in particular when considering $\frac{1}{2}$ BPS strings. What the EFT string completeness hypothesis implies is that for any charge in $C_{\rm S}^{\rm EFT}$ there is both a string flow and a well-defined operator along it. For other charges like e.g. those in $C_{\rm S} - C_{\rm S}^{\rm EFT}$ the existence of a flow would remain an open issue.

That these strings will be present for any 4d EFT depends on whether one can argue that every EFT contains weak coupling/infinite distance limits or not. In [1], it was indeed proposed that any moduli space should be non-compact, which seems a reasonable criterium for EFTs when evaluated at a high enough energy cut-off. This statement has no known counterexample for scale separated EFTs, i.e. that are truly four dimensional for an IR observer. Combining this with our conjecture, we would be predicting the universal presence of fundamental axionic strings in any 4d EFT consistent with quantum gravity!

Swampland Distance Conjecture

A key feature of EFT strings is that the string tension goes to zero monotonically along the RG flow, so the string is getting lighter and lighter as we approach the infinite distance limit in moduli space. When a probe string tension $T_{\mathbf{e}}$ gets below Λ^2 , the EFT necessarily breaks down, as discussed in section 3.4. As already pointed out in [32], this observation can be used to provide a bottom-up derivation the SDC, which predicts the presence of an infinite tower of states becoming exponentially light in the proper field distance at every infinite distance limit in moduli space (see Conjecture 2 of [1]). In particular, one can show that the exponential behaviour of the cut-off becomes just a consequence of having an EFT string satisfying the WGC.

To see this, consider the saxionic path (3.11) in field space. The dependence of the probe string tension $T_{\mathbf{e}}$ with σ can be obtained from the saxionic flow $s^i(\sigma) = s_0^i + \sigma e^i$ generated by the exact backreaction of an EFT string of charge \mathbf{e} , as described in section 3.1. More precisely we have that $T_{\mathbf{e}}(\sigma) = \mathcal{T}_{\mathbf{e}}(\sigma) = M_{\mathrm{P}}^2 e^i \ell_i(\sigma)$ satisfies (3.7). By this correspondence, the field space distance is given by the integral of the string charge along the flow — cf. (3.2). Using (3.7), this reads

$$d_{\sigma} = \frac{1}{M_{\rm P}} \int_0^{\sigma} \mathcal{Q}_{\mathbf{e}} d\sigma = \frac{1}{M_{\rm P}} \int_{\mathcal{T}_{\mathbf{e}}(\sigma)}^{\mathcal{T}_{\mathbf{e}}^0} \frac{1}{\mathcal{Q}_{\mathbf{e}}} d\mathcal{T}_{\mathbf{e}} \le \frac{1}{\gamma} \log \frac{\mathcal{T}_{\mathbf{e}}^0}{\mathcal{T}_{\mathbf{e}}(\sigma)}, \qquad (5.3)$$

where we have imposed the WGC bound $M_{\rm P} \mathcal{Q}_{\mathbf{e}} \geq \gamma \mathcal{T}_{\mathbf{e}}$ with constant γ in the last step. Therefore, the maximum EFT cut-off consistent with the existence of the string goes as

$$\Lambda_{\max}^2 = \mathcal{T}_{\mathbf{e}}(\sigma) < \mathcal{T}_{\mathbf{e}}^0 \exp\left(-\gamma \, \mathrm{d}_{\sigma}\right) \,. \tag{5.4}$$

This is one of the main implications of our proposal. We get that the WGC implies the exponential drop-off of the cut-off predicted by the SDC along paths generated by the EFT

string flows. Using Conjecture 1 we can then argue for the universality of the result, as any infinite field distance can then be associated to a string flow.

As follows from our experience in string theory examples and is captured by Conjecture 2, in most of the cases the string modes will not correspond to the leading tower of states becoming light, but there can be additional towers at a scale $m_* < \sqrt{T}$. In those cases the string tension still provides an upper bound on the cut-off and, therefore, the extremality factor of the string represents a lower bound on the exponential rate of the SDC. More precisely, according to (5.1), the exponential behaviour in (5.4) implies an exponential behaviour for m_* :

$$m_* \le m_*^0 \exp(-\alpha \,\mathrm{d}_\sigma), \quad \alpha = \frac{w\gamma}{2},$$
(5.5)

where recall that $w \in \mathbb{Z}_{>0}$ and γ is the extremality factor of the string.

Notice that the exponential behaviour of the string tension could have also been derived by using the asymptotic logarithmic form of the Kähler potential in (3.8), as the dependence of the string tension on the dual saxions is fixed by supersymmetry (3.9). However, the power of the above result is that we have alternatively derived the exponential behaviour of the tension without assuming any specific form of the Kähler potential, but rather by imposing the EFT string to satisfy the WGC. It is, therefore, a bottom-up model independent result, and no information about the asymptotic geometry of the internal space is required beyond the existence of an approximate axionic shift symmetry. What a Kähler potential of the form (3.8) does give us, is an argument for the EFT string flows to be asymptotically geodesic, see appendix C. This allows us to relate the exponential rate α with the one of the tower of states of the SDC. Moreover, as shown in [32], for a Kähler potential of the form (3.8) the charge-to-mass ratio of an extremal string can be written in terms of the exponents of the saxions in P(s). More precisely $\gamma = \sqrt{\frac{2}{n}}$, with n the so-called singularity type in [6, 8, 69] (see footnote 7), which can be derived from the asymptotic scaling of $P(s) \sim \sigma^n$ for $\sigma \to \infty$ along the string flow (4.9). Therefore we not only derive the SDC from the WGC for our EFT strings, but we also find a very constrained set of specific values for the SDC exponential rate.

The overall picture also fits well with refinements of the SDC, like the Emergent String Conjecture [15]. This conjecture distinguishes two classes of infinite distance limits: decompactification limits or emergent critical string limits. Applied to our 4d setup, one would identify asymptotic limits with a scaling weight w > 1 with decompactification limits, while those with scaling weight w = 1 should contain the limits in which the EFT string is an emergent critical string. In fact, our scheme only demands that the 4d quantum field theory description must be broken above $\sqrt{\mathcal{T}}$. So even in the case w = 1 the breakdown could either be triggered by an infinite number of EFT string oscillations or a tower of a different nature scaling at the same rate. Finally, notice that a determination of w purely from the EFT description would allow us to estimate the EFT cut-off just using EFT data! We leave this exciting possibility for future research and continue with some observations regarding the value of w observed in string compactifications in the following.

EFT membranes, the scaling weight and the species scale

Four-dimensional EFTs with potentials below some cut-off scale Λ are quite generic in string compactifications with fluxes, where the potential is multi-branched, and each branch is not invariant under the discrete shift symmetry associated to the EFT string. In that case, the string is not a gauge invariant operator by itself, and a membrane needs to be attached to it to restore gauge invariance. By consistency, it is then expected that the string-membrane system remains as a localised operator along the string flow, just like the string is in the absence of a potential. In practice, this amounts to require that the membrane is a fundamental EFT object as long as the EFT string satisfies (2.1), i.e.

$$\Lambda^3 < \mathcal{T}_{\rm mem} < M_{\rm P}^2 \Lambda \,. \tag{5.6}$$

Such membranes were dubbed EFT membranes in [32, 42], and their relation with EFT strings was analysed in [32, section 4]. There it was found that the tension of the lightest EFT membrane behaves asymptotically along the EFT string flow as

$$\frac{\mathcal{T}_{\rm mem}}{M_{\rm P}^3} \sim \left(\frac{\mathcal{T}_{\rm str}}{M_{\rm P}^2}\right)^{n/2} \,, \tag{5.7}$$

up to some constant factor, while the scaling of heavier membranes can differ by additional positive integral powers of $\mathcal{T}_{\rm str}/M_{\rm P}^2$. Here *n* is the singularity type of the string flow introduced above.

This asymptotic behaviour has an interesting interpretation in terms of Conjecture 2. To see this, let us define the following EFT mass scale associated with the membrane:

$$E_{\rm mem} \equiv \frac{\mathcal{T}_{\rm mem}}{M_{\rm P}^2}\,,\tag{5.8}$$

which, as in [32, 42], sets the energy scale of gravitational effects induced by the membrane as well as of the corresponding fluxes. Notice that the relation (5.7) can be written in the form (5.1), if we simply replace (E_{mem}, n) by (m_*, w) . We then see that Conjecture 2 can be extended also to E_{mem} only if the total homogeneity degree of P(s) is an *inte*ger. This is indeed what we have observed in all the string/M-theory models analysed in section 6. In this sense, the extension of (5.1) to E_{mem} may provide a physical interpretation of this observation and further support for the universal form (3.8) of the asymptotic Kähler potential.

Now, because by definition along an EFT string flow we have that $\Lambda \leq m_* \leq \mathcal{T}_{\text{str}}^{1/2}$, the condition (5.6) requires that $E_{\text{mem}} < m_*$. Conjecture 2 and (5.7) therefore imply that

$$\frac{E_{\rm mem}^2}{m_*^2} \sim \left(\frac{\mathcal{T}_{\rm str}}{M_{\rm P}^2}\right)^{n-w} < 1 \qquad \Longrightarrow \qquad w \le n \,. \tag{5.9}$$

Additionally, requiring that $m_*^3 \leq \mathcal{T}_{\text{mem}}$ along the string flow leads to $n \leq 3w$. We therefore infer that the asymptotic behaviour of the Kähler potential restricts the scaling weight of the flow via the following inequality

$$w \le n \le 3w. \tag{5.10}$$

As we will see in the next section, all the string examples that we analyse satisfy this inequality. In fact, in the examples analysed in section 6 we find very specific values for the scaling weight, namely

$$1 \le w \le 3, \tag{5.11}$$

even if (5.10) could in principle allow for larger values. As already mentioned, the lower bound identifies m_* with the mass scale of the EFT string modes, together with a tower of (typically KK) states that becomes massless equally fast. The upper bound is however a bit surprising, so let us try to rephrase it in more physical terms.

Let us assume that w > 1 and that m_* corresponds to the scale of a tower of states, as for instance the Emergent String Conjecture would predict. In terms of this tower one can estimate the species scale [70–74] along the asymptotic limit as

$$\Lambda_{\rm sp} \sim \frac{M_{\rm P}}{\sqrt{N}}, \quad \text{with} \quad N \sim \frac{\Lambda_{\rm sp}}{m_*} \sim \frac{\Lambda_{\rm sp}}{\mathcal{T}^{\frac{w}{2}} M_{\rm P}^{1-w}}.$$
 (5.12)

We obtain that

$$\Lambda_{\rm sp}^2 \sim M_{\rm P}^2 \left(\frac{\mathcal{T}}{M_{\rm P}^2}\right)^{\frac{w}{3}} , \qquad (5.13)$$

so the range (5.11) is equivalent to

$$m_*^2 \le \mathcal{T} \le \Lambda_{\rm sp}^2 \,. \tag{5.14}$$

In other words, if $w \leq 3$ we would be able to see the string as a quantum object below or around the species scale, so the semiclassical description of the string will break down at most at Λ_{sp} . From our experience with perturbative string theory, it is very suggestive that a quantum extended object appears before we reach the quantum gravity scale Λ_{sp} , since it is indeed a string-like spectrum the one that cures the UV divergences associated to quantum field theory coupled to gravity. In addition, (5.14) suggests that the magnetic dual axion is no longer fundamental above Λ_{sp} . Since we are associating infinite distance limits with axionic shift symmetries, this would imply that the non-compactness of the moduli space is only manifest below Λ_{sp} . This nicely fits with the Emergence proposal [4, 6, 56] for which infinite field distance limits emerge in the IR from integrating out towers of light states up to the species scale.

Moduli space curvature at infinity

In the original work [1], several conjectures were proposed in addition to the SDC. In particular, Conjecture 3 of [1] proposes that the scalar curvature near points at infinity is non-positive. This seems a natural consequence of the typical hyperbolic metric arising at large field distances, e.g. from a Kähler potential of the form $K = -\log s$. However, this conjecture is false in general, and a counterexample was first pointed out in [75]. In a 4d $\mathcal{N} = 1$ context, we will see that the failure of the conjecture is related to the presence of degenerate string flows. Nevertheless, even if this conjecture is false, we will draw a weaker condition along the same lines, namely that points at infinity have at least one negative holomorphic sectional curvature. This statement follows from our Conjecture 1 and it is realised in all the string theory examples checked in this work.

To begin with, let us come back to our heterotic example in section 4.2, which already presents a counterexample of the conjecture 3 in [1]. We can compute the scalar curvature derived from the Kähler potential in (4.28), obtaining

$$R = \frac{6(81s_1^6 + 162s_1^5s_2 + 135s_1^4s_2^2 - 45s_1^2s_2^4 - 18s_1s_2^5 - 2s_2^6)}{s_2^3(3s_1 + s_2)^3} .$$
(5.15)

One can check that this curvature is negative when $s_2 \to \infty$ but positive when $s_1 \to \infty$. As discussed in section 4.2, there is a fundamental difference between both limits, as the former corresponds to a non-degenerate string flow while the latter to a degenerate one. In other words, there is more than one EFT string becoming tensionless in the latter one. This suggests that it is precisely the existence of these degenerate flows what yields the positive curvature at certain regions near the infinite distance points. In fact, one can show that the curvature will always be negative along the non-degenerate flows and approach the value for the corresponding hyperbolic plane. This can be understood from a more geometric perspective as follows. A perturbative regime in which all elementary string flows are non-degenerate corresponds to a factorisable Kähler potential to leading order. Hence, the field space factorises into different hyperbolic planes and the scalar curvature is therefore negative everywhere. Contrary, degenerate flows occur when the Kähler potential cannot be factorised in the sense that each elementary string flow selects a different leading term in the homogeneous function $P = e^{-K}$. In the language of [8], the leading term of the Kähler potential is different in each growth sector, as explained in footnote 7.¹¹

The above argument only shows that non-degenerate flows imply negative scalar curvature, but it would be interesting to show also the reverse, as examples suggest. This would require a more detailed study of the interplay between positive scalar curvature and degenerate flows, which we leave for future work. Instead, in the following, we will use 4d EFT string flows to prove a weaker condition than the conjecture in [1], namely that all holomorphic sectional curvatures are negative.

Indeed, let us consider an EFT string flow solution of the form (2.12). On the one hand, the solution (2.12) maps a disc centered at the origin of \mathbb{C} and with radius $r_0 = |z_0|$ to a holomorphic disc D in a perturbative region of the EFT field space \mathcal{M} . The area of this disc computed with the pulled-back metric from \mathcal{M} corresponds to the linear energy density (2.27), which is nothing but the string tension at the cut-off scale $\Lambda_0 \sim 1/r_0$. The existence of the EFT string implies that such a tension is finite for some range of values for r_0 , and so should be the corresponding disc area. On the other hand, the radial direction of the disc in \mathbb{C} , e.g. $|z| \in [0, r_0]$, is mapped via (2.13) to an infinite distance trajectory on \mathcal{M} , as discussed in section 3. It follows that the holomorphic disc $D \subset \mathcal{M}$ has negative

¹¹This happens when the singularity type does not increase in the enhancement chain. For example, in (4.28) the singular divisor located at $s^1 = \infty$, i.e. $D_{s^1=\infty}$ has singularity type n = 2 while the intersection $D_{s^1,s^2=\infty}$ has again n = 2. Hence, $D_{s^2=\infty}$ is sort of behaving as a finite distance divisor with n = 0. It makes sense then that the local geometry is not hyperbolic and even has positive curvature, which is something characteristic of finite distance divisors.

curvature. In particular, for a physical string charge with asymptotic behaviour $Q_{\mathbf{e}} \sim \sigma^{-\eta}$ along the trajectory (3.1), one obtains a Gaussian curvature for the disc D of the form

$$K_D \sim -\eta \sigma^{2\eta - 2} \,, \tag{5.16}$$

up to a constant positive numerical factor. Hence, for the range $\eta \in (0, 1]$ leading to an infinite distance we obtain $K_D < 0$. Remarkably, for the choice $\eta = 1$ that corresponds to an asymptotically saturated WGC bound $M_P Q_e = \gamma T_e$ as realised by EFT strings, the curvature is asymptotically constant, while otherwise it decreases along the flow. If \mathcal{M} is a Kähler manifold, this implies that near the point at infinity corresponding to the EFT string RG flow UV endpoint there is at least one holomorphic sectional curvature which is negative. Finally, using Conjecture 1 one may conclude that all points of \mathcal{M} at infinite distance have this property.

Since the scalar curvature of a Kähler manifold is a sum of holomorphic bisectional curvatures, the statement derived from Conjecture 1 is much weaker than Conjecture 3 of [1]. Nevertheless, the former seems to be compatible with all the explicit constructions investigated up to date.

Global symmetries and generalisation to higher dimensions

The identification of infinite distance limits with RG flows of EFT strings also makes manifest the relation between the SDC and the absence of global symmetries, which was first proposed in [6]. The EFT breakdown is required to avoid a 0-form continous axionic shift symmetry and U(1) 1-form global symmetry from the dual \mathcal{B}_2 -field that would be restored otherwise in the limit, which would go against the swampland criterion of no global symmetries. This also fits with the proposal in [17] for which there should be a gauge coupling of a *p*-form gauge field vanishing at every infinite field distance limit. This allows one to merge the conjectures by having the same tower of states satisfying both the WGC the SDC and, therefore, the SDC exponential rate gets bounded by the WGC extremality factor [17, 76]. In our case, this gauge field is the \mathcal{B}_2 -field, and the string modes satisfy both the WGC and the SDC, so indeed the exponential rate is given by the extremality factor in (5.4).

It is also interesting to notice that, instead of using the Completeness Hypothesis, we can also argue for the existence of the magnetically charged string from requiring the absence of generalised global symmetries¹² along the path in \mathcal{M} , and not only at the infinite distance point. The axion is dual to a 2-form gauge field, and the string is required to break the associated 2-form global symmetry. From a topological perspective, this symmetry is associated to a non-trivial cobordism class¹³ in the moduli space that gets trivialised once we add the string defect, in a similar spirit than in [78, 79]. In the case at hand, this triviality of the cobordism group reduces to the more familiar notion of triviality of the

¹²Indeed, the absence of generalised global symmetries is equivalent to completeness of spectrum for compact connected gauge groups, see [77].

¹³More concretely, it corresponds to $\Omega_1(\mathcal{M})$ where \mathcal{M} is the moduli space. The string flow provides the bordism that allows one to deform smoothly the circle in the moduli space associated to the axion to a point. We thank Miguel Montero and Jacob McNamara for this observation.

first homotopy group, which was in fact conjectured in [1]. Geometrically, it implies that the 1-cycle in the moduli space associated to the axion can be shrunk at infinite distance, as realised by the string flow.

It is also interesting to discuss possible generalisations of our proposal to higher dimensions. First, it is easy to convince onself that the interpretation of the string backreaction as an RG flow outlined in [32] and section 3 is not exclusive of four-dimensional theories. Indeed, in a generic $d \ge 4$ -dimensional spacetime, a fundamental object of codimension 2, namely satisfying

$$\Lambda^{d-2} < \mathcal{T} < M_{\rm P}^{d-2} \,, \tag{5.17}$$

will exhibit a backreaction similar to strings in 4d. A well-known case corresponds to D7branes in the 10d EFT that is type IIB supergravity, see for instance [80]. Therefore, a discussion analogous to the one carried in [32] and section 3 should apply for any codimension 2 BPS object in a supersymmetric EFT, independently of the dimension. On the other hand, the connection between infinite distance limits and codimension-two BPS flows is not expected to be generically valid beyond our 4d setting. For instance, in ten-dimensional IIA supergravity there are no codimension-two BPS objects whose backreaction flow could describe the infinite distance limit $e^{\phi} \rightarrow 0$. Hence, in general, one should at least drop one the two requirements: the BPSness or the codimension-2.

Keeping the codimension-2 of the object but dropping the BPS condition would still allow us to have an RG flow interpretation of the infinite distance limits, as only the codimension matters to obtain the logarithmic behaviour originated from the brane couplings being marginally relevant. However, we would loose computational control. Given the relation with global symmetries outlined above, it is in fact natural to expect the existence of codimension-2 objects whenever the singular loci are codimension-1 singularities of the moduli space, as these objects would be required to break the corresponding (d-2)-form global symmetry. The story gets more complicated when the moduli space is not complex, as happens in 10d IIA or in M-theory on a Calabi-Yau three-fold. In those cases, it might very well be that there are still some extended objects required to break an emergent p-form global symmetry at the infinite distance regimes, but that they have a larger codimension. If that is the case, the connection with RG flows might not be present in general. Hence, without further investigation, we cannot conclude anything about the fate of our conjectures in higher dimensions.

6 String theory evidence

In this section, we check Conjectures 1 and 2 for several classes of 4d $\mathcal{N} = 1$ string theory compactifications. In particular, we provide evidence for the universal presence of an EFT string at every infinite field distance limit as well as the correlation (5.1) between the string tension and the tower scale m_* . As usual in string compactifications 4d string tensions are computed by dimensional reduction of different brane actions, and as such they describe the probe, scale-independent tension that we denote by T in this paper. Nevertheless, in terms of saxion values T has the same expression as the effective tension \mathcal{T} . For simplicity and in order to facilitate the comparison of our results with expression like (2.23) and (5.1), we will denote both string tensions as \mathcal{T} in this section.

6.1 Heterotic on Calabi-Yau three-folds

We start by considering again the heterotic string Calabi-Yau compactifications introduced in section 4.2. In that section we have already discussed the saxionic sector describing the dilaton and Kähler moduli, as well as the corresponding string and instanton cones. We now discuss in some generality the behaviour of the relevant mass scales along the EFT string flows (4.9), with charges taking values in (4.22c), and consider also the complex structure sector, which contributes to the Kähler potential (4.23) with the additional term

$$K_{\rm cs} = -\log\left(\mathrm{i}\int_X \Omega \wedge \bar{\Omega}\right) \,. \tag{6.1}$$

For later convenience, let us write the ansatz of the 10d string frame metric

$$ds^2 = e^{2A} ds_4^2 + l_s^2 ds_X^2 , \qquad (6.2)$$

where

$$e^{2A} = \frac{l_s^2 M_{\rm P}^2 e^{2\phi}}{4\pi V_X} = \frac{l_s^2 M_{\rm P}^2}{4\pi s^0} , \qquad (6.3)$$

is fixed by requiring that ds_4^2 is the 4d Einstein frame metric.

F1 flows

We first consider the flow generated by a string with $e_{F1} = 1$, along which only s^0 changes as

$$s^{0}(\sigma) = s_{0}^{0} + \sigma \,, \tag{6.4}$$

and then diverges as $\sigma \to \infty$ (the non-elementary case $e_{F1} > 1$ is completely analogous). Recalling that s^0 is defined as in (4.20), and since all other chiral fields are kept fixed, one can see that this flow corresponds to sending $e^{\phi} \to 0$, while the string frame volume V_X measured in string units remains constant, thus it is the well-known weak (string) coupling limit. The corresponding EFT string tension, measured by the 4d Einstein frame metric, is just given by

$$\mathcal{T}_{\rm F1} = \frac{2\pi e^{2A}}{l_s^2} = \frac{M_{\rm P}^2}{2s^0} = M_{\rm P}^2 \ell_0 \,, \tag{6.5}$$

which reproduces (3.9) upon using (6.3) and (4.29) as expected. The string indeed becomes tensionless in the asymptotic limit $\sigma \to \infty$.

The KK scale measured by 4d Einstein frame metric vanishes as well. To see this, notice that the 4d KK scale can be identified with $m_* = e^A/R_*$, where R_* is the largest compactification length-scale measured in the string frame and e^A is the Weyl-rescaling given in (6.3). From here we obtain that

$$m_*^2 = \frac{e^{2A}}{R_*^2} = \frac{\pi M_{\rm P}^2}{\hat{R}_*^2 s^0} = \frac{2\pi}{\hat{R}_*^2} \mathcal{T}_{\rm F1} \,. \tag{6.6}$$

	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{6}}$
$\mathcal{T}_{\mathrm{F1}}^{1/2}$	F1	
m_*	$m_{\rm KK}$	
$E_{\rm mem}$	NS5	
$\mathcal{T}_{ m mem}^{1/3}$		NS5

Table 2. Mass scalings along the flow generated by F1 strings.

with $\hat{R}_* = 2\pi R_*/l_s$. Since \hat{R}_* does not change along the flow, we conclude that indeed the EFT string tension \mathcal{T}_{F1} scales to zero with the same rate as the KK scale. This confirms Conjecture 2 in (5.1) with scaling weight w = 1. Even though (6.6) was derived in the large volume regime, we expect that it gives a reasonable estimate also for generic string size compactifications. In this case m_* is not necessarily the KK-scale but can be identified with the string scale measured by the 4d Einstein frame metric. Hence $m_*^2 \sim \frac{e^{2A}}{l_*^2} \sim \mathcal{T}_{F1}$.

Furthermore, according to the general discussion around (5.7) the lightest membrane EFT mass scale $E_{\text{mem}} \equiv \mathcal{T}_{\text{mem}}/M_{\text{P}}^2$ will also scale as m_* , since the relevant contribution to the Kähler potential (4.23) is $-\log s^0$, which corresponds to the singularity type n = 1. We summarise the scaling of the relevant quantities in table 2.

NS5 flows

As discussed in section 4.2, the set of EFT strings associated with large Kähler moduli limits correspond to NS5-branes wrapping effective nef divisors, see (4.22c). Suppose that $D \simeq e^a D_a$ is such a divisor. Then the corresponding string flow is

$$\boldsymbol{s}(\sigma) = \boldsymbol{s}_0 + \mathbf{e}\,\sigma\,.\tag{6.7}$$

Here $s = s^a[D_a] \in \mathcal{K}(X)$ and $\mathbf{e} = [D] = e^a[D_a] \in \operatorname{Nef}(X)_{\mathbb{Z}}$ and the corresponding EFT string tension, using (4.19) and (4.24), is given by

$$\mathcal{T}_{\mathbf{e}} = \frac{\pi e^{2A}}{l_s^2} \int_D e^{-2\phi} J \wedge J = M_{\mathrm{P}}^2 e^a \ell_a = \frac{3M_{\mathrm{P}}^2 \kappa(\mathbf{e}, \boldsymbol{s}, \boldsymbol{s})}{2 \kappa(\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s})}, \qquad (6.8)$$

where e.g. $\kappa(\mathbf{e}, \mathbf{s}, \mathbf{s}) \equiv \kappa_{abc} e^a s^b s^c \equiv \mathbf{e} \cdot \mathbf{s} \cdot \mathbf{s}$. We clearly have $\mathcal{T}_{\mathbf{e}} \sim 3M_{\rm P}^2/2\sigma$ for $\sigma \to \infty$, as expected. Notice that, in order to keep the saxion s^0 fixed along (6.7), we need to take the strong coupling limit $e^{\phi} \to \infty$ as $\sigma \to \infty$. More precisely, since

$$e^{2\phi} = \frac{\kappa(\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s})}{6s^0}, \qquad (6.9)$$

 $e^{2\phi}$ must diverge as $\kappa(s, s, s)$. This means that there may appear new light degrees of freedom due to strong coupling effects, which become light faster than the KK scale associated to the Calabi-Yau threefold. Let us see how this works for the $E_8 \times E_8$ heterotic string, while we will consider the SO(32) model in section 6.2. Its strong coupling limit is provided by the Hořava-Witten construction of M-theory compactified on $X \times S^1/\mathbb{Z}_2$ [59, 60]. The estimate of the new KK scale along the interval S^1/\mathbb{Z}_2 is the same as for IIA/M-theory, and then coincides with the mass of a D0-brane. Hence, taking into account that the Einstein frame Weyl rescaling (6.3) remains constant along the flow, this gives a mass scale

$$m_*^2 = \frac{2\pi \, e^{2A-2\phi}}{l_s^2} = \frac{3M_P^2}{\kappa(s,s,s)} \,. \tag{6.10}$$

Notice that, since s^0 and the complex structure moduli are constant along the flow, m_* scales as the membrane EFT mass scale (5.8):

$$E_{\rm mem}^2 \simeq M_{\rm P}^2 e^K = \frac{M_{\rm P}^2 e^{K_{\rm cs}}}{s^0 \,\kappa(\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s})} = \frac{e^{K_{\rm cs}}}{s^0} \,m_*^2 \,. \tag{6.11}$$

By comparing (6.8) and (6.10), we see that the relation between m_*^2 and $\mathcal{T}_{\mathbf{e}}$ will depend on the scaling of $\kappa(\mathbf{s}, \mathbf{s}, \mathbf{s})$ and $\kappa(\mathbf{e}, \mathbf{s}, \mathbf{s})$ in the asymptotic limit. Keeping in mind that the triple intersection of three nef divisors is always non-negative, we can distinguish three cases, each one realising (5.1) with a different scaling weight w:

- 1. $\kappa(\mathbf{e}, \mathbf{e}, \mathbf{e}) > 0$. In this case (5.1) is realised with scaling weight w = 3.
- 2. $\kappa(\mathbf{e}, \mathbf{e}, \mathbf{e}) = 0$, but $\kappa(\mathbf{e}, \mathbf{e}, \mathbf{e}') > 0$ for some $\mathbf{e}' \in \operatorname{Nef}(X)_{\mathbb{Z}}$. In this case \mathbf{e} belongs to the boundary of $\overline{\mathcal{K}}(X)$ and (5.1) is realised with scaling weight w = 2.
- 3. $\kappa(\mathbf{e}, \mathbf{e}, \mathbf{e}') = 0$ for any $\mathbf{e}' \in \operatorname{Nef}(X)_{\mathbb{Z}}$. Again \mathbf{e} belongs to the boundary of $\overline{\mathcal{K}}(X)$ but now (5.1) is realised with scaling weight w = 1.

Before discussing in detail the above classification, we anticipate that this matches the three main types of infinite distance singular limits that can be realised in a Calabi-Yau threefold [8, 9] and are associated with the different possible scalings of $\kappa(s, s, s) \sim \sigma^n$ and $\kappa(\mathbf{e}, \mathbf{s}, \mathbf{s}) \sim \sigma^{n-1}$ for $\sigma \to \infty$. Indeed, from (6.8) and (6.10) it follows that the integer n precisely coincides with the scaling weight w appearing in (5.1). The integer n was denoted the singularity type, and the three cases are known as Type IV, III and II, corresponding to n = 3, 2, 1 respectively. This integer corresponds to the effective nilpotency order of the log-monodromy transformation in the mirror complex structure moduli space, and it was matched with the above properties of the intersection numbers in [9]. Hence, the scaling weight of the flow is directly linked to the singularity type of the asymptotic limit. As expected, n = 3 corresponds to a total decompactification limit, while n = 1 is the limiting case in which the string modes become light at the same rate of a KK scale.

So far we have only considered the KK scale (6.10) associated with the Hořava-Witten interval. In order to understand the possible role of the Calabi-Yau KK mass scales, we need to better describe the geometries associated with the above classification. To this end, we can borrow part of the results of [15], which provides a complementary viewpoint on the above classification.

Our three cases display then the following features:

Case 1. In this case $\kappa(\mathbf{s}, \mathbf{s}, \mathbf{s}) \simeq \kappa(\mathbf{e}, \mathbf{e}, \mathbf{e})\sigma^3$ and $\kappa(\mathbf{e}, \mathbf{s}, \mathbf{s}) \simeq \kappa(\mathbf{e}, \mathbf{e}, \mathbf{e})\sigma^2$, and then the string-frame volume diverges as σ^3 . Hence $m_*^2 \simeq M_P^2 \sigma^{-3}$ and Conjecture 2 is realised with

scaling weight w = 3. Note also that, since the Weyl rescaling (6.3) is constant, the Calabi-Yau KK mass scale $m_{\rm KK} = e^A/(l_{\rm s}V_X^{1/6})$ asymptotically scales like $m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{s^0}\sigma^{-1}$, and then is much heavier than m_* . Furthermore, from (6.8) it is clear that $\mathcal{T}_{\mathbf{e}'} \simeq M_{\rm P}^2\sigma^{-1}$ for any string charge $\mathbf{e}' \in \operatorname{Nef}(X)_{\mathbb{Z}}$. This means that, in the terminology introduced around (4.15), these flows are maximally degenerate in the Kähler sector. A concrete example of an EFT string generating such a flow is provided by the *H*-string in the two-moduli example of section 4.2.

Case 2. In this case, the self-intersection $C = \mathbf{e} \cdot \mathbf{e}$ corresponds to a non-trivial Mori curve, since its intersection with any other nef divisor is non-negative. This case is in direct correspondence with the *J*-class A limit of [15]. As proved in [81, 82] and emphasised in [15], a Calabi-Yau three-fold realising this possibility corresponds to a T^2 fibration over a two-fold, where *C* is a multiple of the T^2 fiber. An example is provided by the *L*-divisor in the two moduli model of section 4.2. Along (6.7) we asymptotically have $\kappa(\mathbf{s}, \mathbf{s}, \mathbf{s}) \simeq 3\kappa(\mathbf{e}, \mathbf{e}, \mathbf{s}_0)\sigma^2$ and $\kappa(\mathbf{e}, \mathbf{s}, \mathbf{s}) \simeq 2\kappa(\mathbf{e}, \mathbf{e}, \mathbf{s}_0)\sigma$, where $\kappa(\mathbf{e}, \mathbf{e}, \mathbf{s}_0) = \int_C \mathbf{s}_0 = \int_C \mathbf{s} > 0$ measures the constant volume of *C*. Hence, from (6.10) we get $m_*^2 \simeq M_P^2 \sigma^{-2}$ and Conjecture 2 is realised with scaling weight w = 2. Since the volume of *C* is constant, the scaling $\sim \sigma^2$ of the internal volume corresponds to an expansion of the four-dimensional base of the T^2 fibration. Hence, as in Case 1 we have $m_{KK}^2 \simeq \frac{M_P^2}{s^0} \sigma^{-1}$, which is asymptotically much heavier than m_*^2 . Furthermore from (6.8) it follows that the order of degeneracy of the flow (defined below (4.15)) is given by the number of linearly independent charges \mathbf{e}' (including \mathbf{e}) for which $\kappa(\mathbf{e}', \mathbf{e}, \mathbf{e}) = 0$.

Case 3. This is related to the *J*-class B limit of [15]. In this case $\mathbf{e} \cdot \mathbf{e} = 0$ and then $\kappa(\mathbf{s}, \mathbf{s}, \mathbf{s}) \simeq 3\kappa(\mathbf{e}, \mathbf{s}_0, \mathbf{s}_0)\sigma$, while $\kappa(\mathbf{e}, \mathbf{s}, \mathbf{s}) \equiv \kappa(\mathbf{e}, \mathbf{s}_0, \mathbf{s}_0) > 0$ gives the constant volume of the nef divisor \mathbf{e} . Then $m_*^2 \simeq M_P^2 \sigma^{-1}$ and Conjecture 2 is realised with scaling weight w = 1. As discussed in [15], this case corresponds to a K3 or T^4 fibration over a \mathbb{P}^1 , where a multiple of the fibre can be identified with the divisor class \mathbf{e} . The scaling σ of the internal volume corresponds to the expansion of the base \mathbb{P}^1 . Hence, we again get $m_{\mathrm{KK}}^2 \simeq \frac{M_P^2}{s^0} \sigma^{-1}$ which, however, now scales like m_*^2 and could be taken as an alternative definition thereof. In [15] it was proven that if \mathbf{e} and \mathbf{e}' represent two non-proportional nef divisors, then $C' = \mathbf{e} \cdot \mathbf{e}'$ represents a non-trivial curve. Applied to the present setting, this implies that for any EFT charge \mathbf{e}' non-proportional to \mathbf{e} , we have that $\kappa(\mathbf{e}', \mathbf{s}, \mathbf{s}) \simeq 2\kappa(\mathbf{e}', \mathbf{e}, \mathbf{s}_0)\sigma$, where $\kappa(\mathbf{e}', \mathbf{e}, \mathbf{s}_0) \equiv \int_{C'} \mathbf{s} > 0$ is a constant measuring the volume of C'. Then from (6.8) we see that the corresponding tension $\mathcal{T}_{\mathbf{e}'}$ is not asymptotically vanishing and the flow is non-degenerate.

Note also that one can also check the index convexity (5.2). Since in this sector $1 \leq w_{\mathbf{e}} \leq 3$, the only case that needs to be checked corresponds to both \mathbf{e}_1 and \mathbf{e}_2 of scaling weight 1 (that is, case 3). That is $\mathbf{e}_1 \cdot \mathbf{e}_1 = \mathbf{e}_2 \cdot \mathbf{e}_2 = 0$. It follows that $\mathbf{e} = \mathbf{e}_1 + \mathbf{e}_2$ has certainly vanishing triple self-intersection, $\kappa(\mathbf{e}, \mathbf{e}, \mathbf{e}) = 0$, and then $w_{\mathbf{e}} \leq 2$.

The above results are summarised in table 3.

Notice that this table does not include the behaviour of any string tension beyond the one of the EFT string generating the flow. In principle there could be non-EFT

	$\sigma^{-\frac{3}{2}}$	σ^{-1}	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{3}}$	$\sigma^{-\frac{1}{6}}$
$\mathcal{T}_{ m str}^{1/2}$			NS5		
m_*	M-th. KK – Case 1	M-th. KK – Case 2	M-th. KK – Case 3		
CY $m_{\rm KK}$			CY $m_{\rm KK}$		
$E_{\rm mem}$	NS5 - Case 1	NS5 - Case 2	NS5 - Case 3		
$\mathcal{T}_{ m mem}^{1/3}$			NS5 - Case 1	NS5 - Case 2	NS5 - Case 3

Table 3. Mass scalings along the flows generated by NS5-strings in heterotic models.

strings whose tension decreases faster than σ^{-1} . This happens for instance in the example of section 4.2, where the non-EFT string tension $\mathcal{T}_{\mathbf{e}_B}$ decreases like σ^{-3} along the flow generated by the EFT string charge \mathbf{e}_H . In this case we have that $\mathcal{T}_{\mathbf{e}_B}^{1/2} \sim m_*$, and so the presence of the light non-EFT string does not change the result for m_* obtained from our previous analysis. In the following we will assume that this is always the case, either because the non-EFT string tensions are asymptotically at or above m_*^2 or because they do not generate an infinite tower of oscillation modes. It would be interesting to check this expectation in other explicit examples.

Complex structure flows

When some complex structure (CS) modulus is taken to be large, we may approach an infinite distance locus where an approximate axionic shift symmetry arises. We can then choose a local parametrisation splitting the asymptotic complex structure moduli $\tau^{\alpha} \simeq \tau^{\alpha} + 1$ into the axions and saxionic partners, as in the general description proposed in the present paper. In particular, at the 4d EFT level, this perturbative regime admits a description in terms of dual \mathcal{B}_2 potentials, charged strings and corresponding BPS flows, as in section 2.

The classification of the possible infinite distance limits in the complex structure moduli space of a Calabi-Yau 3-fold can be borrowed from [6, 8], leading to the three cases discussed above for the Kähler sector. Indeed, by mirror symmetry, any asymptotic region of the CS moduli space is expected to be equivalent to a corresponding asymptotic region of the Kähler structure moduli space of the mirror Calabi-Yau. Hence, the two mirror descriptions should share the same saxionic cone structure and spectrum of strings (henceforth dubbed CS strings), whose flows describe the possible infinite-distance limits. In particular, it is clear that the CS EFT strings are mirror to the NS5 EFT strings discussed above. By exploiting the SYZ description of mirror symmetry [83], one can also more concretely identify the 4d CS strings with 10d KK-monopoles wrapping internal divisors.

The correspondence between CS strings and NS5 strings cannot be naively extended to the estimate of the relevant mass scales along the corresponding flows. In particular, the saxion s^0 defined in (4.20) does *not* contain the flowing CS moduli and then not only the Kähler moduli but also the dilaton remains constant along the flow. Correspondingly, the strong coupling mass scale (6.10) does not become asymptotically light. Absent some scale that plays the role of the M-theory scale in NS5-flows, this suggests that along CS flows the tower scale m_* is mirror to the Calabi-Yau KK-mass scales appearing in table 3. This identification would imply that Conjecture 2 is always realised with scaling weight w = 1. This conclusion can be easily checked in simple toroidal orbifold models, see appendix F.1 for an explicit example. It would be interesting to perform a direct estimate of m_* for general Calabi-Yau geometries, to confirm whether our naive expectation for w is realised or not.

6.2 Type I on Calabi-Yau three-folds

Type I models are related to heterotic models by S-duality. Hence, their Calabi-Yau compactifications are very similar to the heterotic ones, described in subsections 4.2 and 6.1. In particular, the complex structure moduli have Kähler potential (6.1), and the remaining moduli sector is still described by a Kähler potential of the form (4.23). However, the saxions s^a and s^0 are defined differently from the heterotic ones (4.19) and (4.20). Rather, they are the image of (4.19) and (4.20) under S-duality

$$s = s^{a}[D_{a}] \equiv e^{-\phi}J$$
 , $s^{0} \equiv e^{-\phi}V_{X} = \frac{1}{3!}e^{2\phi}\kappa(s, s, s)$, (6.12)

where J denotes the string frame Kähler form of the compactification space X. Hence, the EFT saxionic, string and instanton sectors are the same as in the heterotic case while their microscopic interpretations are different and related by S-duality. In particular, the 10d dilaton and the internal string frame volume are given by the following saxionic functions

$$e^{2\phi} = \frac{6s^0}{\kappa(s,s,s)}, \quad V_X = \sqrt{\frac{6(s^0)^3}{\kappa(s,s,s)}}.$$
 (6.13)

Note also that the Weyl-rescaling e^{2A} appearing in the string-frame ansatz (6.2) now takes the form

$$e^{2A} = \frac{l_{\rm s}^2 M_{\rm P}^2 \sqrt{6}}{4\pi \sqrt{s^0 \kappa(\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s})}} \,. \tag{6.14}$$

Consider for instance the flow generated by a D1-string, which is identical to the F1 flow (6.4) of the heterotic model. In the type I case, from (6.12) it is clear that the asymptotic limit $s^0 \to \infty$, with s^a fixed, implies that $e^{\phi} \to \infty$. Hence, the flow cannot be described microscopically within the perturbative type I model, but rather we need to use the S-dual description of subsection 6.1.

Consider now D5-brane strings, which are S-dual to the NS5-strings considered in subsection 6.1. We saw that the heterotic NS5-string flows make the 10d dilaton diverge and in subsection 6.1 we used the M-theory uplift of the $E_8 \times E_8$ model to study them. The type I models now allow us to extend that discussion to the SO(32) model as well. Indeed, the D5-brane strings flows are as in (6.7), while s^0 remains constant. Hence from (6.13) it follows that $e^{2\phi}$ vanishes as $1/\kappa(s, s, s)$ along the flow and the 10d type I model remains weakly coupled. On the other hand, from (6.13) it also follows that $V_X \to 0$ as $1/\sqrt{\kappa(s, s, s)}$, and the classical definition of the saxions becomes questionable.

We can be more specific. The string-frame Kähler form can be written as a function of the saxions as follows

$$J = \sqrt{\frac{6s^0}{\kappa(\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s})}} \,\boldsymbol{s} \,. \tag{6.15}$$

We can now adopt the same classification in terms of intersection numbers as in subsection 6.1. Consider first Case 2, the internal space can be regarded as a T^2 -fibration. In the type I setting, the classical volume of the T^2 -fibre vanishes asymptotically as σ^{-1} , while the base volume remains constant. Hence, we certainly cannot trust the classical description and we should further go to some other dual picture. For instance, we can imagine to perform a fibre-wise double T-duality and go to a D3-string of the kind described in the following subsection 6.3. More directly, we could try to identify the relevant m_*^2 with the winding scale associated with the vanishing T^2 -fiber:

$$m_*^2 = \frac{e^{2A} \operatorname{Vol}(T^2)}{\ell_s^2} \sim \frac{M_P^2}{\sqrt{s^0 \sigma^2}} \qquad \text{for } \sigma \to \infty.$$
(6.16)

This is consistent with Conjecture 2, with scaling weight w = 2 as in the $E_8 \times E_8$ case, although the two microscopic interpretations of m_*^2 are completely different.

In Case 1 the entire classical string frame volume goes to zero as $\sigma^{-\frac{3}{2}}$. We can make an estimate of m_*^2 by restricting to a toroidal model which scales homogeneously and using the winding mass formula (6.16). Since $\kappa(s, s, s) \sim \sigma^3$, we now get $m_*^2 \sim M_{\rm P}^2 \sigma^{-2}$ as in Case 2. This supports again Conjecture 2 with scaling weight w = 2. Note that this differs from the scaling weight w = 3 found in the $E_8 \times E_8$ model.

Case 3 is slightly different. The internal space can be regarded as a T^4 /K3-fibration over \mathbb{P}^1 , in which the string frame \mathbb{P}^1 volume diverges as $\sqrt{\sigma}$, while the fibre's volume tends to zero as $1/\sigma$. We can then try to estimate m_*^2 by identifying it with the base KK-mass scale

$$m_*^2 = \frac{e^{2A}}{l_s^2 \operatorname{Vol}(\mathbb{P}^1)} \sim \frac{M_{\rm P}^2}{s^0 \sigma} \qquad \text{for } \sigma \to \infty \,. \tag{6.17}$$

This confirms Conjecture 2 with scaling weight w = 1. Alternatively, we may use the winding mass estimate (6.16) associated with a shrinking T^4 -fiber, getting the same scaling weight.

Finally, one may perform a naive estimate of the relevant scales under the CS-string flows, by applying a reasoning similar to the one in heterotic models. As in there, absent some other scales that could play the role of m_* , one would obtain that Conjecture 2 is realised with scaling weight w = 1. See appendix F.1 for a check of this conclusion in an explicit model.

6.3 Type IIB/F-theory compactifications

We now consider IIB/F-theory compactifications to four-dimensions and its weak-coupling limit, see for instance [80] for a review. In these models the compactification space X is Kähler and the axio-dilaton $\tau = C_0 + ie^{-\phi}$ is non-constant and undergoes non-trivial monodromies around 7-branes. It is also useful to think of such compactification in terms of a dual M-theory compactification over a Calabi-Yau four-fold Y that is elliptically fibred over the base space X. From the IIB viewpoint, it is convenient to adopt the 10d Einsteinframe description. We will then use an ansatz of the form (6.2) where, however, ds^2 denotes the 10d Einstein frame metric, which will be used to measure 10d length scales in this subsection. Note that the Weyl-rescaling is now

$$e^{2A} = \frac{l_{\rm s}^2 M_{\rm P}^2}{4\pi V_X} \,. \tag{6.18}$$

The four-dimensional $\mathcal{N} = 1$ EFT has been discussed in detail in [80, 84] in the constant warping approximation. Warping effects [85] may also be incorporated as in [86–90], but we will not consider them in the following.

Kähler saxions and EFT strings

We will first focus on the saxionic sector that parametrises the Kähler sector. As usual, the Kähler moduli v_a appear in the expansion of the (Einstein frame) Kähler form

$$J = v_a[D^a], (6.19)$$

where $[D^a]$ is the Poincaré dual of a basis of divisors $D^a \in H_4(X, \mathbb{Z})$. Assume first for simplicity that $H^3(Y) = 0$. In this case, the Kähler sector is parametrised by the EFT saxions

$$\boldsymbol{s} \equiv \frac{1}{2}J \wedge J \in H^4(X, \mathbb{R}) \quad \Rightarrow \quad s^a \equiv \int_{D^a} \boldsymbol{s} = \frac{1}{2}\int_{D^a} J \wedge J = \frac{1}{2}\kappa^{abc} v_b v_c \,. \tag{6.20}$$

In order to better determine the corresponding saxionic cone Δ , we have to specify the relevant instanton charges. In the large volume regime these are naturally associated with Euclidean D3-branes wrapping effective divisors. This leads to the identification

$$\mathcal{C}_{\mathrm{I}} = \mathrm{Eff}^{1}(X)_{\mathbb{Z}}, \qquad (6.21)$$

see appendix A.2 for the notation. Since $h^{2,0}(X) = 0$ for a three-fold X which is the base of Calabi-Yau four-fold, the divisors generate $H_4(X, \mathbb{Z})$ and then we can identify the lattice $M_{\mathbb{Z}}$ with the torsion-free homology group $H_4(X, \mathbb{Z})_{\text{t.f.}}$.

The Kähler potential for the chiral fields $t^a = a^a + is^a$ associated with the saxions s^a is obtained by inverting (6.20) to express

$$K_{\rm ks} = -2\log \int_X J \wedge J \wedge J = -2\log \kappa(\boldsymbol{v}, \boldsymbol{v}, \boldsymbol{v}), \qquad (6.22)$$

where $\kappa(\boldsymbol{v}, \boldsymbol{v}, \boldsymbol{v}) \equiv \kappa^{abc} v_a v_b v_c$, as a function of $s^a = \text{Im} t^a$. From (2.17) one can then compute the dual saxions

$$\ell_a = \frac{3v_a}{\kappa(\boldsymbol{v}, \boldsymbol{v}, \boldsymbol{v})} = \frac{\int_{C_a} J}{2V_X}, \qquad (6.23)$$

where $C_a \in H_2(X, \mathbb{Z})$ is a basis of two-cycles dual to D^a (i.e. $C_a \cdot D^b = \delta_b^a$) and $V_X = \frac{1}{3!}\kappa(\boldsymbol{v}, \boldsymbol{v}, \boldsymbol{v})$ is the (Einstein frame) volume of X.

From our general definition (4.5), we can therefore identify Δ with the interior of the dual of the cone of effective divisors $\text{Eff}^1(X)$. This coincides with the interior of the cone in $H^{2,2}(X,\mathbb{R})$ generated by the Poincaré dual of *movable* curves $\text{Mov}_1(X)$ [91] — see also appendix A.2 and [92] for a review:

$$\Delta \simeq \operatorname{Int}[\operatorname{Mov}_1(X)] \qquad (\text{via Poincaré duality}). \tag{6.24}$$

By applying (4.12), we then see that the EFT string charges can be identified with the set of movable curves

$$\mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}} = \mathrm{Mov}_1(X)_{\mathbb{Z}} \,. \tag{6.25}$$

Physically, the corresponding EFT strings correspond to D3-branes wrapping movable curves inside X. Roughly speaking, a curve is movable if it is part of a family of curves that spans the entire X. More general BPS strings correspond instead to D3-branes wrapping effective curves: $C_{\rm S} = {\rm Eff}_1(X)_{\mathbb{Z}}$.¹⁴ Here the relevant lattice is $N_{\mathbb{Z}} = H_2(X,\mathbb{Z})_{\rm t.f.}$, and the pairing with the dual lattice $M_{\mathbb{Z}}$ generated by the instanton charges coincides with the intersection number between four and two cycles. One can also check that a probe D3-brane wrapping an effective curve $C = e^a C_a$ gives an effective string of tension $\mathcal{T}_{\mathbf{e}} = M_{\rm P}^2 \langle \mathbf{e}, \boldsymbol{\ell} \rangle$, in agreement the expected EFT value.

The correspondence between elements of the Kähler cone $\mathcal{K}(X)$ and elements of the saxionic cone (6.24) is defined through the map $J \mapsto s \equiv \frac{1}{2}J \wedge J$ (combined with Poincaré duality). Note that this map is injective but *not* generically surjective [93]. Hence, in general, the saxionic cone Δ parametrises a space which is larger than $\mathcal{K}(X)$. In particular, there will be EFT string charges in (6.25) that generate flows that take us away from the Kähler cone. When this happens, we cross a boundary of $\mathcal{K}(X)$ and at this point a curve $C \subset X$ collapses. However, by construction no divisor collapses at any point of the EFT string flow, and from the 4d perspective we remain in the perturbative region (4.4) where all non-perturbative effects that break the axionic shift symmetries $\int_D C_4^{\text{RR}}$ are suppressed. Therefore, on physical grounds we expect that the actual saxionic space of the theory is an extension of $\mathcal{K}(X)$ that accommodates all the asymptotic limits generated by (6.25).

This expectation is supported by the so-called 'small modifications' of the space X, which have been observed in different instances. These are essentially transitions to another space X' through the contraction of a curve $C \subset X$ and the blow-up of a curve $C' \subset X'$, with no divisor collapsing along them. In the upstairs elliptically fibered Calabi-Yau fourfold Y they correspond to flops of the geometry. From the 4d viewpoint they correspond to strings whose tension vanishes classically but may receive quantum corrections — see e.g. [94, 95] for related discussions.

It is therefore natural to glue together the Kähler cones of the spaces related by these transitions, to get an extended Kähler cone $\mathcal{K}(X)_{\text{ext}}$. Based on our 4d physical intuition of EFT strings, we would then expect that $\mathcal{K}(X)_{\text{ext}}$ can be alternatively parametrised by the above saxionic cone, that is, that the map

$$J \in \mathcal{K}(X)_{\text{ext}} \quad \mapsto \quad \mathbf{s} \equiv \frac{1}{2} J \wedge J \in \Delta,$$
 (6.26)

is a bijection. We will not try to prove this expectation in full generality, but rather to collect some evidence supporting it.

Part of the evidence comes from the fact that one can construct a huge class of Ftheory models (6.26) where the base three-fold X belongs to the class of *Mori Dream*

¹⁴Since $Mov_1(X)$ is a subset of $\overline{Eff}_1(X)$ but not of $Eff_1(X)$, a priori a movable curve is *pseudo*-effective but not necessarily effective. The EFT string completeness proposed in section 5.2 suggests that in F-theory models any movable curve admits an effective representative.

spaces [96].¹⁵ This class includes toric and Fano spaces [97] as particular subclasses and, basically by definition, enjoy the following decomposition of the cone of movable divisors into Kähler chambers connected by small modifications:

$$\operatorname{Mov}^{1}(X) = \bigcup_{X' \text{ small mod. of } X} \operatorname{Nef}(X') \equiv \overline{\mathcal{K}(X)}_{\operatorname{ext}}.$$
 (6.27)

So, if $D \in Mov^1(X)$, through a sequence of small transitions one can reach a Kähler chamber in which D is nef. By combining this property together with some results of [92, 98], one can conclude that (6.26) is indeed a bijection.

Further support is provided by the F-theory weak coupling limit, in which the IIB compactification can be described in terms of a double cover Calabi-Yau three-fold \hat{X} . It is known that the different Kähler chambers subdividing $\mathcal{K}_{ext}(\hat{X})$ are related by flops and that $\overline{\mathcal{K}_{ext}}(\hat{X})$ coincides with the cone of movable divisors [99], hence $\overline{\mathcal{K}_{ext}}(\hat{X}) = \text{Mov}^1(\hat{X})$. By restricting this identity to classes which are invariant under the orientifold involution one gets $\overline{\mathcal{K}_{ext}}(\hat{X})_+ = \text{Mov}^1(\hat{X})_+$, where $\overline{\mathcal{K}_{ext}}(\hat{X})_+$ can indeed be identified with the extended Kähler cone of the IIB compactification and $\text{Mov}^1(\hat{X})_+$ with the cone of movable divisors over the corresponding F-theory base space X. In fact, our working assumption $H^3(Y) = 0$ implies that $H^2(\hat{X})_- = 0$ and than we expect the restriction to orientifold-even classes to trivialise to an identity map.

Examples

We now consider two simple illustrative examples and discuss, without trying to be exhaustive, the behaviour of the relevant mass scales under the flows generated by EFT string charges (6.25). A more involved example of this sort is considered in appendix F.2.

Example 1: $X = \mathbb{P}^3$. Consider first the simplest possible compactification space: $X = \mathbb{P}^3$. In this case $M_{\mathbb{Z}} \simeq H_4(X, \mathbb{Z}) \simeq \mathbb{Z}$ is generated by the hyperplane divisor $L \simeq \mathbb{P}^2$, which also generates the BPS instanton charges

$$\mathcal{C}_{\mathrm{I}} = \mathrm{Eff}^{1}(X)_{\mathbb{Z}} = \mathrm{Mov}^{1}(X)_{\mathbb{Z}} = \mathrm{Nef}^{1}(X)_{\mathbb{Z}} = \mathbb{Z}_{\geq 0}L.$$
(6.28)

Dually, $N_{\mathbb{Z}} \simeq H_2(X, \mathbb{Z}) \simeq \mathbb{Z}$ is generated by the two-sphere $C \simeq \mathbb{P}^1 = L \cdot L$. This generates also the possible string charges

$$\mathcal{C}_{\mathrm{S}} = \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}} = \mathrm{Eff}_{1}(X)_{\mathbb{Z}} = \mathrm{Mov}_{1}(X)_{\mathbb{Z}} = \mathbb{Z}_{\geq 0}C.$$
(6.29)

The Kähler form can be written as J = v[L] and since $L^3 = 1$ the saxion is $s = \frac{1}{2}v^2$. The Kähler potential (6.22) then reduces to $K = -3 \log s$, up to irrelevant constants. The possible EFT string flows (4.9) reduce to $s(\sigma) = s_0 + e\sigma$, with $e \in \mathbb{Z}_{\geq 0} = C_{\mathrm{S}}^{\mathrm{EFT}}$. Under such a flow, asymptotically, the volume scales homogeneously like $V_X \sim \sigma^{\frac{3}{2}}$. This implies that the KK-scale $m_*^2 = \frac{e^{2A}}{l_s^2} V_X^{-\frac{1}{3}} = \frac{M_{\mathrm{P}}^2}{4\pi} V_X^{-\frac{4}{3}}$ scales as $M_{\mathrm{P}}^2 \sigma^{-2}$. This is in agreement with Conjecture 2 with scaling weight w = 2.

¹⁵We thank Antonella Grassi for discussions on Mori Dream spaces and related aspects.

The scaling of the lightest membranes can be easily inferred from scaling of the Kähler potential. Indeed, as follows from the general relation (5.7), the scale (5.8) set by the lightest membranes falls off as $E_{\text{mem}} \sim M_{\text{P}} \sigma^{-\frac{3}{2}}$, so $n = 3 \neq w$. Notice also that the membrane tension scales as $\mathcal{T}_{\text{mem}} \sim M_{\text{P}}^3 \sigma^{-\frac{3}{2}}$ along the flow, and is then heavier than m_*^3 .

Example 2: $X = \{\mathbb{P}^1 \text{ fibration over } \mathbb{P}^2\}$. A more interesting example is provided by choosing as internal space X the toric *n*-twisted \mathbb{P}^1 fibration over \mathbb{P}^2 defined by the gauged linear sigma model

	u_1	u_2	u_3	u_4	u_5	FI
$U(1)_{1}$	1	1	1	-n	0	$v_1 > 0$
$U(1)_{2}$	0	0	0	1	1	$v_2 > 0$

where n = 1, 2, ... A detailed discussion about this base can be found in [80]. If $\mathcal{D}_I = \{u_I = 0\}$ denote the toric divisors, let us pick the following (non-independent) divisors

$$D^1 = \mathcal{D}_1 \simeq \mathcal{D}_2 \simeq \mathcal{D}_3$$
, $D^2 = \mathcal{D}_5 \simeq E + nD^1$, $E = \mathcal{D}_4 \simeq D^2 - nD^1$. (6.30)

The relevant lattices are $M_{\mathbb{Z}} = H_4(X, \mathbb{Z}) \simeq \mathbb{Z}^2$ and $N_{\mathbb{Z}} = H_2(X, \mathbb{Z}) \simeq \mathbb{Z}^2$. The cone of nef divisors $\operatorname{Nef}^1(X)_{\mathbb{Z}}$ is generated by D^1 and D^2 , while the cone of effective divisors $\operatorname{Eff}^1(X)_{\mathbb{Z}} = \mathcal{C}_{\mathrm{I}} \subset M_{\mathbb{Z}}$ is generated by D^1 and E. The intersection numbers are summarised by

$$\mathcal{I} = (D^1)^2 D^2 + n D^1 (D^2)^2 + n^2 (D^2)^3, \qquad (6.31)$$

where the coefficient of each monomial gives the corresponding intersection number. We can expand the Kähler form in the nef basis, to describe the Kähler cone

$$\mathcal{K}(X) = \left\{ J = v_1[D^1] + v_2[D^2] \in H^2(X, \mathbb{R}) | v_1, v_2 > 0 \right\},$$
(6.32)

where v_1, v_2 are the two Kähler moduli that measure the volumes of the dual effective curves

$$C_1 \simeq D^1 \cdot E \simeq D^1 \cdot (D^2 - nD^1) \quad , \quad C_2 \simeq D^1 \cdot D^1 \,,$$
 (6.33)

which satisfy $C_a \cdot D^b = \delta_a^b$. Note also that $E \cdot D^2 = 0$. We can identify C_2 with the \mathbb{P}^1 fibre and C_1 with the push-forward of \mathbb{P}^1 in the base through the section defined by $u_4 = 0$.

The saxions are given by the following expansion

$$s = \frac{1}{2}J \wedge J = s^{1}[C_{1}] + s^{2}[C_{2}] \in H^{4}(X, \mathbb{R})$$

with $s^{1} = v_{1}v_{2} + \frac{1}{2}nv_{2}^{2}, \quad s^{2} = \frac{1}{2}(v_{1} + nv_{2})^{2}.$ (6.34)

The actual saxionic cone is obtained by imposing (4.5) with C_{I} generated by the effective divisors D^{1} and E:

$$\Delta = \left\{ \boldsymbol{s} \in H^4(X, \mathbb{R}) \left| \langle D^1, \boldsymbol{s} \rangle = s^1 > 0, \ \langle E, \boldsymbol{s} \rangle = s^2 - ns^1 > 0 \right\}.$$
(6.35)

From our definition (4.12) it is easy to see that the cone of EFT string charges:

$$\mathcal{C}_{\rm S}^{\rm EFT} = \{ \mathbf{e} = e^1 \, C_1 + e^2 \, C_2 \in \operatorname{Mov}_1(X)_{\mathbb{Z}} | \, e^1 \ge 0 \,, \, e^2 - ne^1 \ge 0 \}$$
(6.36)

is generated by the charges $\mathbf{e}_1 \equiv C_1 + nC_2 = D^1 \cdot D^2$ and $\mathbf{e}_2 \equiv C_2$.

Note that in this case we have $Mov^1(X) = Nef^1(X)$ and then (6.26) with $\mathcal{K}(X)_{ext} \equiv \mathcal{K}(X)$ should be one-to-one. This can be explicitly checked by inverting (6.34) into

$$v_1 = \sqrt{2(s^2 - ns^1)}, \quad v_2 = \frac{1}{n} \left(\sqrt{2s^2} - \sqrt{2(s^2 - ns^1)}\right),$$
 (6.37)

which indeed cover the entire Kähler cone as we vary $s \in \Delta$. Furthermore

$$V_X = \frac{\sqrt{2}}{3n} \left[(s^2)^{\frac{3}{2}} - (s^2 - ns^1)^{\frac{3}{2}} \right]$$
(6.38)

and then, up to an irrelevant constant, the explicit form of the Kähler potential (6.22) is

$$K_{\rm ks} = -2\log\left[(s^2)^{\frac{3}{2}} - (s^2 - ns^1)^{\frac{3}{2}}\right], \qquad (6.39)$$

and the dual saxions are

$$\ell_1 = \frac{3n\sqrt{s^2 - ns^1}}{2\left[(s^2)^{\frac{3}{2}} - (s^2 - ns^1)^{\frac{3}{2}}\right]}, \qquad \ell_2 = \frac{3\left(\sqrt{s^2 - \sqrt{s^2 - ns^1}}\right)}{2\left[(s^2)^{\frac{3}{2}} - (s^2 - ns^1)^{\frac{3}{2}}\right]}.$$
 (6.40)

Note that ℓ_1 and ℓ_2 can (classically) take any possible value and then we can identify the dual saxionic cone \mathcal{P} with $\mathcal{K}(X)$: $\mathcal{P} = \{ \boldsymbol{\ell} = \ell_1[D]^1 + \ell_2[D]^2 | \ell_1, \ell_2 > 0 \}.$

We are ready to discuss the mass scalings under the EFT string flows (4.9). We first choose a charge $\mathbf{e} = e^1C_1 + e^2C_2$ with $e^1 > 0$ (and $e^2 \ge ne^1$). In other words, by decomposing \mathbf{e} in terms of the generators \mathbf{e}_1 and \mathbf{e}_2 , we are assuming that the contribution of \mathbf{e}_1 is non-vanishing (in particular, the most elementary choice is $\mathbf{e} = \mathbf{e}_1$.) We can then easily see that the internal volume scales as $V_X \sim \sigma^{\frac{3}{2}}$ for $\sigma \to \infty$. Furthermore, $v_2 \sim \sqrt{\sigma}$ and, if $e^2 > ne^1$, also $v_1 \sim \sqrt{\sigma}$. In any case, the estimate of the KK mass scaling goes as in the \mathbb{P}^3 example discussed above. In particular, Conjecture 2 is verified with scaling weight w = 2. Note also that all these flows, including the elementary one with $\mathbf{e} = \mathbf{e}_1$, are degenerate of order 2 — see the definition below (4.15) — since both ℓ_1 and ℓ_2 vanish in the asymptotic limit. Moreover, along this flow, the membrane scale (5.8) can be easily obtained from (5.7), with n = 3.

It remains to discuss the possibility $e^1 = 0$, for which it is sufficient to restrict to the elementary charge $\mathbf{e} = \mathbf{e}_2 = C_2$. The corresponding flow (4.9) becomes $s^1 = s_0^1$ and $s^2 = s_0^2 + \sigma$. Asymptotically for $\sigma \to \infty$, we have $V_X \sim s_0^1 \sqrt{\sigma}$, $v_1 \sim \sqrt{\sigma}$ and $v_2 \sim \frac{s_0^1}{\sqrt{\sigma}}$. This implies that the KK-scale $m_*^2 = \frac{e^{2A}}{\ell_s^2 v_1}$ associated with the growing $C_1 \simeq \mathbb{P}^1$ scales like $1/\sigma$ and then realises (2) with scaling weight w = 1. Note also that $\ell_2 \sim 1/\sigma$, while ℓ_1 remains asymptotically finite. Hence the flow is non-degenerate. Furthermore, the membrane scale (5.8) falls off as $E_{\text{mem}} \sim M_{\text{P}} \sigma^{-\frac{3}{2}}$ along such a flow, realising (5.7) with n = 3. One may be skeptic about the reliability of these conclusions, since the volume of the fibral curve C_2 shrinks to zero size. However, this EFT flow is a particular realisation of a *J*-class A limit of [20, 100], in which it is argued that there exists a dual weakly coupled heterotic description in which the D3-brane wrapping C_2 is dualised to a F1-string. The D3-brane wrapping C_2 precisely corresponds to our EFT string of charge $\mathbf{e} = \mathbf{e}_2 = C_2$ and then its flow can be mapped to an F1 flow of section 6.1.

In fact, the above results should qualitatively hold for more general J-class A limits with heterotic dual, in which X is a more general \mathbb{P}^1 -fibration. The D3-brane wrapping the \mathbb{P}^1 -fibre is an example of elementary EFT string that, borrowing the results of [20, 100], satisfies 2 with scaling weight w = 1 and is non-degenerate.

Weak-coupling limit and odd moduli

Of course, one can also consider asymptotic limits along other moduli space directions. Here we consider the weak-coupling limit, in which an F-theory compactification is well described in terms of a compactification on a double cover Calabi-Yau three-fold \hat{X} in presence of O7-planes.

In the weak coupling limit $\text{Im} \tau \to \infty$, where $\tau = C_0 + ie^{-\phi}$ is the IIB axio-dilaton. In this limit the associated exponentially suppressed non-perturbative corrections are generated by D(-1)-instantons and the strings carrying a corresponding axionic charges are given by D7-branes wrapping the entire internal space.

It is interesting to observe that, as emphasised for instance in [101], in these backgrounds there exists a "half" D(-1)-brane, whose path-integral contribution $e^{i\pi\tau}$ to the 4d EFT is invariant only under an even integral shifts of τ : $\tau \simeq \tau + 2$ (while $\tau \neq \tau + 1$).¹⁶ Correspondingly, any consistent EFT string must generate an *even* monodromy of τ . In microscopic terms, these means that these EFT correspond to stacks of an *even* number e_{D7} of D7-branes wrapping the internal space. This conclusion is consistent with the fact that such D7-branes should carry an Sp(e_{D7}) gauge group, since they have four mutually transversal directions with respect to the O7-planes.

Along the flow generated by such EFT strings, the dilaton changes as $\text{Im} \tau = \text{Im} \tau_0 + \sigma e_{\text{D7}}$. If e_{D7} were the only string charges supported by the D7-brane, the saxions s^a defined in (6.26) would not run. Since these correspond to Einstein frame volumes, all string frame volumes $\text{vol}_{\text{st}}(D) = (\text{Im} \tau)^{-1} \int_D \mathbf{s}$ would actually vanish as $\sigma \to 0$, leading to a breakdown of the 10d supergravity description. One should then go to some 10d dual frame, for instance by picking a toroidal configuration and performing six T-dualities. In this way one would get a IIB compactification with O5-planes, in which the EFT string is represented by a D1 brane, obtaining a setting which is qualitatively analogous to the type I case discussed above, hence confirming Conjecture 2 with scaling weight w = 1.

However, the D7-string can generically support some induced D3-brane charge due to internal curvature corrections and world-volume fluxes. Such charges would activate a flow of the s^a saxions as well, so that the string frame volume of some divisors would remain finite. The internal world-volume fluxes could also carry additional string charges associated with a non-trivial odd cohomology group $H^2(\hat{X})_-$ which has been so far assumed to be trivial (since we assumed $H^3(Y) = 0$ in the dual M-theory Calabi-Yau geometry). In the case of non-trivial $H^2(\hat{X})_-$ there appear additional 'odd' moduli $\beta^{\alpha} = c^{\alpha} - \tau b^{\alpha}$

¹⁶We stress that this 2Z periodicity of τ is a 4d effect, which should be interpreted as a spontaneous breaking of the 10d Z periodicity, due to the presence of the O7-plane.

obtained by expanding $C_2 - \tau B_2$ in a basis $[\tilde{D}_{\alpha}] \in H^2(X, \mathbb{Z})_-$, Poincaré dual to a set of odd divisors \tilde{D}_{α} . The fluxed D7-strings can then induce around them monodromies of the axions $\operatorname{Re} \beta^{\alpha}$.

Note that from the integral periodicity of the IIB NS-NS B_2 potential it follows that $b^{\alpha} \simeq b^{\alpha} - n^{\alpha}$ with $n^{\alpha} \in \mathbb{Z}$. This induces the identification

$$\beta^{\alpha} \simeq \beta^{\alpha} + n^{\alpha} \tau \,. \tag{6.41}$$

This *perturbative* duality is an example of the duality group $\mathcal{G}_{\mathbb{Z}}$ appearing in the saxionic cone conjecture of section 4.4. It also acts on the other moduli t^a , since if $H^2(X,\mathbb{Z})_-$ is non-trivial their definition is modified by b^{α} -dependent terms [102]. More precisely, we have

$$t^{a} \simeq t^{a} - \tilde{\kappa}^{a}_{\alpha\beta} n^{\alpha} \beta^{\beta} - \frac{1}{2} \tilde{\kappa}^{a}_{\alpha\beta} n^{\alpha} n^{\beta} \tau$$
(6.42)

where $\tilde{\kappa}^a_{\alpha\beta}$ is the intersection number between the even divisor D^a and the odd divisors $\tilde{D}_{\alpha}, \tilde{D}_{\beta}$. This duality implies that one may isolate a fundamental region for this saxionic sector, for instance $0 \leq \text{Im}\beta^{\alpha} \leq \text{Im}\tau$ corresponding to $-1 \leq b^{\alpha} \leq 0$. This conical fundamental region is indeed rational polyhedral, in agreement with the saxionic cone conjecture of section 4.4.

A detailed study of this extended moduli sector and of the corresponding EFT strings originating from fluxed D7-branes would require a careful definition of the flux quantisation conditions and of the corresponding string and instanton lattices. We leave this task, together with a more accurate test of the validity of our general framework for these classes of models, to the future.

Finally, one may address CS string flows. Following the same strategy as in the heterotic and type I models, one may perform a naive estimate of the relevant compactification scales by looking at the NS5 string flows in a mirror manifold. If there is no relevant scale below the mirror KK scales, this would again imply that Conjecture 2 is realised with scaling weight w = 1. If however there is some lighter tower of states, this conclusion could be modified. For instance, one could consider the would-be mirror duals of type IIA D0-branes in the orientifold limit, which would be non-BPS D3-branes wrapping special Lagrangian three-cycles. If these states generate a tower of states, they would likely behave like M-theory scale in table 4, implying the same scaling weights w = 1, 2, 3 as in there. Again, it would be interesting to perform a more detailed analysis of these CS flows to elucidate which of these two options, or even if a third one, is realised.

6.4 Type IIA Calabi-Yau orientifolds

Let us now consider Type IIA compactified on the projection of Calabi-Yau threefold X under the O6 orientifold involution $\iota: X \to X$. We now summarise the relevant ingredients to describe its EFT, focusing on the closed string moduli sector.

The Calabi-Yau moduli are encoded in the (string frame) Kähler form J and the (3,0)-form Ω , whose normalisation is fixed by

$$\frac{\mathrm{i}}{8}\Omega \wedge \bar{\Omega} = \frac{1}{3!}J \wedge J \wedge J \,, \tag{6.43}$$

and an overall constant phase has been fixed by the requirement that the O6-planes are calibrated by Re Ω . J and Ω must satisfy the projection condition $\iota^* J = -J$ and $\iota^* \Omega = \overline{\Omega}$, respectively.

There are two sets of chiral fields parametrising closed string moduli. As in the heterotic case, a first set of chiral fields is provided by the $b_2(X)_-$ complexified (string frame) Kähler moduli $t^a \equiv a^a + is^a$ parametrising $B_2 + iJ$

$$B_2 + iJ \equiv t^a [D_a^+] \equiv (a^a + is^a) [D_a^+], \qquad (6.44)$$

where now $[D_a^+]$ provide a basis for the odd cohomology classes $H^2(X; \mathbb{R})_-$.¹⁷ The remaining moduli, which include the dilaton, complex structure moduli and axionic partners, are encoded in a second set of $b_3(X)_+$ chiral fields $\hat{t}^{\alpha} \equiv \hat{a}^{\alpha} + i\hat{s}^{\alpha}$, identified through the expansion

$$C_3 + i e^{-\phi} \operatorname{Re}\Omega \equiv \hat{t}^{\alpha} \left[\Sigma_{\alpha}^{-} \right] \equiv \left(\hat{a}^{\alpha} + i \hat{s}^{\alpha} \right) \left[\Sigma_{\alpha}^{-} \right], \qquad (6.45)$$

where $[\Sigma_{\alpha}^{-}]$ forms a basis for the even three-form classes $H^{3}(X;\mathbb{R})_{+}$. Notice that the saxions \hat{s}^{α} parametrise the dilaton and the complex-structure moduli of X only through a non-trivial field redefinition, which furthermore depends also on the Kähler structure moduli because of the normalisation (6.43).

From the above description it follows that we can split the saxionic cone into

$$\Delta = \Delta_{\rm K} \times \hat{\Delta} \,, \tag{6.46}$$

where $\Delta_{\rm K}$ is generated by the Kähler saxions s^a , while the remaining saxions \hat{s}^{α} take values in $\hat{\Delta}$. We will also use the index-free notation

$$\mathbf{s} \equiv [J] = s^{a}[D_{a}^{+}] \in H^{2}(X, \mathbb{R})_{-},$$

$$\hat{\mathbf{s}} \equiv e^{-\phi}[\operatorname{Re}\Omega] = \hat{s}^{\alpha}[\Sigma_{\alpha}^{-}] \in H^{3}(X, \mathbb{R})_{+}.$$
(6.47)

The string frame volume of X depends only on the Kähler saxions s^a :

$$V_X(s) = \frac{1}{3!} \int_X J \wedge J \wedge J = \frac{1}{3!} \kappa(\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s}), \qquad (6.48)$$

with $\kappa(\mathbf{s}, \mathbf{s}, \mathbf{s}) \equiv \kappa_{abc} s^a s^b s^c$ as in the heterotic models. On the other hand, by following [103] one can express the odd class $[e^{-\phi} \text{Im} \Omega] \in H^3(X, \mathbb{R})_-$ as a function of the \hat{s}^{α} saxions only, and define an associated Hitchin function

$$\mathcal{H}(\hat{s}) \equiv \frac{\mathrm{i}}{8} \int_{X} e^{-2\phi} \Omega \wedge \bar{\Omega} \qquad \rightarrow \qquad \delta \mathcal{H} = \frac{1}{2} \int_{X} \delta \hat{s} \wedge [e^{-\phi} \mathrm{Im} \Omega] \,. \tag{6.49}$$

Notice that $\mathcal{H}(\hat{s})$ is homogeneous of degree two and that the 10d dilaton can be written as

$$e^{2\phi} = \frac{V_X(s)}{\mathcal{H}(\hat{s})} \,. \tag{6.50}$$

¹⁷[S] refers to the Poincaré dual of a *p*-cycle S, which in our conventions is defined by $\int_{S} \omega = \int_{X} \omega \wedge [S]$, for any closed *p*-form ω . Since the O6-plane involution inverts the orientation, Poincaré duality connects even/odd cycles with odd/even cohomology classes.

In fact, this description of the moduli is accurate in a perturbative regime in which the backreaction of fluxes and localised sources can be neglected and the warping can be approximated to be constant. In such a case, the 10d string frame metric is as in (6.2)with

$$e^{2A} = \frac{l_{\rm s}^2 M_{\rm P}^2}{2\pi \mathcal{H}(\hat{s})},$$
 (6.51)

and the relevant terms contributing to the EFT Kähler potential are [102]

$$K = -\log V_X(s) - 2\log \mathcal{H}(\hat{s}). \qquad (6.52)$$

We also note also that in this perturbative regime the entire moduli space of the closed string sector corresponds to axionic symmetries, in the sense that all chiral fields can be split into axion + saxion. Hence, all of them can be dualised to linear multiples, as in (2.17). In particular, the dual saxions are

$$\ell_{a} = -\frac{1}{2} \frac{\partial K}{\partial s^{a}} = \frac{1}{4V_{X}} \int_{D_{a}^{+}} J \wedge J \equiv \frac{3 \kappa ([D_{a}^{+}], \boldsymbol{s}, \boldsymbol{s})}{2 \kappa (\boldsymbol{s}, \boldsymbol{s}, \boldsymbol{s})},$$

$$\hat{\ell}_{\alpha} = -\frac{1}{2} \frac{\partial K}{\partial \hat{s}^{\alpha}} = -\frac{1}{2\mathcal{H}} \int_{\Sigma_{\alpha}^{-}} e^{-\phi} \operatorname{Im} \Omega.$$
(6.53)

We stress that, by using the results of [103], $\hat{\ell}_{\alpha}$ are functions of \hat{s}^{α} only.

The choice of basis $\{[D_a^+]\}_a, \{[\Sigma_\alpha^+]\}_\alpha$ and their interpretation in terms Poincaré dual classes of cycles can be fixed by demanding that the axions a^a , \hat{a}^{α} have periodicity one. Similarly to the case of the axio-dilaton in type IIB compactifications with O7-planes, the periodicity of type IIA axions is fixed by the set of instanton charges in the compactification, which depends on the precise action of the orientifold involution on the relevant homology classes. Consider a basis $D_i \in H_4(X;\mathbb{Z})$ of $b_2(X)$ divisors and associated basis $C^j \in$ $H_2(X,\mathbb{Z})$ of two-cycles, such that $D_i \cdot C^j = \delta_i^j$. By taking linear combinations of the cycles C^{j} we can then construct a set of odd two-cycles C^{a}_{-} that generate the orientifoldodd lattice $H_2(X,\mathbb{Z})_-$. We expect that there exists a set of dual even four-cycles D_a^+ , which satisfy $D_a^+ \cdot C_-^b = 2\delta_a^b$ and can be identified with the Poincaré dual of the classes $[D_a^+]$ in (6.44). Note that in general D_a^+ may only generate a sublattice of $H_4(X;\mathbb{Z})_+$. For instance, suppose that $\{C^a_{-}\}_a$ can be chosen to be a subset of $\{C^j\}_j$. In this case the dual four-cycles would be of the form $D_a^+ \equiv D_a + \iota(D_a)$. The worldsheet instantons wrapping C_{-}^{a} would correspond to crosscaps in the quotient geometry, as in [104], and so Dirac quantisation would imply that a NS5-branes can only wrap four-cycles of the form $D^+ = D + \iota(D).$

Similarly, one can always consider a symplectic basis of three-cycles $\Sigma_{\mu}, \tilde{\Sigma}^{\nu} \in H_3(X; \mathbb{Z})$, such that $\Sigma_{\mu} \cdot \tilde{\Sigma}^{\nu} = \delta^{\nu}_{\mu}$. Again, we can identify a set of even three-cycles $\tilde{\Sigma}^{\alpha}_{+}$ generating $H_3(X;\mathbb{Z})_+$ and we expect that there exists a dual set of odd-three cycles Σ^{-}_{α} , such that $\Sigma^{-}_{\alpha} \cdot \tilde{\Sigma}^{\beta}_{+} = 2\delta^{\beta}_{\alpha}$. Their Poincaré duals can then be used in the expansion (6.45). Suppose for instance that $\{\tilde{\Sigma}^{\alpha}_{+}\}_{\beta} \subset \{\tilde{\Sigma}^{\mu}\}_{\mu}$. Then the minimal 4d instantons charges of the compactification will be given by Euclidean D2-branes wrapping each of the $\tilde{\Sigma}^{\alpha}_{+}$, whose worldvolume gauge group may be either O(1) or USp(2). Whenever a D2-brane wrapping $\tilde{\Sigma}^{\alpha}_{+}$ yields a gauge group O(1), Dirac quantisation implies that the dual 4d strings are made up from

	$\sigma^{-\frac{3}{2}}$	σ^{-1}	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{3}}$	$\sigma^{-\frac{1}{6}}$
$\mathcal{T}_{ m st}^{1/2}$			NS5		
m_*	M-th. KK – Case 1	M-th. KK – Case 2	M-th. KK – Case 3		
CY $m_{\rm KK}$			CY $m_{\rm KK}$		
$E_{\rm mem}$	D2 - Case 1	D2 - Case 2	D2 - Case 3		
$\mathcal{T}_{ m mem}^{1/3}$			D2 - Case 1	D2 - Case 2	D2 - Case 3

Table 4. Mass scalings along the flows generated by NS5-strings in IIA orientifold models.

D4-branes wrapping the odd three-cycles $\Sigma_{\alpha}^{-} \equiv \Sigma_{\alpha} - \iota(\Sigma_{\alpha})$. Alternatively, if the gauge group is USp(2) then we may choose $\Sigma_{\alpha}^{-} \equiv \Sigma_{\alpha}$ whenever $\iota(\Sigma_{\alpha}) = -\Sigma_{\alpha}$, but then it is more convenient to redefine $\tilde{\Sigma}_{+}^{\alpha} \to 2\tilde{\Sigma}_{+}^{\alpha}$ to account for the double instanton charge. In both cases we end up with an even intersection number $\Sigma_{\alpha}^{-} \cdot \tilde{\Sigma}_{+}^{\beta} = 2\delta_{\alpha}^{\beta}$ in the covering space.

We can now discuss each sector from the viewpoint proposed in the present paper.

Kähler moduli and NS5 string flows

The properties of this sector are very similar to those discussed in the heterotic case, and so we will be short. One minor technical difference is due to the orientifold projection. So, we can identify $\Delta_{\rm K}$ with the orientifold-odd projection $\mathcal{K}(X)_{-}$ of the Calabi-Yau Kähler cone $\mathcal{K}(X)$. The corresponding BPS instantons are provided by orientifold-odd holomorphic curves, while EFT strings correspond to NS5-branes wrapping even nef divisors (which are assumed to be also effective, as in [40]):

$$\mathcal{C}_{\mathrm{S}}^{\mathrm{K}} = \mathrm{Nef}(X)_{\mathbb{Z}}^{+}.$$
(6.54)

Furthermore, if $D \simeq e^a D_a^+$ is a generic effective divisor, by using (6.51) one can easily check that the corresponding BPS string tension $\mathcal{T}_{\mathbf{e}} = \frac{\pi e^{2A}}{2l_s^2} \int_D e^{-2\phi} J \wedge J$ obtained by dimensional reduction matches the EFT formula $\mathcal{T}_{\mathbf{e}} = M_P^2 e^a \ell_a$.

Note also the analogy between (6.50) and (6.9). Hence, the dilaton diverges along any flow generated by these EFT strings as in the heterotic case, and the entire discussion on scaling of the relevant UV masses presented for that case can be applied *verbatim* also in this case. So we refer the reader to the previous subsection for details and just emphasise that Conjecture 2 is realised, with possible scaling weights w = 1, 2, 3.

Let us however point out that the membrane spectrum of the heterotic and IIA models is quite different. In particular, in IIA the lightest membranes correspond to D2-branes. However, this fact does not affect the universality of the asymptotic behaviour of the relevant mass scales. We can then summarise the relevant mass scalings with table 4.

Complex structure sector

The description of the saxionic cone factor $\hat{\Delta}$ appearing in (6.46) is more subtle. Naively, this should identify the possible values of $\hat{s} = e^{-\Phi}[\operatorname{Re}\Omega] \in H^3(X,\mathbb{R})_+$. From this definition, it is clear that if $\hat{s} \in \hat{\Delta}$, then also $\lambda \hat{s} \in \hat{\Delta}$ for any $\lambda > 0$. Hence, at least classically, $\hat{\Delta}$ is indeed a cone. Furthermore, by extending the results of [103] to our orientifolded setting,

the moduli spaces described by $\hat{\mathbf{s}}$ can be locally identified with an open neighbourhood of $H^3(X,\mathbb{R})_+$. On the one hand, by combining this observation with the conical structure, we can imagine to subdivide $\hat{\Delta}$ into the union of open subcones of $H^3(X,\mathbb{R})_+$. On the other hand, a global definition of $\hat{\Delta}$ as in (4.5) is not obvious. Part of the complication is due to the fact that, at large volume, the BPS instanton sector $\hat{\mathcal{C}}_{I}$ corresponds to Euclidean D2branes wrapping internal special Lagrangian cycles calibrated by $e^{-\phi} \text{Re}\Omega$. Differently from previous examples, such instantonic branes can violate the BPS condition at some point of the accessible saxionic domain, by crossing a stability wall. As in [105], one may adapt the definition of [106, appendix B] to our setting and distinguish two kinds of stability walls: i) threshold stability walls, where the three-cycle splits into mutually BPS three-cycles and *ii*) marginal stability walls, where such constituents are not mutually BPS. Applying the general scheme of section 4.1, we define the set of instanton charges $\hat{\mathcal{C}}_{I}$ in terms of the three-cycles that at most cross walls of threshold stability along the asymptotic limits contained in $\hat{\Delta}$. The non-trivial part of this definition is that it should be compatible with the initial choice of $\hat{\Delta}$ through (4.5). In the following we will assume that such compatible pair $(\hat{\mathcal{C}}_{I}, \hat{\Delta})$ can be found, as one can check in some simple examples.

Regarding 4d strings, BPS charges $\hat{\mathbf{e}} \in C_{\mathrm{S}}$ correspond to D4-branes wrapping SLag three-cycles calibrated by $-e^{-\phi}\mathrm{Im}\,\Omega$, with vanishing worldvolume flux. Their tension is given by

$$\mathcal{T}_{\hat{\mathbf{e}}} = -\frac{\pi e^{2A}}{l_{\rm s}^2} \int_{\Sigma} e^{-\phi} \mathrm{Im}\,\Omega = M_{\rm P}^2 \,\hat{e}^{\alpha} \hat{\ell}_{\alpha}\,. \tag{6.55}$$

Now, just like for BPS instantons, the SLag condition could be satisfied only at some sublocus of the saxionic domain. However, a BPS string solution like that of section 2 requires that $\hat{\mathbf{e}}$ remains BPS along its own flow, which in particular constrains the class of three-cycles that can host an EFT string. In particular, crossing a marginal stability wall along a string flow would signal an inconsistency, as the system would leave the BPS locus. If the string is non-EFT, non-perturbative effects unaccounted for in the solution of section 2 could in principle fix this apparent inconsistency, but for the case of EFT strings this seems unlikely to happen. We therefore conclude that a necessary condition for EFT strings made up of D4-branes wrapping SLag three-cycles is that they can only cross threshold stability walls along their flow. This conclusion provides yet another constraint on the choice of $\hat{\Delta}$, through the definition (4.12), or equivalently (4.17).

Unfortunately, our current general understanding of this saxionic sector, and of the corresponding EFT string sector, is quite limited. Hence we will illustrate the above observations and discuss the EFT string flows in a simple concrete toroidal model. There we will indeed see that indeed any EFT string charge populating $C_{\rm S}^{\rm EFT}$ admits a SLag representative at any point in $\hat{\Delta}$. We believe that this is not a lucky accident, but rather a general property of the EFT string sector.

Toroidal orbifolds. Let us consider the case where $X = (T^2 \times T^2 \times T^2)/\Gamma$, with $\Gamma \in SU(3)$ some orbifold action, and an orientifold involution of the form $\iota : (z_1, z_2, z_3) \to (\bar{z}_1, \bar{z}_2, \bar{z}_3)$. For simplicity, we will focus on the case where $\Gamma = \mathbb{Z}_2 \times \mathbb{Z}'_2$, with the choice of discrete torsion of [107], and a choice of complex structure of the form $\tau_i = i \frac{R_{2i}}{R_{2i-1}}$. Nevertheless, our discussion can be easily generalised to other cases, see e.g. [108–110] for reviews.

With these choices, the holomorphic (3, 0)-form reads

$$\Omega = R_1 R_3 R_5 dz_1 \wedge dz_2 \wedge dz_3 = (R_1 dy_1 + i R_2 dy_2) \wedge (R_3 dy_3 + i R_4 dy_4) \wedge (R_5 dy_5 + i R_6 dy_6).$$
(6.56)

A basis for even three-form classes $[\Sigma_{\alpha}^{-}]$ is given by

$$\begin{bmatrix} \Sigma_0^- \end{bmatrix} = 4 \mathrm{d} y_1 \wedge \mathrm{d} y_3 \wedge \mathrm{d} y_5, \qquad \begin{bmatrix} \Sigma_1^- \end{bmatrix} = -4 \mathrm{d} y_1 \wedge \mathrm{d} y_4 \wedge \mathrm{d} y_6, \begin{bmatrix} \Sigma_2^- \end{bmatrix} = -4 \mathrm{d} y_2 \wedge \mathrm{d} y_3 \wedge \mathrm{d} y_6, \qquad \begin{bmatrix} \Sigma_3^- \end{bmatrix} = -4 \mathrm{d} y_2 \wedge \mathrm{d} y_4 \wedge \mathrm{d} y_5,$$
(6.57)

while one for odd three-form classes $[\tilde{\Sigma}^{\beta}_{+}]$ is

$$[\tilde{\Sigma}^0_+] = -2\mathrm{d}y_2 \wedge \mathrm{d}y_4 \wedge \mathrm{d}y_6 , \qquad [\tilde{\Sigma}^1_+] = 2\mathrm{d}y_2 \wedge \mathrm{d}y_3 \wedge \mathrm{d}y_5 , [\tilde{\Sigma}^2_+] = 2\mathrm{d}y_1 \wedge \mathrm{d}y_4 \wedge \mathrm{d}y_5 , \qquad [\tilde{\Sigma}^3_+] = 2\mathrm{d}y_1 \wedge \mathrm{d}y_3 \wedge \mathrm{d}y_6 .$$

$$(6.58)$$

Notice that their Poincaré duals satisfy $\Sigma_{\alpha}^{-} \cdot \tilde{\Sigma}_{+}^{\beta} = 2\delta_{\alpha}^{\beta}$. The relative factor of 2 in between these two basis takes into account that D2-branes wrapping the even three-cycles $\tilde{\Sigma}_{+}^{\beta}$ dual to (6.58) yield O(1) instantons. By using this basis in the expansion (6.45), we can identify $\hat{a}^{\alpha} = \frac{1}{8} \int_{T^6} C_3 \wedge [\tilde{\Sigma}_{+}^{\alpha}]$ and

$$\hat{s}^{0} = \frac{1}{4}e^{-\phi}R_{1}R_{3}R_{5}, \qquad \hat{s}^{1} = \frac{1}{4}e^{-\phi}R_{1}R_{4}R_{6},
\hat{s}^{2} = \frac{1}{4}e^{-\phi}R_{2}R_{3}R_{6}, \qquad \hat{s}^{3} = \frac{1}{4}e^{-\phi}R_{2}R_{4}R_{5}.$$
(6.59)

The saxions \hat{s}^{α} measure the internal volume of O(1) instantons wrapping calibrated threecycles, while $2\hat{s}^{\alpha}$ measure the volume of U(1) instantons. Similarly, $2\hat{a}^{\alpha}$ would be the axion with unit periodicity in the unorientifolded theory, while $\operatorname{Re} \hat{t}^{\alpha} = \hat{a}^{\alpha}$ is the actual unit periodicity axion upon orientifolding. In terms of these fields we have that

$$\mathcal{H}(\hat{s}) = \frac{i}{32} \int_{T^6} e^{-2\phi} \Omega \wedge \bar{\Omega} = 8\sqrt{\hat{s}^0 \hat{s}^1 \hat{s}^2 \hat{s}^3}, \qquad (6.60)$$

and so the Kähler potential reads

$$K = -\sum_{\alpha=0}^{3} \log \operatorname{Im} \hat{t}^{\alpha} + \dots$$
(6.61)

where we have omitted the piece for the Kähler moduli, which will play no role in the following. The dual saxions are then given by

$$\hat{\ell}_{\alpha} = \frac{1}{2\hat{s}^{\alpha}} \,. \tag{6.62}$$

From here it follows a very simple structure for the saxionic cones of this sector

$$\hat{\Delta} = \left\{ (\hat{s}^0, \hat{s}^1, \hat{s}^2, \hat{s}^3) \in \mathbb{R}^4 | \hat{s}^0, \hat{s}^1, \hat{s}^2, \hat{s}^3 > 0 \right\},$$
(6.63a)

$$\hat{\mathcal{P}} = \left\{ (\hat{\ell}_0, \hat{\ell}_1, \hat{\ell}_2, \hat{\ell}_3) \in \mathbb{R}^4 | \hat{\ell}_0, \hat{\ell}_1, \hat{\ell}_2, \hat{\ell}_3 > 0 \right\},$$
(6.63b)

and it is instructive to see how the cone structure for 4d instantons and strings discussed in section 4.1 arises in this context. As discussed above both objects arise by wrapping D2and D4-branes, respectively, on calibrated three-cycles.¹⁸ As usual in these kinds of models one may obtain a good idea of the spectrum of BPS objects by considering factorisable three-cycles of the form

$$\Pi = 2(n^1, m^1)(n^2, m^2)(n^3, m^3), \qquad n^i, m^i \in \mathbb{Z}$$
(6.64)

where we follow the notation of [107].¹⁹ If this three-cycle is not invariant under the orientifold projection one needs to add the orientifold image, which lies at

$$\iota(\Pi) = 2(n^1, -m^1)(n^2, -m^2)(n^3, -m^3).$$
(6.65)

Consider a Euclidean D2-brane wrapping the even three-cycle $\tilde{\Sigma} = \Pi + \iota(\Pi)$. Both Π and $\iota(\Pi)$ will be simultaneously BPS if $\int_{\Pi} e^{-\phi} \operatorname{Re}\Omega > 0$ and $\int_{\Pi} e^{-\phi} \operatorname{Im}\Omega = 0$. The second condition is equivalent to

$$\tilde{\varepsilon} \equiv m^1 m^2 m^3 \hat{\ell}_0 - m^1 n^2 n^3 \hat{\ell}_1 - n^1 m^2 n^3 \hat{\ell}_2 - n^1 n^2 m^3 \hat{\ell}_3 = 0.$$
(6.66)

In the generic case

$$I_{\Pi} \equiv \Pi \cdot \iota(\Pi) = 8n^{1}n^{2}n^{3}m^{1}m^{2}m^{3}$$
(6.67)

does not vanish and then (6.66) describes a marginal stability wall. This can for instance be seen by adapting to the present context the Fayet-Iliopoulos analysis of [112], and geometrically it boils down to applying the angle theorem [113, 114] to the present setup. On the one hand, in the region of moduli space in which $I_{\Pi}\tilde{\varepsilon} < 0$ an open string tachyonic mode will develop between the D2-branes, and they will recombine into a single smooth object representing $\tilde{\Sigma}$. On the other hand for $I_{\Pi}\tilde{\varepsilon} > 0$ the system will be non-BPS and $\tilde{\Sigma}$ will not have a BPS representative. Instantons of this sort will not belong to the instanton cone $\hat{\mathcal{C}}_{I}$ defined above, as for any Π with $I_{\Pi} \neq 0$ there will be some asymptotic limit crossing the marginal stability wall. The way to construct factorisable cycles that belong to $\hat{\mathcal{C}}_{\mathrm{I}}$ is to set $I_{\Pi} = 0$, while choosing wrapping numbers n^i, m^i such that (6.66) has non-trivial solutions. For instance one may fix $m^1 = 0$, and then $n^1, n^2, n^3, m^2 > 0, m^3 < 0$. The resulting system has a wall of threshold stability at $|m^2n^3|\hat{\ell}_2 = |n^2m^3|\hat{\ell}_3$, and at both sides of the wall Π and $\iota(\Pi)$ recombine into a single BPS three-cycle. Finally, one can construct the simplest elements of $\hat{\mathcal{C}}_{I}$ by choosing wrapping numbers such that all the coefficients in (6.66) vanish, while still imposing that $\int_{\Pi} e^{-\phi} \operatorname{Re} \Omega > 0$. It is easy to see that, if the most general D2-brane instanton charge is of the form $[\tilde{\Sigma}] = \hat{m}_{\alpha}[\tilde{\Sigma}^{\alpha}_{+}]$, then the cone of instanton charges $\hat{\mathcal{C}}_{I}$ reads

$$\hat{\mathcal{C}}_{\mathrm{I}} = \left\{ \hat{\mathbf{m}} = (\hat{m}_0, \hat{m}_1, \hat{m}_2, \hat{m}_3) \in \mathbb{Z}_{\geq 0}^4 \right\} \,. \tag{6.68}$$

¹⁹More precisely, a three-cycle $(n^1, m^1)(n^2, m^2)(n^3, m^3)$ corresponds, in the covering space T^6 , to the class

$$\Pi = \left(n^{1}a_{1} + m^{1}a_{2}\right) \times \left(n^{2}a_{3} + m^{2}a_{4}\right) \times \left(n^{3}a_{5} + m^{3}a_{6}\right),$$

¹⁸One could also consider D4-branes and D6-branes wrapping coisotropic five-cycles, similarly to the construction of [111]. Since a priori these objects do not result into new 4d charges, we will not consider them in the following.

where a_i is the one-cycle class of T^6 along the coordinate y_i .

Again, these are not the most general charges for a $\frac{1}{2}$ BPS instanton, but those charges that do not belong to (6.68) will not be BPS everywhere on the perturbative asymptotic region associated with saxionic cone $\hat{\Delta}$.

A similar story applies to 4d strings made up from D4-branes wrapping an odd threecycle of the form $\Sigma = \Pi - \iota(\Pi)$. Now the simultaneous BPS conditions for Π and $-\iota(\Pi)$ will amount to $\int_{\Pi} e^{-\phi} \operatorname{Im} \Omega < 0$ and $\int_{\Pi} e^{-\phi} \operatorname{Re} \Omega = 0$, the latter condition being equivalent to

$$\varepsilon \equiv n^1 n^2 n^3 \hat{s}^0 - n^1 m^2 m^3 \hat{s}^1 - m^1 n^2 m^3 \hat{s}^2 - m^1 m^2 n^3 \hat{s}^3 = 0.$$
 (6.69)

This again indicates the locus of a marginal stability wall for the generic case $I_{\Pi} \neq 0$. The total string charge of this system is given by

$$[\Sigma] = \hat{e}^{\alpha}[\Sigma_{\alpha}^{-}] = m^{1}m^{2}m^{3}[\Sigma_{0}^{-}] - m^{1}n^{2}n^{3}[\Sigma_{1}^{-}] - n^{1}m^{2}n^{3}[\Sigma_{2}^{-}] - n^{1}n^{2}m^{3}[\Sigma_{3}^{-}], \quad (6.70)$$

so that along the flow (4.9) generated by these charges the stability parameter ε varies as

$$\varepsilon(r) = \varepsilon_0 + \frac{1}{2} I_{\Pi} \,\sigma(r) \,. \tag{6.71}$$

Therefore, along the flow of a D4-brane with charges (6.70), the quantity $I_{\Pi}\varepsilon$ will increase its value until the wall of marginal stability is crossed and the system becomes non-BPS. Clearly, such a pathological behaviour is not admissible for an EFT string, and one should look for a more specific set of charges. As for instantons, this is achieved by considering factorisable three-cycles such that $I_{\Pi} = 0$, and so the D4-branes cross walls of threshold stability or no stability wall at all. If we characterise the 4d string charges of this sector as $\hat{e}^{\alpha}[\Sigma_{\alpha}^{-}]$, this procedure selects the following cone

$$\mathcal{C}_{\rm S}^{\rm EFT} = \left\{ \hat{\mathbf{e}} = (\hat{e}^0, \hat{e}^1, \hat{e}^2, \hat{e}^3) \in \mathbb{Z}_{\geq 0}^4 \right\} \,, \tag{6.72}$$

which coincides with the definition (4.12). In fact, due to the simplicity of this setup, we have that $\overline{\Delta} = \mathcal{P}^{\vee}$ and so $\mathcal{C}_{S}^{EFT} = \mathcal{C}_{S}$.

Let us consider a few simple string flows in this setup, in order to check Conjecture 2. For instance one may consider one of the generators of the cone (6.72), like for instance $\hat{\mathbf{e}} = (1, 0, 0, 0)$. This corresponds to a D4-brane wrapping the three-cycle $\Sigma = \Pi - \iota(\Pi)$, with $\Pi = 2(0, 1)(0, 1)(0, 1)$. Together they generate the elementary flow $\hat{s}^0 = \hat{s}_0^0 + \sigma$, while $\hat{s}^1, \hat{s}^2, \hat{s}^3$ and the Kähler moduli remain fixed. It is clear that $\mathcal{T}_{\hat{\mathbf{e}}} = M_{\mathrm{P}}^2 \hat{\ell}_0 \sim M_{\mathrm{P}}^2 \sigma^{-1}$, as in (4.14). Furthermore, $e^{2\phi}, e^{2A} \sim \sigma^{-\frac{1}{2}}$ and the radii evolve as $R_1, R_3, R_5 \sim \sigma^{\frac{1}{4}}$ and $R_2, R_4, R_6 \sim \sigma^{-\frac{1}{4}}$. Since we are drawn to a weak coupling limit, the lightest tower of modes correspond to KK states along the larger radii R_1, R_3, R_5 or winding modes along the smaller radii R_2, R_4, R_6 . In both cases we obtain $m_*^2 \sim M_{\mathrm{P}}^2 \sigma^{-1}$, realising Conjecture 2 with w = 1. The result is summarised in table 5. The same result applies to other elementary flows.

Non-elementary flows may be constructed by considering bound states of the above elementary charges, or also by wrapping D4-branes on $\Sigma = \Pi - \iota(\Pi)$ with Π factorisable. For instance we can take $\Pi = 2(0, 1)(n^2, m^2)(-n^3, m^3)$ with $n^2, n^3, m^2, m^3 > 0$. Together with its orientifold image, this system corresponds to the string charges $\hat{\mathbf{e}} = (m^2m^3, n^2n^3, 0, 0)$,

	$1/\sigma^{1/2}$	$1/\sigma^{1/6}$	$\sigma^{1/6}$	$\sigma^{1/2}$
$\mathcal{T}_{ m str}^{1/2}$	D4			
m_*	$m_{ m KK}$			
$E_{\rm mem}$	$\mathrm{D}p$			NS5
$\mathcal{T}_{ m mem}^{1/3}$		$\mathrm{D}p$	NS5	

Table 5. Elementary flow $\hat{s}^i \to \infty$.

and as discussed above it only crosses wall of threshold stability along $\hat{\Delta}$. It generates the flow

$$\hat{s}^{0} = \hat{s}^{0}_{0} + m^{2}m^{3}\sigma, \qquad \hat{s}^{1} = \hat{s}^{1}_{0} + n^{2}_{b}n^{3}_{b}\sigma, \qquad (6.73)$$

while \hat{s}^2 , \hat{s}^3 and the Kähler moduli remain fixed. Again one can check that $\mathcal{T}_{\hat{\mathbf{e}}}$ scales as in (4.14). Furthermore $e^{2\phi} \sim e^{2A} \sim \sigma^{-1}$, the first two radii scale as $R_1 \sim \sigma^{\frac{1}{2}}$, $R_2 \sim \sigma^{-\frac{1}{2}}$ while the other four radii remain constant. The lightest tower of the compactification again corresponds to KK states or winding modes, now along the radii R_1 and R_2 respectively. In both cases $m_*^2 \sim M_{\rm P}^2 \sigma^{-2}$ realising Conjecture 2 with w = 2. Similar results are obtained for non-elementary flows of this sort, and even bound states of them. Some possibilities are collected in table 6.

Notice that in both of these examples one may consider BPS D2-brane instantons on cycles $\tilde{\Sigma} = \tilde{\Pi} + \iota(\tilde{\Pi})$ with $\tilde{\Pi}$ of the form (6.64), and such that $\langle \mathbf{m}, \mathbf{e} \rangle \equiv \frac{1}{2} \Sigma \cdot \tilde{\Sigma}$ is negative. However, if that is the case, the marginal stability parameter $I_{\tilde{\Pi}}\tilde{\varepsilon}$ will grow along the corresponding string flow, and the instanton will become non-BPS. Following the general philosophy of section 4.1 we neglect their effect, which is encoded in the fact that they do not belong to \hat{C}_{I} . It would be interesting to confirm this expectation by taking into account their precise effect on the EFT, possibly along the lines of [105].

Finally, we may consider 4d membranes made up of D*p*-branes wrapping even cycles, whose volume stays constant along each string flow. Their tension \mathcal{T}_{mem} then evolves as $e^{3A-\phi}l_s^{-3}$ and one can easily check that it scales in agreement with (5.7). For the case of an elementary flow, this implies $E_{\text{mem}} \sim m_* \sim \sigma^{-1/2}$, i.e. n = w = 1, as shown in table 5.

6.5 M-theory on G_2 spaces

We finally consider an M-theory compactification on a smooth G_2 -holonomy space X — see for instance [115, 116] for a summary on the relevant quantities defining the corresponding EFT. The 11d space-time metric reads

$$ds^2 = e^{2A} ds_4^2 + l_M^2 ds_X^2 , \qquad (6.74)$$

where ds_X^2 is the dimensionless G_2 -holonomy metric, l_M is the M-theory Planck length appearing in the 11d Einstein-Hilbert term $\frac{2\pi}{l_M^9} \int R$. The Weyl rescaling necessary to a get 4d Einstein frame metric is

$$e^{2A} = \frac{l_{\rm M}^2 M_{\rm P}^2}{4\pi V_X} \,, \tag{6.75}$$

	GHI
5	EP0
; ;	9
)	
;	21
1	
	7

String charge	w = 1	w = 2	w = 3
$(e^1, 0, 0, 0, 0, 0, 0) \\$	M-th. KK		
$(e^1, e^2, 0, 0, 0, 0, 0)$		M-th. KK	
$(e^1, e^2, e^3, 0, 0, 0, 0)$			M-th. KK
$(0,0,0,\hat{e}^0,0,0,0)$	$m_{ m KK},m_{ m w}$		
$(0,0,0,\hat{e}^0,\hat{e}^1,0,0)$			
$(0,0,0,\hat{e}^0,\hat{e}^1,\hat{e}^2,0)$		$m_{ m KK},m_{ m w}$	
$(0,0,0,\hat{e}^0,\hat{e}^1,\hat{e}^2,\hat{e}^3)$			
$(e^1,0,0,\hat{e}^0,0,0,0)$		$m_{ m KK}$	
$(e^1, e^2, 0, \hat{e}^0, 0, 0, 0)$		$m_{\rm KK}$, M-th. KK	
$(e^1, e^2, e^3, \hat{e}^0, \hat{e}^1, \hat{e}^2, \hat{e}^3)$			$m_{\rm KK}$, M-th. KK

Table 6. Behavior and identification of m_* in comparison with the string tension for some choices of string charges. We have collected ($\mathbf{e}, \hat{\mathbf{e}}$) with the components $e^a, \hat{e}^\alpha \in \mathbb{Z}_{>0}$. Strings with charge \mathbf{e} drive a flow within the Kähler moduli sector; strings with charges $\hat{\mathbf{e}}$ trigger a flow involving the dilaton and the complex structure moduli. The scalings are invariant for permutations of NS5 string charges e^a or D4 string charges \hat{e}^α . It can be checked that other mixed linear combinations of \mathbf{e} and $\hat{\mathbf{e}}$ do not deliver scaling weights w > 3.

where V_X is the internal volume in l_M -units. The information contained in the G_2 metric is completely encoded in the corresponding associative 3-form Φ , normalised so that

$$\frac{1}{7} \int_X \Phi \wedge *\Phi = V_X \,. \tag{6.76}$$

Here $*\Phi$ can be identified with the co-associative four-form and can be written as a function of Φ . The G_2 holonomy condition is equivalent to imposing $d\Phi = 0$ and $d*\Phi = 0$. In a smooth large volume regime, all chiral coordinates are of the (s)axionic type: the chiral fields $t^i \equiv a^i + is^i$ are identified by expanding the complex combination $[A_3 + i\Phi] \in$ $H^3(X, \mathbb{C})$, where A_3 is the (flat) M-theory potential, into an integral cohomology basis

$$\boldsymbol{t} = \boldsymbol{a} + \mathrm{i}\boldsymbol{s} \equiv [A_3 + \mathrm{i}\Phi] = t^i[\Sigma_i] \in H^3(X, \mathbb{C}), \qquad (6.77)$$

where $[\Sigma_i]$ are Poincaré dual to a basis of four-cycles $\Sigma_i \in H_4(X; \mathbb{Z})_{t.f.}$, with the subscript "t.f." selecting the torsion free part. Up to irrelevant constants, the Kähler potential is given by

$$K = -3\log \int_X \Phi \wedge *\Phi \,. \tag{6.78}$$

Here both Φ and $*\Phi$ must be considered as functions of the saxions s^i . The dual saxions ℓ_i defined in (2.17) are given by the expansion

$$\boldsymbol{\ell} = \ell_i[\tilde{\Sigma}^i] = \frac{1}{2V_X}[*\Phi] \in H^4(X, \mathbb{R}), \qquad (6.79)$$

where $\tilde{\Sigma}^i$ is a basis of three-cycles dual to the four-cycles Σ_i , such that $\Sigma_i \cdot \tilde{\Sigma}^j = \delta_i^j$.

Strings, instantons and cone structure

The effective four-dimensional BPS instantons and strings correspond to M2-branes wrapping associative (i.e. Φ -calibrated) three-cycles C and M5-branes wrapping coassociative (i.e. $*\Phi$ -calibrated) four-cycles S, see for instance [117] for a recent discussion. In order to make explicit the link with our general discussion of section 4, we must make the identifications $M_{\mathbb{Z}} = H_3(X,\mathbb{Z})_{t.f.}$ and $N_{\mathbb{Z}} = H_4(X,\mathbb{Z})_{t.f.}$. The BPS instanton and string charges take then values in

$$\mathbf{m} \equiv [C] = m^i [\Sigma_i] \in H_3(X, \mathbb{Z})_{\text{t.f.}}, \qquad \mathbf{e} \equiv [S] = e_i [\tilde{\Sigma}^i] \in H_4(X, \mathbb{Z})_{\text{t.f.}}.$$
(6.80)

By direct dimensional reduction one can check that the action of a Euclidean M2-brane wrapping C takes the form $S_{M2} = -2\pi i \langle \mathbf{m}, \boldsymbol{t} \rangle$ and then generates corrections of the form (4.3), as in [118]. Similarly, wrapping an M5-brane on S one gets a string with tension $\mathcal{T}_{\mathbf{e}} = M_{\mathrm{P}}^2 \langle \mathbf{e}, \boldsymbol{\ell} \rangle$, in agreement with the expected 4d EFT formula.

However, in order to identify $C_{I} \subset H_{3}(X,\mathbb{Z})_{t.f.}$ and $C_{S} \subset H_{4}(X,\mathbb{Z})_{t.f.}$ one may encounter the M-theory counterpart of the walls of marginal stability discussed for the IIA models. In order to address this issue one first needs to better specify the saxionic space. Unfortunately, differently from the classical moduli of Calabi-Yau Kähler structures, not so much is known about the global structure of the moduli space of classical G_2 -holonomy structures. It is known that this moduli space can be locally identified with an open subset of $H^3(X,\mathbb{R})$ [119], parametrised by the saxions $s \equiv [\Phi]$. This local subset can always be extended to a conical one, since an overall constant positive rescaling of the metric preserves the G_2 -holonomy and rescales Φ (although as we approach the 'tip' of the cone the classical approximation clearly breaks down). Regarding the global structure, it is known that $s \equiv [\Phi]$ must satisfy a list of positivity conditions, which have been enumerated in [117] — see eqs. (3.5)-(3.9) therein. In the same paper, motivated by the analogy with the Kähler case, it was suggested that these conditions could be actually sufficient to completely characterise the G_2 -holonomy space. One of these conditions is

$$\langle \mathbf{m}, \mathbf{s} \rangle = \int_C \Phi > 0 \quad \text{with} \quad \mathbf{m} \equiv [C],$$
 (6.81)

for any C admitting an associative representative. From our perspective, this condition is naturally related to our definition of saxionic cone (4.5). However, as already point out in [117], the existence of an associative representative can in fact be Φ -dependent. This issue is just the M-theory counterpart of the walls of mariginal stability encountered in the IIA models. According to our general prescription, we rather assume that we can pick a saxionic cone Δ defined by (6.81) for any three-cycle C asymptotically admitting an associative representative along any infinite distance limit inside Δ . The maximal set of BPS instanton charges $C_{\rm I}$ is then defined as in (4.6).

It would be clearly important to test and possibly make more precise this definition of saxionic cone. Unfortunately, to our knowledge, the present understanding of the moduli space of G_2 -holonomy spaces is still too poor to address these issues. For instance, even the convexity of the domain of the possible values of saxions defined by $s \in [\Phi]$ is not obvious at all. Furthermore, the deep interior of Δ may include regimes in which the

classical geometrical description of the compactification space breaks down, still preserving the approximate axionic symmetry. Δ may in fact cover different G_2 -manifolds, related by phase transitions which do not affect the approximate shift symmetry. We have already encountered similar effects in the IIB/F-theory context in section 6.3.²⁰

In the present setting, EFT strings should correspond to M5-branes wrapping coassociative 4-cycles S which have non-negative intersection number with all three-cycles corresponding to BPS instanton charges: $C \cdot S = \langle \mathbf{m}, \mathbf{e} \rangle \ge 0$, for any $\mathbf{m} = [C] \in \mathcal{C}_{\mathrm{I}}$. Notice that (6.78) is again of the form (3.8), with P(s) homogeneous of degree 7. Hence, as in all our previous examples, the string tension of such EFT strings tends to zero along the corresponding RG flows, and it is interesting to compare it with other asymptotically vanishing energy scales. Indeed, we expect the appearance of light KK modes since the volume of the associative three-cycles C with strictly positive $C \cdot S > 0$ become infinite along the string flow of charge $\mathbf{e} = [S]$. We now illustrate these effects by focusing on Joyce's models — see also [18] for related results in TGS models.

Joyce's model

We now analyse these effects in Joyce's compact models [119, 123, 124], which are obtained as the resolutions X of toroidal orbifolds T^7/Γ . More precisely, Γ should preserve the associative three-form

$$\Phi = \eta_{123} + \eta_{145} + \eta_{167} + \eta_{246} - \eta_{257} - \eta_{347} - \eta_{356} \,. \tag{6.82}$$

Here $\eta_{abc} \equiv \eta_a \wedge \eta_b \wedge \eta_c$ where $\eta_a = R_a dy_a$ (no sum over repeated indices) is the vielbein of T^7 parametrised by periodic coordinates $y_a \simeq y_a + 1$, $a = 1, \ldots, 7$. The radii R_a must be restricted in order to be compatible with the Γ projection and parametrise the 'untwisted' moduli sector. The corresponding saxions measure the volume of associative three-cycles C obtained as projection of Γ -invariant T^3 's inside T^7 , which contribute to C_1 . Analogously, the projection of Γ -invariant T^4 's inside T^7 contribute C_S . The moduli space and the spectrum of BPS instantons and strings receives contributions also from the 'twisted sector' coming from the resolution of the orbifold singularities. In order to better illustrate what can happen, let us focus on the concrete model in which $\Gamma = \mathbb{Z}_2 \times \mathbb{Z}_2 \times \mathbb{Z}_2$ whose generators $\langle \alpha, \beta, \gamma \rangle$ act on the toroidal coordinates in the following way

$$\begin{aligned} \alpha : (y_1, \dots, y_7) &\mapsto (y_1, y_2, y_3, -y_4, -y_5, -y_6, -y_7) ,\\ \beta : (y_1, \dots, y_7) &\mapsto (y_1, -y_2, -y_3, y_4, y_5, \frac{1}{2} - y_6, -y_7) ,\\ \gamma : (y_1, \dots, y_7) &\mapsto (-y_1, y_2, -y_3, y_4, \frac{1}{2} - y_5, y_6, \frac{1}{2} - y_7) . \end{aligned}$$

$$(6.83)$$

²⁰Take for instance the non-compact G_2 spaces constructed in [120, 121] and studied in [116, 122], which provide local descriptions of resolved codimension-seven singularities. The simplest examples include resolutions of cones which can be described as \mathbb{R}^3 fibrations over a coassociative base $S = S^4$ or $S = \mathbb{P}^2$. The singular conical geometries are recovered by taking $\operatorname{Vol}(S) \to 0$. In this limit no non-trivial three-cycle shrinks to zero-size (the associative three-cycles are rather given by the non-compact \mathbb{R}^3 fibres). Hence, we are still deep inside the saxionic cone Δ and the axionic symmetries are perturbatively preserved.

A detailed discussion of its properties can be found in [124], which we now summarise. First of all, it is immediate to check that the 7 three-forms appearing in (6.82) are separately invariant under (6.83). In fact, they generate $H^3(T^7/\Gamma, \mathbb{Z})$. Correspondingly, we can identify seven associative 3-cycles $C^a \simeq T^3$, $a = 1, \ldots, 7$. For instance, C^1 is parametrised by (y_1, y_2, y_3) (while the other coordinates are constant), C^2 is parametrised by (y_1, y_4, y_5) , and so forth. Analogously, one can construct seven coassociative 4-cycles S_a , calibrated by 7 four-forms appearing in $*\Phi$. These associative 3- and 4-cycles survive after the singularities have been repaired. Correspondingly, one can introduce seven 'untwisted' saxions

$$s^a = \int_{C^a} \Phi \,. \tag{6.84}$$

The resolution of the singularities provides 36 additional 3-cycles \tilde{C}^{α} (and 36 4-cycles). The singular locus of T^7/Γ is the disjoint union of 12 T^3 — take for instance the T^3 defined by $y^4 = y^5 = y^6 = y^7 = 0$. Around each of these T^3 orbifolds singularities T^7/Γ looks like $\mathbb{T}^3 \times \mathbb{C}^2/\mathbb{Z}_2$. One can then locally smooth-out each singularity by blowing-up a twosphere $l \simeq S^2$ at the origin of $\mathbb{C}^2/\mathbb{Z}_2$, getting a four-dimensional space S which has the same topology of the hyperkähler Eguchi-Hanson space. Around each resolved singularity X looks like $T^3 \times S$ and by taking all the possible combinations of the form $S^1_{(A)} \times l$, with $S^1_{(A)} \subset T^3$ for A = 1, 2, 3, one generates $3 \times 12 = 36$ additional 3-cycles \tilde{C}^{α} , and correspondingly, 36 additional saxionic coordinates:

$$\tilde{s}^{\alpha} = \int_{\tilde{C}^{\alpha}} \Phi \,. \tag{6.85}$$

Analogously, one can construct 36 additional 4-cycles \tilde{S}_{α} of the form $T^2 \times l$, with $T^2 \subset T^3$.

We will not try to fully explore the general structure of the saxionic cone, but rather focus on the EFT string flows (4.9) associated with purely 'untwisted' charges $\mathbf{e} = (e^1, \ldots, e^7)$. The twisted saxions \tilde{s}^{α} do not change along these flows and then are asymptotically subleading, as in the resolved $T^6/(\mathbb{Z}_2)^3$ F-theory model discussed in appendix F.2. Note that, since Φ is a calibration, $|\tilde{s}^{\alpha}|$ represents a lower bound for the volume of \tilde{C}^{α} , which is saturated if the \tilde{C}^{α} is calibrated. In this regime, one may then use the approximate EFT worked out in [125, 126], but do not actually need it. Indeed, in order to estimate the relevant light energy scales it is sufficient to focus on the leading contribution to the Kähler potential, which is just given by

$$K = -\log(s^1 \dots s^7) + \dots$$
 (6.86)

so the dual saxions read

$$\ell_a = \frac{1}{2s^a} \,. \tag{6.87}$$

This observation also holds for more general Joyce's models, not just the one discussed above.

As usual, the tension of an EFT-string scales according to the universal behaviour (4.14) along its own flow. Consider for instance a string originating an M5-brane
wrapping a four-cycle $\tilde{S} = e^a \tilde{S}_a$, where \tilde{S}_a is a basis of four-cycles dual to C^a . The string tension is given by

$$\mathcal{T}_{\mathbf{e}} = \frac{2\pi}{l_M^2} e^{2A} \int_{\tilde{C}} *\Phi = M_{\rm P}^2 e_a \ell^a \,. \tag{6.88}$$

where we have used (6.75) and (6.79).

In order to estimate the behavior of the other relevant energy scales is straightforward and proceeds as in the previous examples. The natural candidates for m_* are the lightest among KK, whose masses are set by

$$m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{R_*^2 V_X} \,,$$
 (6.89)

where R_* is the radius that grows faster along the string flow. However, the EFT breaking may be caused also by the light M2 branes wrapped on internal two-cycles. In order to estimate the corresponding mass scale, let us preliminary notice that, from (6.83), the orbifold should not allow for any non-trivial bulk two-cycles. Thus, M2 particles may only originate from the blown-up two-spheres, which combine with the torus S^{1} 's to form the 3-cycles \tilde{C}^{α} . As already noticed, $|\tilde{s}^{\alpha}|$ provides a lower bound for the volume of the corresponding 3-cycle \tilde{C}^{α} . It follows that, along the flows generated by untwisted charges, for any blown-up two-cycle l we can write a lower bound $\operatorname{Vol}(l) \geq \frac{c}{R_*}$ for some constant c. Since the mass of the corresponding wrapped M2-brane is given by $m_{M2} = 2\pi e^A \operatorname{Vol}(l)/l_M$, this bound implies that

$$m_{\rm M2}^2 \gtrsim \frac{c^2 M_{\rm P}^2}{R_*^2 V_X} \,.$$
 (6.90)

Comparing it to (6.89) we conclude that $m_{\rm M2}/m_{\rm KK} \gtrsim c$ and we can then identify m_* with the KK-mass $m_{\rm KK}$.

The explicit relation between the radii R_a and the saxions s^a is given in appendix F.3. Notice that the saxions can be written as products of three radii, since they parametrise the volume of three-cycles. This implies that the radii are homogeneous functions of degree 1/3. Using that the volume $V_X = \frac{1}{7}e^{-K/3}$ is an homogeneous function of degree 7/3 due to (6.86), we conclude that $R^2_*V_X$ is an homogeneous function of degree three. This implies that it scales asymptotically as

$$R_*^2 V_X \sim \sigma^w$$
, with $w \le 3$, (6.91)

along the string flow. As shown in appendix F.3, w is not only bounded by 3 but is actually integral: w = 1, 2, 3. Combined with (6.89), this determines the scaling weight of the flow as

$$m_*^2 = m_{\rm KK}^2 \simeq M_{\rm P}^2 \left(\frac{\mathcal{T}_{\mathbf{e}}}{M_{\rm P}^2}\right)^w \,, \tag{6.92}$$

realising Conjecture 2. The specific result for w depends on the choice of \mathbf{e} and is summarised by table 7 for the different string flows. It is easy to see that an elementary flow will have w = 1 as only one saxion is sent to infinity and therefore $R_*^2 V_X \sim \sigma$ (see table 8). Analogously, the non-elementary string flow in which all saxions are sent to infinity will

String charge	w = 1	w = 2	w = 3
\mathbf{e}_a	$m_{ m KK}$		
\mathbf{e}_{ab}			
$\mathbf{e}_{abc} otin \mathcal{C}_3$		$m_{ m KK}$	
$\mathbf{e}_{abcd} \in \mathcal{C}_4$			
$\mathbf{e}_{abc}\in\mathcal{C}_3$			
$\mathbf{e}_{abcd} otin \mathcal{C}_4$			
\mathbf{e}_{abcde}			$m_{\rm KK}$
\mathbf{e}_{abcdef}			
$\mathbf{e}_{abcdefg}$			

Table 7. Behavior of m_* in comparison with the string tension. Here we have introduced $\mathbf{e}_{a_1a_2...a_m} = c^1\mathbf{e}_{a_1} + c^2\mathbf{e}_{a_2} + \ldots + c^m\mathbf{e}_{a_m}$, where $c^a \in \mathbb{Z}_{>0}$ and \mathbf{e}_a denotes a basis of elementary string charges: $(\mathbf{e}_a)^b = \delta_a^b$. We have further introduced the sets $\mathcal{C}_3 = \{\mathbf{e}_{123}, \mathbf{e}_{145}, \mathbf{e}_{167}, \mathbf{e}_{246}, \mathbf{e}_{257}, \mathbf{e}_{347}, \mathbf{e}_{356}\}$ and $\mathcal{C}_4 = \{\mathbf{e}_{4567}, \mathbf{e}_{2367}, \mathbf{e}_{2345}, \mathbf{e}_{1367}, \mathbf{e}_{1346}, \mathbf{e}_{1256}, \mathbf{e}_{1247}\}$.

	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{6}}$
$\mathcal{T}_{ m str}^{1/2}$	M5	
m_*	$m_{\rm KK}$	
$E_{\rm mem}$	M2, M5	
$\mathcal{T}_{ m mem}^{1/3}$		M2, M5

 Table 8. Mass scalings along the flow generated by an elementary string obtained from a wrapping M5-brane.

have a scaling weight equal to the homogeneity degree of $R^2 V_X$, i.e. w = 3. The derivation of the scaling weight for other types of flows can be found in appendix F.3.

Finally, we will also have membranes coming from either M2 branes spanning three external spacetime directions or M5 branes wrapped on three-cycles \tilde{C} such that $\int_{\tilde{C}} \Phi$ remains constant along the string flow. The membrane scale is given by

$$E_{\rm mem} \sim e^{\frac{1}{2}K} M_{\rm P} \sim \mathcal{T}_{\rm e}^{n/2} \tag{6.93}$$

where n is an integer that can take values up to the homogeneity degree of e^{K} , i.e. $n \leq 7$. It is interesting to notice that in these M-theory examples, even if n can be up to seven, we still find that the scaling weight w is not bigger than three!

7 Conclusions

In this work we have studied the physics emerging at large field distances of 4d $\mathcal{N} = 1$ EFTs by characterising the asymptotic limits in terms of fundamental axionic BPS strings, dubbed EFT strings. Whenever such an EFT string is present in the theory, it induces a

classical backreaction profile on the saxionic partner to the axion under which the string is magnetically charged, that drives the saxion to infinite distance at the string core. Hence, such a backreaction profile can be mapped to a particular infinite distance boundary of the field space selected by the string charges. As in [38, 127] we can also interpret this backreaction profile as an RG flow of the brane couplings, which tells us how the string tension behaves as we change the cut-off of the theory. We arrive then to a further correspondence between string RG flows and infinite distance limits in field space such that going to the UV corresponds to probing large field distances. Whether the converse is true, meaning whether every infinite distance limit can be mapped to a string RG flow, is rather non-trivial and will depend on whether there is an axionic shift symmetry emerging at every infinite distance limit. Based on string theory evidence, we propose that this is indeed the case; a proposal that we encode in the following conjecture:

Distant Axionic String Conjecture (DASC): Every infinite field distance limit of a 4d EFT consistent with quantum gravity can be realised as an RG flow UV endpoint of an EFT string.

The RG flow triggered by an EFT string implies that the string tension goes to zero monotonically at the UV/large field distance limit. Hence, our conjecture implies the universal presence of a string becoming tensionless at every infinite distance limit. When the string tension \mathcal{T} becomes below the EFT cut-off-squared, the semiclassical description of the string breaks down and the EFT gets sensitive to the string excitation modes. Hence, $\mathcal{T}^{1/2}$ acts as the maximum value that the EFT cut-off could take, which occurs at a smaller energy scale the larger the distance we travel in field space. Therefore, this implies an upper bound on the field distance that can be described by an EFT with a fixed energy cut-off. If the string satisfies the WGC, we have also shown, following [32], that the tension behaves exponentially on the proper field distance. This is the key ingredient behind the derivation of the SDC from our proposal. In fact, this allows us to give a lower bound for the exponential rate of the SDC tower in 4d $\mathcal{N} = 1$ EFTs in terms of the string extremality factor. The same exponential behaviour arises if we use the typical asymptotic log-structure (3.8) of the Kähler potential arising in string compactifications instead of the WGC, which suggests that this log-structure might indeed be a universal quantum gravity feature at the asymptotic limits.

Our identification between string RG flows and infinite field distance limits also allow us to provide a peculiar realisation of the Emergence proposal [4–6, 56] as well as understanding the presence of the string (and consequently, the SDC) as a quantum gravity obstruction to restore a global symmetry at infinite distance in moduli space, as also proposed in [6, 17]. In addition to the continuous axionic shift symmetry which becomes exact at infinity, there would also be a U(1) 2-form global symmetry emerging as the gauge coupling of the 2-form gauge field goes to zero. Furthermore, we also show that the Conjecture 3 proposed in [1] about having negative scalar curvature is false in general, although a weaker version survives; namely, the existence of the string flow implies that points at infinity have at least one negative holomorphic sectional curvature. More generally, there could be other infinite tower of states becoming light at a faster rate at the asymptotic limit, which would therefore signal the EFT breakdown before it gets sensitive to the string modes. Interestingly, based again on the string theory evidence, we propose that the leading tower of states always satisfies the following relation to the string tension:

$$m_*^2 \simeq M_{\rm P}^2 A \left(\frac{\mathcal{T}}{M_{\rm P}^2}\right)^w$$
 for some positive integer $w = 1, 2, \dots$ (7.1)

for some constant A. In other words, we propose that each EFT string is characterised by an integral scaling weight w, which dictates the behaviour of m_* in terms of the string tension \mathcal{T} along the corresponding asymptotic flow.

We provide evidence for our two conjectures by analysing large classes of examples in $4d \mathcal{N} = 1$ string theory compactifications. This includes compactifications of heterotic on Calabi-Yau three-folds, Type IIA/B on orientifolds, F-theory on elliptically fibred Calabi-Yau four-folds and M-theory on certain G2 manifolds. Surprisingly, we only find three possible values for the scaling weight w in all these classes of examples, namely w = 1, 2, 3. It would be interesting to understand the physical meaning of such constrained set of possibilities which, as we have pointed out, imply that $\mathcal{T}^{1/2}$ is at or below the species scale. Given the link between the SDC and the existence of dualities, a tantalising possibility is that the value of w is characterising the possible dualities that can get manifest at the asymptotic limits, which would give us information about the nature of the dual tower.

It is also important to notice that not all BPS charges correspond to EFT strings. For this to happen, non-perturbative corrections breaking the shift symmetry must get suppressed along the RG flow, which imposes certain positivity constraints on the EFT string charges. In particular, a BPS string is an EFT string whenever all BPS instantons relevant for the asymptotic regime carry non-negative magnetic charge under the 2-form gauge field that couples to the string. This leads us to characterise the asymptotic region of an EFT in terms of a saxionic cone Δ hinting a possible semi-toric structure of the kind discussed in [62]. Our physical framework then naturally suggests a saxionic generalisation of the Morrison-Kawamata cone conjectures [62, 64, 65]. In physical terms, this saxionic cone conjecture basically states that the full set of EFT string charges $C_{\rm S}^{\rm EFT}$ (and hence Δ) is generated by a finite number of elementary duality-inequivalent charges. Furthermore, our definition of EFT strings shares many similarities with the definition of 5d supergravity strings in [40]. In particular, all our examples clearly indicate that also our EFT strings only exist in supergravity theories and do not survive any rigid/decompactification limit.

The DASC implies that any charge in $C_{\rm S}^{\rm EFT}$ should be physically realised by an EFT string, or by a superposition of them. In specific models this physically motivated *EFT* completeness translates into some non-trivial mathematical conjectures. For instance, in large volume heterotic compactifications we identified a sector of $C_{\rm S}^{\rm EFT}$ with NS5-branes wrapping nef divisors. The EFT completeness would then imply that on Calabi-Yau threefolds all nef divisors should admit an effective representative, as already conjectured by similar arguments in [40]. As another example, in large volume F-theory models on elliptically fibred Calabi-Yau four-folds a sector of $C_{\rm S}^{\rm EFT}$ is identified with D3-branes wrapping movable curves of the base X of the elliptic fibration. Hence, the EFT completeness similarly implies that in such bases X any movable curve should admit an effective representative. It would be interesting to better investigate these mathematical 'byproduct' conjectures, and extend them to other geometrical structures representing microscopically the possible EFT strings. Our perspective on the F-theory models has also highlighted a potential relevance of Mori Dream spaces [96] in this context, which deserves a more thorough study.

It would also be interesting to understand better the physics associated to non-EFT strings. In those cases, the string flow brings us to a finite boundary of the saxionic cone, where we cannot trust anymore the EFT description as some instanton corrections become unsupressed. In certain cases, these strings even become tensionless at these finite distance boundaries, at least at the classical level. Hence, they are similar to the tensionless strings that characterise the conformal field theories engineered at finite field distance points in higher dimensions. Furthermore, it would be important to better investigate the role of possible walls of BPS stability. As we have illustrated in some concrete examples, they may appear in the microscopic realisation of the 4d BPS instantons and strings and can crucially affect the identifications of the corresponding cones of charges.

Our results are valid even in the presence of a scalar potential, as long as the potential energy remains below the EFT cut-off along the infinite distance limit. Analogously, they also apply to setups with SUSY spontaneously broken below the EFT cut-off. Hence, it is justified to apply our bounds on the field range directly to phenomenological models, including large field inflation, since we also give a precise concrete lower bound for the exponential rate of the tower. The generalisation to higher dimensions is, however, unclear. In principle, the same story holds in higher dimensions if we can replace the EFT string by some BPS codimension-2 object. In general, however, BPS and codimension-2 are conditions that are not always compatible, and one of the two must be dropped. As a first goal, it would be interesting to formulate our conjecture in terms of (possibly non-BPS) codimension-2 objects whenever the moduli space contains codimension-1 singularities, as the presence of these objects can be argued from the absence of global symmetries. It would also be interesting to investigate the possible connection between the RG-flow interpretation of our EFT strings and the holographic perspective on the infinite distance limits proposed in [19, 128].

Our work emphasises the relevance of extended objects to understand the quantum gravity principles underlying the SDC, even if these extended objects do not give rise to the leading tower of states becoming light. This motivates to revisit the physics at asymptotic limits in higher dimensions to check whether some relation along the lines of (7.1) holds for arbitrary dimension. Our work also heavily uses the map between large field distances when moving in moduli space and large field distances induced by the presence of low codimension objects in an EFT. This latter perspective might hopefully shed some light into the search of a bottom-up rationale for the SDC, regardless of string theory.

Acknowledgments

We would like to thank Andreas Braun, Iñaki García-Etxebarria, Antonella Grassi, Thomas Grimm, Jim Halverson, Jacob McNamara, Ernesto C. Mistretta, Miguel Montero, Ángel Uranga, Cumrun Vafa, Timo Weigand and Max Wiesner for useful comments and discussions. SL is supported by the Dutch Research Council (NWO) via a Start-Up grant; during the initial stage of the project, SL was also supported by a fellowship of Angelo Della Riccia Foundation, Florence and a fellowship of Aldo Gini Foundation, Padova. FM is supported through the grants SEV-2016-0597 and PGC2018-095976-B-C21 from MCIU/AEI/FEDER, UE. LM is supported in part by the MIUR-PRIN contract 2017CC72MK_003. IV is supported by Grant 602883 from the Simons Foundation.

A Terminology

In this appendix we collect the basic definitions made and used throughout the main text. In A.1 we gather the definitions that characterise the string flows introduced in section 2, and whose casuistics was analysed in section 4.1. In A.2 we summarise the terminology regarding cones in algebraic Kähler geometry used in sections 4 and 6.

A.1 Glossary of strings and their flows

BPS string solution: holomorphic map $t^i : D \to \mathcal{M}$ from a disc $D \in \mathbb{C}$ to the EFT field space \mathcal{M} that saturates the bound (2.10), assuming a vanishing or negligible superpotential for the chiral fields $t^i = a^i + is^i$. The string charges e^i are encoded in the monodromy (2.11) around the string core.

Fundamental string: string solution with a singular core that cannot be resolved with a 4d quantum field theory approach, so the string corresponds to a fundamental localised object in the theory. Its tension satisfies (2.1) for any admissible choice of EFT cut-off Λ .

BPS axionic string: BPS string with a flow that, in the limit (2.14), displays a holomorphic profile of the form (2.12), and a metric warp factor (2.15) where the Kähler potential has a continuous shift symmetry $a^i \rightarrow a^i + e^i \times \text{const.}$

EFT string: BPS fundamental axionic string whose continuous shift symmetry is exact at the perturbative level, and all non-perturbative corrections are suppressed for any value in $t^i(D)$. In a given asymptotic EFT regime with a saxionic cone (4.5) the cone of EFT string charges $C_{\rm S}^{\rm EFT}$ can be characterised by (4.12), or equivalently by (4.17).

String RG flow: In an EFT with a cut-off Λ , the actual EFT string profile is a coarsegrained approximation of (2.12), involving Fourier modes up to Λ . Within a disc of radius $r_{\Lambda} \sim 1/\Lambda$ around the string core the profile is approximately constant, and all of its energy is described by the couplings of the localised string operator (2.22). Changing Λ induces a flow of such couplings, which can be measured by evaluating the solution (2.13a) at r_{Λ} . In this sense, the saxionic profile (2.13a) towards $r \to 0$ is understood as an EFT string RG flow towards the UV.

Effective string tension $\mathcal{T}(\Lambda)$: Coupling that appears in the EFT piece of the action (2.22) describing a string. This coupling is computed by evaluating (2.23) on the EFT string solution (2.13a) at the distance $r_{\Lambda} \sim 1/\Lambda$, where Λ is the EFT cut-off scale. As such, it varies along the string RG flow. The asymptotic behaviour of $\mathcal{T}_{\mathbf{e}}$ along formal limit $\Lambda \to \infty$ can be used to understand the behaviour of the probe string tension $\mathcal{T}_{\mathbf{e}}$ along the saxionic direction (3.1) selected by its charges. \mathcal{T} also depends on the flow parameters (s_0, r_0) in (2.13a) which may be related to a point s in saxionic field space via some IR regularisation mechanism.

Probe string tension T(s): Tension that is associated to 4d string but it is computed in a microscopic completion of the theory, e.g. by performing dimensional reduction in string compactifications. For EFT strings in axionic regimes it also takes the form (2.23), but it only depends on the expectation values for the saxions and it is independent of the EFT cut-off scale Λ . T signals a breakdown of the semiclassical description of the 4d EFT whenever $T \leq \Lambda^2$.

Elementary string flow: A BPS string flow is dubbed *elementary* if it is generated by a BPS string charge that cannot be decomposed into a positive linear combination of other BPS string charges. Otherwise we call it *non-elementary*.

Flow degeneracy order: An EFT string flow describes a one-dimensional path in the dual saxionic cone \mathcal{P} defined in (4.7), and it necessarily ends in one of its boundary faces $\mathcal{F}_{\mathbf{e}}$, see (4.15). We say that the string flow degeneracy is of order p if it ends on a codimension-p face $\mathcal{F}_{\mathbf{e}}$. A string flow is called *non-degenerate* if p = 1, otherwise the flow is called *degenerate*.

Singularity type: Given an Kähler potential of the form (3.8) and a saxionic EFT string flow (3.1), the corresponding singularity type n is identified by the asymptotic scaling $P(s) \sim \sigma^n$ along the flow — see also footnote 7.

A.2 Algebraic geometry

Here we summarise our notations for some relevant cones in algebraic Kähler geometry that we use throughout this paper. We mostly follow the notation of [92, 98]. The following cones are obtained as positive real spans of various types of curves and divisors, and possibly closures thereof. They are subsets of \mathbb{R} -spans $N^1(X)_{\mathbb{R}}$ and $N_1(X)_{\mathbb{R}}$ of the Néron-Severoni groups $N^1(X)_{\mathbb{Z}}$ (of divisors modulo numerical equivalence) and $N_1(X)_{\mathbb{Z}}$ (of curves modulo numerical equivalence) — see for instance [129]. The cones we consider are:

$$\operatorname{Eff}^{1}(X) \equiv \{ \text{cone generated by effective divisors} \}, \qquad (A.1a)$$

 $\operatorname{Eff}_1(X) \equiv \{ \text{cone generated by effective curves} \},$ (A.1b)

 $\operatorname{Nef}^{1}(X) \equiv \{ \text{cone generated by nef divisors} \},$ (A.1c)

- $Amp^{1}(X) \equiv \{ \text{cone generated by ample divisors} \}, \qquad (A.1d)$
- $Mov^1(X) \equiv \{ \text{cone generated by movable divisors} \},$ (A.1e)
- $Mov_1(X) \equiv \{ \text{cone generated by movable curves} \}.$ (A.1f)

We recall that an effective divisor is a positive linear combination holomorphic subvarieties of complex codimension one. Similarly, and effective curve is a positive linear combination of holomorphic subvarieties of complex dimension one. Furthermore the closure of $\text{Eff}^1(X)$, $\overline{\text{Eff}}^1(X)$, is known as cone of pseudo-effective \mathbb{R} -divisors, while $\overline{\text{Eff}}_1(X)$ is also know as Mori cone of pseudo-effective \mathbb{R} -curves, often denoted $\overline{\text{NE}}(X)$. The intersection defines a natural pairing between $N^1(X)_{\mathbb{R}}$ and $N_1(X)_{\mathbb{R}}$. By using this pairing, we can define $\text{Nef}^1(X)$ as the dual of the Mori cone or, equivalently, of $\text{Eff}_1(X)$:

$$\operatorname{Nef}^{1}(X) = \operatorname{Eff}_{1}(X)^{\vee}. \tag{A.2}$$

Notice that, by definition, $\operatorname{Nef}^1(X)$ is closed. By Kleiman criterion — see e.g. [129] — one can then identify $\operatorname{Amp}^1(X)$ (which is open) with the interior of $\operatorname{Nef}^1(X)$:

$$\operatorname{Amp}^{1}(X) = \operatorname{Int}[\operatorname{Nef}^{1}(X)].$$
(A.3)

and, viceversa, $\operatorname{Nef}^1(X)$ can be identified with the closure of the cone of ample divisors: $\operatorname{Nef}^1(X) = \overline{\operatorname{Amp}}^1(X)$. Notice also that $\operatorname{Amp}^1(X)$ is Poincaré dual to the Kähler cone:

$$\mathcal{K}(X) \equiv \{ \text{K\"ahler cone} \} \simeq \operatorname{Amp}^{1}(X)$$
 (via Poincaré duality). (A.4)

A detailed discussion on the cones of movable divisors and curves can be found in [92]. For our purposes, it is sufficient to loosely define movables divisors and curves as those that can be moved across the entire space X. Notice the sequence of inclusions

$$\operatorname{Amp}^{1}(X) \subset \operatorname{Nef}^{1}(X) \subset \operatorname{Mov}^{1}(X) \subset \overline{\operatorname{Eff}}^{1}(X), \qquad (A.5)$$

while in general nef and movable \mathbb{R} -divisors could be non-effective (but only pseudoeffective). Analogously, we have $\operatorname{Mov}_1(X) \subset \overline{\operatorname{Eff}}_1(X)$. Furthermore, by [91] we can identify $\operatorname{Mov}_1(X)$ with the cone dual to the cone of effective divisors:

$$Mov_1(X) = Eff_1(X)^{\vee}.$$
(A.6)

Finally, all the above cones are defined over the real numbers. The corresponding integral cones are then obtained by intersection with $N^1(X)_{\mathbb{Z}}$ and $N_1(X)_{\mathbb{Z}}$, for example

$$\operatorname{Eff}^{1}(X)_{\mathbb{Z}} \equiv \operatorname{Eff}^{1}(X) \cap N^{1}(X)_{\mathbb{Z}}, \quad \operatorname{Mov}_{1}(X)_{\mathbb{Z}} \equiv \operatorname{Mov}_{1}(X) \cap N_{1}(X)_{\mathbb{Z}}, \quad \dots \quad (A.7)$$

In most of our examples $N^1(X)_{\mathbb{Z}} = H^2(X, \mathbb{Z})_{t.f.}$ and $N_1(X)_{\mathbb{Z}} = H_2(X, \mathbb{Z})_{t.f.}$, where the subscript t.f. selects the torsion free part of the group.

B Gravitational contributions to the energy density

Equation (2.33) has been obtained by using the energy-momentum tensor of the localised string and of the backreacted (s)axionic fields. In this appendix we discuss the inclusion of the gravitational contribution to $\mathcal{E}(r)$, showing that it does not affect (2.33). We will follow and generalise [130], which focused on a simple dilatonic model equivalent to the single axionic case with Kähler potential $K = -\log \operatorname{Im} t$.

As in [130], we start from the total stress-energy pseudo-tensor Θ_{mn} , which includes also the contribution of the gravitational stress-energy pseudo-tensor, computed with respect to the Minkowski vacuum. This is defined as follows (see for instance [131]). One writes the metric as $g_{mn} = \eta_{mn} + h_{mn}$, where η_{mn} is the Minwkoski metric, and then splits the Einstein tensor into $G_{mn} = G_{mn}^{(1)} + \Delta G_{mn}$, where $G_{mn}^{(1)} = R_{mn}^{(1)} - \frac{1}{2}\eta_{mn}\eta^{pq}R_{pq}^{(1)}$ is the contribution linear in h_{mn} , with

$$R_{mn}^{(1)} = \frac{1}{2} \left(\partial_m \partial_p h^p{}_n + \partial_n \partial_p h^p{}_m - \partial^p \partial_p h_{mn} - \partial_m \partial_n h^p{}_p \right) . \tag{B.1}$$

Then, the Einstein equation $G_{mn} = M_{\rm P}^{-2}T_{mn}$ can be written in the form $G_{mn}^{(1)} = M_{\rm P}^{-2}\Theta_{mn}$, where $\Theta_{mn} = T_{mn} - M_{\rm P}^2\Delta G_{mn}$ is the total stress-energy pseudo-tensor. Notice that, onshell, one can use the Einstein equation to identify Θ_{mn} with $M_{\rm P}^2 G_{mn}^{(1)}$. For our metric ansatz (2.4), we then get the following non-vanishing contributions to Θ_{mn}

$$\Theta_{\mu\nu} = 2\eta_{\mu\nu}M_{\rm P}^2\partial_z\partial_{\bar{z}}e^{2D} \equiv \frac{1}{2}\eta_{\mu\nu}M_{\rm P}^2\frac{\partial^2 e^{2D}}{\partial y^\alpha \partial y^\alpha},\tag{B.2}$$

where we have split $x^m = (x^{\mu}, y^{\alpha})$, with $x^{\mu} = (t, x)$ and $y^1 + iy^2 = z$. This form of Θ_{mn} is valid for any solution of the Einstein equations of the form (2.4), and we assume it to be valid for an appropriate localised regularisation of our string solution (2.12). The overall normalisation adopted in (2.15) ensures that the metric reduces to the canonical Minkowski one at r_0 . We can then obtain the total linear energy density by integrating Θ_{tt} over the disk of radius r_0 . The result can be written as a boundary term

$$\mathcal{E}_{\text{tot}} = \int_{D(r_0)} d^2 y \,\Theta_{tt} = -\frac{1}{2} M_{\text{P}}^2 \int_{D(r_0)} d^2 y \frac{\partial^2 e^{2D}}{\partial y^{\alpha} \partial y^{\alpha}} = -\frac{1}{2} M_{\text{P}}^2 \oint_{r=r_0} d\theta \frac{\partial e^{2D}}{\partial \log r} = -\pi M_{\text{P}}^2 \frac{\partial e^{2D}}{\partial \log r} \Big|_{r=r_0},$$
(B.3)

which is then independent of the localised regularisation of the solution. In the last step of (B.3), we have assumed that the Kähler potential is invariant under axionic shifts and the string flow solution is axially symmetric, as in section 2.2. From (2.15) and (2.13a) we then obtain

$$\frac{\partial e^{2D}}{\partial \log r}\Big|_{r=r_0} = -\frac{\partial K(s(r))}{\partial \log r}\Big|_{r=r_0} = -\left[\frac{\partial K}{\partial s^i}\frac{\partial s^i}{\partial \log r}\right]\Big|_{r=r_0} = \frac{1}{\pi}e^i\ell_i(r_0) \ . \tag{B.4}$$

Hence, we arrive at

$$\mathcal{E}_{\text{tot}}(r_0) = M_{\text{P}}^2 e^i \ell_i(r_0) \,, \tag{B.5}$$

which is the same as (2.33) at $r = r_0$. Since r_0 is arbitrary, we conclude that (2.33) is valid also if we take into account gravitational effects, as anticipated. In other words, the gravitational field does not contribute to the effective tension of the string. This conclusion is confirmed by the perturbative field theory calculation of appendix D.

Equation (B.5) generalises what was obtained in [130] for the simple dilatonic model. As in [130], this result may be interpreted as a 'non-renormalisation' of the string tension, in the following sense: (B.5) means that the string backreaction inside the disk of radius r_0 , for fixed boundary conditions $\ell_i(r_0) = \ell_i^0$, does not 'renormalise' the 'bare' string tension $M_{\rm P}^2 e^i \ell_i^0$ which one would get from a probe string on the unperturbed vacuum satisfying the same boundary conditions. On the other hand, in the EFT viewpoint adopted in the present paper and [32], we may identify r_0 with a floating cut-off length scale r_{Λ} and obtain a scale dependent effective string tension.

C Geodesic paths in field space and string flows

In section 2 we computed the scalar flows induced by fundamental strings. The paths that the chiral fields draw in field space are given by (2.12). In this section we address the question of whether those paths are geodesic in field space.

As a preliminary remark, it is worth noticing that the solution (2.12) is independent of the field space metric. Thus, generically, the trajectories are *not* geodesic. In order to see this explicitly, let us focus on the radial saxionic trajectories described by (2.13a) and keep the axions a^i fixed. The geodesic equations that can be obtain from the action (2.3), assuming that the Kähler potential depends solely on the saxions s^i , read

$$\frac{\mathbf{D}^2 s^i}{\mathbf{d\lambda}^2} \equiv \frac{\mathbf{d}^2 s^i}{\mathbf{d\lambda}^2} + \Gamma^i{}_{jk} \frac{\mathbf{d}s^j}{\mathbf{d\lambda}} \frac{\mathbf{d}s^k}{\mathbf{d\lambda}} = 0, \qquad (C.1)$$

where λ is an affine parameter and Γ^{i}_{jk} are the standard Christoffel symbols associated with the saxionic metric \mathcal{G}_{ij} defined (2.18). Identifying λ with the proper field distance (3.2) and employing the explicit saxionic flow (2.13a), we obtain that the geodesic equation (C.1) is generically violated in following way

$$\frac{\mathrm{D}^2 s^i}{\mathrm{d}\lambda^2} = f^i \qquad \text{with} \quad f^i \equiv \frac{M_\mathrm{P}^2}{2\mathcal{Q}_\mathbf{e}^2} \left(\frac{\mathcal{G}^{ij}}{M_\mathrm{P}^2} - \frac{e^i e^j}{\mathcal{Q}_\mathbf{e}^2}\right) \partial_j \mathcal{Q}_\mathbf{e}^2. \tag{C.2}$$

Hence the deviation of the radial flow from a geodesic is encoded in the 'force' f^i , which satisfies the orthogonality condition $\mathcal{G}_{ij}e^i f^j = 0$. Its norm is given by

$$\|\boldsymbol{f}\|^{2} \equiv \mathcal{G}_{ij}f^{i}f^{j} = \frac{M_{\mathrm{P}}^{2}}{4\mathcal{Q}_{\mathbf{e}}^{4}} \left(\frac{\mathcal{G}^{ij}}{M_{\mathrm{P}}^{2}} - \frac{e^{i}e^{j}}{\mathcal{Q}_{\mathbf{e}}^{2}}\right) \partial_{i}\mathcal{Q}_{\mathbf{e}}^{2}\partial_{j}\mathcal{Q}_{\mathbf{e}}^{2}.$$
 (C.3)

Thus, the radial flow is geodesic if and only if

$$\|\boldsymbol{f}\| = 0. \tag{C.4}$$

For moduli spaces described by a single chiral field, only one saxion can flow. The physical charge of the string inducing such a flow is $Q_{\mathbf{e}}^2 = M_{\mathbf{P}}^2 \mathcal{G}_{ss} e^2$. Thus, (C.4) is trivially satified. For higher dimensional moduli spaces, separable Kähler potentials $K = \sum_i K^{(i)}(s^i)$ still deliver geodesic flows if they are elementary, namely only one string charge among e^i is non-vanishing, while more general flows may not be geodesic.

On the other hand we would now like to argue that if the Kähler potential on the saxionic cone Δ takes the form (3.8) then, under some additional reasonable assumption, the EFT flows are *asymptotically geodesic*. Let us first consider an EFT charge $\mathbf{e} \in \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$ which sits within Δ (and not on its boundary). By writing the flow in the form (4.9) and using the homogeneity of $\mathcal{G}_{ij}(\mathbf{s}) \equiv \frac{1}{2} \partial_i \partial_j K(\mathbf{s})$ (with $\partial_i \equiv \frac{\partial}{\partial s^i}$) and $\mathcal{Q}_{\mathbf{e}}^2(\mathbf{s}) \equiv M_{\mathrm{P}}^2 \mathcal{G}_{ij}(\mathbf{s}) e^i e^j$, in the asymptotic limit $\sigma \to \infty$ we have

$$\mathcal{Q}_{\mathbf{e}}^{2}(\boldsymbol{s}) = \mathcal{Q}_{\mathbf{e}}^{2}(\boldsymbol{s}_{0} + \sigma \mathbf{e}) = \frac{1}{\sigma^{2}} \left[\mathcal{Q}_{\mathbf{e}}^{2}(\mathbf{e}) + M_{\mathrm{P}}^{2}\mathcal{O}(\sigma^{-1}) \right], \qquad (C.5a)$$
$$(\partial_{i}\mathcal{Q}_{\mathbf{e}}^{2})(\boldsymbol{s}) = \frac{1}{2}M_{\mathrm{P}}^{2}e^{j}e^{k}(\partial_{i}\partial_{j}\partial_{k}K)(\boldsymbol{s}_{0} + \sigma \mathbf{e}) = \frac{1}{2\sigma^{3}}M_{\mathrm{P}}^{2} \left[e^{j}e^{k}(\partial_{i}\partial_{j}\partial_{k}K)(\mathbf{e}) + \mathcal{O}(\sigma^{-1}) \right]$$
$$= -\frac{2}{\sigma^{3}}M_{\mathrm{P}}^{2} \left[e^{j}\mathcal{G}_{ij}(\mathbf{e}) + M_{\mathrm{P}}^{2}\mathcal{O}(\sigma^{-1}) \right]. \qquad (C.5b)$$

Notice that we can consider $\mathcal{G}_{ij}(\mathbf{e})$ non-degenerate since $\mathbf{e} \in \Delta$ and we are assuming that $K = -\log P$ is well defined over the entire Δ . Hence, also $\mathcal{Q}_{\mathbf{e}}^2(\mathbf{e})$ is well defined. Furthermore, $\mathcal{Q}_{\mathbf{e}}^2(\mathbf{e}) = \frac{n}{2}$, where n > 0 is the singularity type which in this case it is simply the homogeneity degree of P. We can then similarly obtain

$$(e^{i}\partial_{i}\mathcal{Q}_{\mathbf{e}}^{2})(\boldsymbol{s}) = -\frac{2}{\sigma^{3}} \left[\mathcal{Q}_{\mathbf{e}}^{2}(\mathbf{e}) + \mathcal{O}(\sigma^{-1})\right],$$

$$\left(\mathcal{G}^{ij}\partial_{i}\mathcal{Q}_{\mathbf{e}}^{2}\partial_{j}\mathcal{Q}_{\mathbf{e}}^{2}\right)(\boldsymbol{s}) = \frac{4M_{\mathrm{P}}^{2}}{\sigma^{4}} \left[\mathcal{Q}_{\mathbf{e}}^{2}(\mathbf{e}) + \mathcal{O}(\sigma^{-1})\right],$$

(C.6)

Together with (C.5a), these imply that

$$\lim_{\sigma \to \infty} \|\boldsymbol{f}\|^2 (\boldsymbol{s}_0 + \sigma \mathbf{e}) = 0.$$
 (C.7)

This argument does not immediately work if \mathbf{e} sits on the boundary of Δ , since $\mathcal{G}_{ij}(\mathbf{e})$ may not be well defined. However, in this case it is reasonable to assume that one can go to some saxionic base in which part of the saxions do not participate at all in the flow. Correspondingly, one can restrict Δ to a lower-dimensional cone $\hat{\Delta}$, which contains the corresponding restriction $\hat{\mathbf{e}}$ of the original charge. The above reasoning can then be repeated, reaching the same conclusion. In particular, if one can restrict to a one-dimensional $\hat{\Delta}$, then the radial flow is exactly geodesic.

Let us check the validity of this argument in some concrete example. Consider first the separable Kähler potential

$$K = -\sum_{i} n_i \log s^i \,. \tag{C.8}$$

It is not difficult to verify that elementary flows are exactly geodesic and any non-elementary flow is asymptotically geodesic.

As another example, consider the Kähler potential (4.28) that was examined in section 4.2. For the choice of charges $\mathbf{e} = (e^0, e^1, e^2) = (1, 0, 0)$, it can be easily checked that the flow is exactly geodesic. Instead, choosing $\mathbf{e} = (0, 1, 0)$ we obtain

$$\|\boldsymbol{f}\|^{2} = M_{\rm P}^{4} \frac{9(s^{1})^{2}(s^{2})^{3}[3(s^{1})^{2} + 3s^{1}s^{2} + (s^{2})^{2}]^{2}}{[9(s^{1})^{3} + 9(s^{1})^{2}s^{2} + 3s^{1}(s^{2})^{2} + (s^{2})^{3}]^{3}},$$
(C.9)

Hence, the associated string flow is not generically geodesic. However, asymptotically we have $s^1 \sim \sigma \rightarrow \infty$ while s^0, s^2 are fixed, and then

$$\|\boldsymbol{f}\|^2 \sim \frac{M_{\rm P}^4}{\sigma^3} \to 0.$$
 (C.10)

Thus, such an elementary flow can be regarded as asymptotically geodesic. Analogously, for an elementary charge $\mathbf{e} = (0, 0, 1)$ we get

$$\|\boldsymbol{f}\|^{2} = M_{\rm P}^{4} \frac{9(s^{1})^{4} [3(s^{1})^{2} + 3s^{1}s^{2} + (s^{2})^{2}]^{2}}{s^{2}(3s^{1} + s^{2})[9(s^{1})^{3} + 9(s^{1})^{2}s^{2} + 3s^{1}(s^{2})^{2} + (s^{2})^{3}]^{3}}.$$
 (C.11)

Asymptotically along the corresponding flow we have $s^2 \sim \sigma \rightarrow \infty$, while s^0, s^1 are fixed, so that

$$\|\boldsymbol{f}\|^2 \sim \frac{M_{\rm P}^4}{\sigma^4} \to 0.$$
 (C.12)

Therefore, also this elementary flow is asymptotically geodesic.

Finally, one can explicitly check that (C.7) holds for generic non-elementary charges $(e^0, e^1, e^2) \in \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$, with $e^i \in \mathbb{Z}_{>0}$.

D String RG flows in Field Theory

On field theoretical grounds, fundamental objects are localised operators that couple to the bulk fields via their tensions and charges. As such, their couplings to the bulk fields are generically subjected to renormalisation and, subsequently, to an RG flow. At the same time, the inclusion of extended objects within effective descriptions modifies the background geometry, reflecting on nontrivial solutions for the bulk fields. As an example, the BPS solution (2.12) may be considered as the effect of the background upon the dual saxions ℓ_i induced by a single fundamental string.

The link between these two seemingly unrelated descriptions were threaded in [38], where it was shown that the RG flow induced by extended objects coincides with the effect of their backreaction. In the following we will delve into this identification restricting to the case of supersymmetric strings introduced in section 2. As we will shortly see, the tension of BPS strings obeys a *non-renormalisation condition*, generalising [52, 130]. From this viewpoint, we will show how the BPS solution (2.12) may be regarded as the RG flow of the dual saxions.

In $\mathcal{N} = 1$ locally supersymmetric theories, the coupling of a BPS-fundamental string [42, 43] to the bulk supergravity is easily written in terms of linear multiplets. The components of each linear multiplet L_i comprise a real scalar ℓ_i , dubbed *dual saxion* in section 2, a real gauge two-form \mathcal{B}_{2i} and matter Weyl fermions ψ_i^{α} . The full action describing the coupling of a fundamental string to a bulk supergravity whose matter content is solely encoded within linear multiplets is

$$S = S_{\text{bulk}} + S_{\text{string}}, \qquad (D.1)$$

with the bulk part

$$S_{\text{bulk}} = M_{\text{P}}^2 \int \frac{1}{2} R * 1 - \frac{1}{2} \int \mathcal{G}^{ij}(\ell) \left(M_{\text{P}}^2 \,\mathrm{d}\ell_i \wedge * \mathrm{d}\ell_j + \frac{1}{M_{\text{P}}^2} \mathcal{H}_{3\,i} \wedge * \mathcal{H}_{3\,j} \right) \,, \tag{D.2}$$

and the localised string contribution

$$S_{\text{string}} = -\int_{\mathcal{S}} \sqrt{-h} \,\mathcal{T}_{\text{str}} + e^i \int_{\mathcal{S}} \mathcal{B}_{2\,i} \,, \qquad \text{with} \quad \mathcal{T}_{\text{str}} = M_{\text{P}}^2 \,\int_{\mathcal{S}} |e^i \ell_i| \,. \tag{D.3}$$

From a field theoretical perspective, the string tension appearing in (D.3) plays the role of a coupling parameter and, as such, is subjected to renormalisation. In order to study its renormalisation, we employ the background field method and expand the bulk fields $g_{\mu\nu}$, ℓ_i and \mathcal{B}_{2i} around some *fixed* background values $g^0_{\mu\nu}$, ℓ_i^0 and \mathcal{B}_{2i}^0 :

$$g_{\mu\nu}(u,v) = g^{0}_{\mu\nu} + h_{\mu\nu}(u,v), \qquad \ell_{i}(u,v) = \ell^{0}_{i} + \hat{\ell}_{i}(u,v), \qquad \mathcal{B}_{2i}(u,v) = \mathcal{B}^{0}_{2i} + \hat{\mathcal{B}}_{2i}(u,v).$$
(D.4)

Here we have denoted with $h_{\mu\nu}$, $\hat{\ell}_i$ and $\hat{\mathcal{B}}_{2i}$ the variations of the fields around their background values and, employing the SO(1,1) × SO(2) symmetries of the system, these are assumed to solely depends on the coordinates $\mathbf{u} = (u, v)$, which parametrise the directions transverse to the string.²¹ Close enough to the string, while still at distances $r > \Lambda^{-1}$, its backreaction on the field metric is negligible and the background metric may be just considered Minkowski: $g^0_{\mu\nu} = \eta_{\mu\nu}$. We will further assume the string to be static, stretching along the (t, x)-spacetime directions at fixed (u, v)-position and that the only nontrivial component of \mathcal{B}_{2i} are $\mathcal{B}_{01i} \equiv \mathcal{B}_i$, with the background expansion $\mathcal{B}_i(u, v) = \mathcal{B}_i^0 + \hat{\mathcal{B}}_{2i}(u, v)$.

Let us introduce the *bare* string action

$$S_{\text{string}}^{(0)} = -\int_{\mathcal{S}} \mathrm{d}t \,\mathrm{d}x \,\mathcal{T}_{\text{bare}} + e^i \int_{\mathcal{S}} \mathcal{B}_{2i}^0 \,, \tag{D.5}$$

which is (D.3) evaluated on the background and with the bare tension

$$\mathcal{T}_{\text{bare}} = M_{\text{P}} e^i \ell_i^0 \,. \tag{D.6}$$

Owing to the coupling of the string to the background perturbations $h_{\mu\nu}$, $\hat{\ell}_i$ and $\hat{\mathcal{B}}_{2i}$, interactions of the string as in figure 6 are possible. These self-interactions are expected to renormalise the string tension. In order to explicitly compute the renormalisation induced by the Feynman diagrams in figure 6, we proceed as in [52].

We regard the strings as *external* sources for the bulk fields. Namely, we rewrite the contribution (D.3) of the string to the bulk action by expanding it linearly in field perturbations

$$S_{\text{string}} = S_{\text{string}}^{(0)} - \int d^4x \left(h^{\mu\nu} J_{\mu\nu} + \hat{\ell}_i J_\ell^i + \hat{\mathcal{B}}_i J_\mathbf{e}^i \right) + \dots$$
(D.7)

²¹The coordinates (u, v) are the real counterpart of the complex coordinates (z, \bar{z}) introduced in (2.4).

where we introduced the 'currents'

$$J_{\mu\nu} = -\frac{\delta}{\delta h^{\mu\nu}} S_{\text{string}} = \frac{|e^{j}\ell_{j}^{0}|}{2} \begin{pmatrix} 1 & 0 & 0 & 0\\ 0 & -1 & 0 & 0\\ 0 & 0 & 0 & 0\\ 0 & 0 & 0 & 0 \end{pmatrix} \delta(\mathcal{S}) , \qquad (D.8a)$$

$$J_{\ell}^{i} = -\frac{\delta}{\delta \hat{\ell}_{i}} S_{\text{string}} = M_{\text{P}}^{2} e^{i} \delta\left(\mathcal{S}\right) \,, \tag{D.8b}$$

$$J_{\mathbf{e}}^{i} = \frac{\delta}{\delta \hat{\mathcal{B}}_{i}} S_{\text{string}} = e^{i} \delta\left(\mathcal{S}\right) \,, \tag{D.8c}$$

which play the role of classical sources for the graviton, dual saxions and two-forms.

It is convenient to momentarily collect the field variations as $\psi^{\mathcal{A}} \equiv (h^{\mu\nu}, \hat{\ell}_i, \hat{\mathcal{B}}_i^{\mu\nu})$ and the sources as $J_{\mathcal{A}} \equiv (J^{\mu\nu}, J_{\ell}^i, J_{\mathbf{e}}^i)$. Then, the bulk action (D.2) is recast as

$$S = S_{\text{string}}^{(0)} + \int dt \, dx \, d^2 \mathbf{u} \int d^2 \mathbf{u}' \left(\frac{1}{2} \psi^{\mathcal{A}}(\mathbf{u}) Q_{\mathcal{A},\mathcal{B}}(\mathbf{u};\mathbf{u}') \psi^{\mathcal{B}}(\mathbf{u}') - J_{\mathcal{A}}(\mathbf{u}) \psi^{\mathcal{A}}(\mathbf{u}) \right) \,. \tag{D.9}$$

Integrating out the $\psi^{\mathcal{A}}$, we arrive at

$$S = S_{\text{string}}^{(0)} + \frac{\mathrm{i}}{2} \int \mathrm{d}t \,\mathrm{d}x \,\mathrm{d}^2 \mathbf{u} \int \mathrm{d}^2 \mathbf{u}' J_{\mathcal{A}}(\mathbf{u}) \Delta^{\mathcal{A},\mathcal{B}}(\mathbf{u};\mathbf{u}') J_{\mathcal{B}}(\mathbf{u}') \,, \tag{D.10}$$

with $\Delta^{\mathcal{A},\mathcal{B}} = i(Q_{\mathcal{A},\mathcal{B}})^{-1}$ the field propagators in two-dimensional Euclidean space. Explicitly, the propagators for the graviton $h_{\mu\nu}$, dual saxions ℓ_i and two-forms \mathcal{B}_i propagators are respectively given by

$$\Delta^{\mu\nu,\rho\sigma}(\mathbf{u};\mathbf{u}') = (\eta^{\mu\rho}\eta^{\nu\sigma} + \eta^{\mu\sigma}\eta^{\nu\rho} - \eta^{\mu\nu}\eta^{\rho\sigma})\Delta(\mathbf{u};\mathbf{u}'), \qquad (D.11a)$$

$$\Delta_{ij}^{\ell}(\mathbf{u};\mathbf{u}') = \mathcal{G}_{ij}(\ell^0)\Delta(\mathbf{u};\mathbf{u}'), \qquad (D.11b)$$

$$\Delta_{ij}^{\mathcal{B}}(\mathbf{u};\mathbf{u}') = -M_{\rm P}^4 \mathcal{G}_{ij}(\ell^0) \Delta(\mathbf{u};\mathbf{u}'), \qquad (\text{D.11c})$$

where $\Delta(\mathbf{u}; \mathbf{u}')$ is the two-dimensional scalar propagator

$$\Delta(\mathbf{u};\mathbf{u}') = \frac{\mathrm{i}}{4\pi} \log|\mathbf{u} - \mathbf{u}'| \equiv \frac{\mathrm{i}}{4\pi} \log r \,. \tag{D.12}$$

The second term that appears in (D.10) can be rewritten as

$$\frac{\mathrm{i}}{2} \int_{\mathcal{S}} \mathrm{d}t \, \mathrm{d}x \, J_{\mathcal{A}}(\mathbf{u}) \Delta^{\mathcal{A},\mathcal{B}}(\mathbf{u};\mathbf{u}) J_{\mathcal{B}}(\mathbf{u}) \,, \tag{D.13}$$

and is responsible for the renormalisation of the string tension. However, as is clear from the behaviour of the two-dimensional Euclidean propagator (D.12), the contribution (D.13) entails a propagator evaluated at zero distance. Thus, (D.13) is divergent both in the infrared and in the ultraviolet and has to be appropriately regularised. The simplest regularisation is achieved by converting the propagator (D.12) to momentum space and using a sharp momentum cutoff as follows:

$$\Delta^{\text{reg}}(\mathbf{u};\mathbf{u}) = \int_{\mu \le |p| \le \Lambda} \frac{\mathrm{d}^2 p}{(2\pi)^2} \tilde{\Delta}(p) = -\frac{\mathrm{i}}{M_{\mathrm{P}}^2} \int_{\mu \le |p| \le \Lambda} \frac{\mathrm{d}^2 p}{(2\pi)^2 p^2} = -\frac{\mathrm{i}}{2\pi M_{\mathrm{P}}^2} \log \frac{\Lambda}{\mu} \,. \tag{D.14}$$



Figure 6. The self interactions of a string due to the graviton $g_{\mu\nu}$, dual saxion ℓ and gauge two-form \mathcal{B}_2 .

where μ and Λ denote, respectively, an IR energy scale and the EFT cutoff. Therefore, (D.13) would contribute to the string tension with

$$\frac{\mathrm{i}}{2} \int_{\mathcal{S}} \mathrm{d}t \,\mathrm{d}x \, J_{\mathcal{A}}(\mathbf{u}) \Delta^{\mathcal{A},\mathcal{B}}(\mathbf{u};\mathbf{u}) J_{\mathcal{B}}(\mathbf{u}) = (C_{\mathrm{g}} + C_{\ell} + C_{\mathcal{B}}) \log \frac{\Lambda}{\mu} \,, \tag{D.15}$$

where we have singled out the contribution from gravity $C_{\rm g}$, the dual saxions C_{ℓ} , and the gauge two-forms $C_{\mathcal{B}}$. Therefore, the renormalised string tension is the bare tension to which one needs to add appropriate counterterms to cancel the divergences originating from (D.15);

$$\mathcal{T}_{\rm ren} = \mathcal{T}_{\rm bare} + (C_{\rm g} + C_{\ell} + C_{\mathcal{B}}) \log \frac{\Lambda}{\mu} \,. \tag{D.16}$$

However, the explicit computations of the contributions in (D.15) reveals that

$$C_{\ell} = -C_{\mathcal{B}} = \frac{1}{4\pi} M_{\rm P}^2 \mathcal{G}_{ij}(\ell_0) e^i e^j = \mathcal{Q}_{\mathbf{e}}^2(\ell_0) \,, \qquad C_{\rm g} = 0 \,. \tag{D.17}$$

As a result, the string tension is *de facto not renormalised*; namely, given a background specified by the value ℓ_i^0 , the renormalised string tension coincides with its bare value, computed upon the same background:

$$\mathcal{T}_{\rm ren}(\ell^0) = \mathcal{T}_{\rm bare}(\ell^0) \,. \tag{D.18}$$

The (non-)renormalisation of the tension can be alternatively understood as the cumulative effect of the backreaction due to bulk fields, which *dresses* the bare tension with additional contributions. Loosely speaking, the tension of a string evaluated at a distance $r_* > \delta$ receives contributions from the energy stored within the annulus $\delta < r < r_*$ compared to the string tension evaluated at the cutoff length δ .²² However, extended sources such as strings do contribute to the stress-energy tensor also with localised contributions.

²²Compare this case to the renormalisation of the electron mass. In presence of an electric field, the electron mass evaluated at the length scale δ_2 gets renormalised, with respect to the mass at $\delta_1 < \delta_2$, as $m(\delta_2) - m(\delta_1) = \mathcal{E}_{el}$, where \mathcal{E}_{el} is the energy of the electric field stored within δ_1 and δ_2 [132]. However, this argument is not general and holds when the energy comes solely from gauge fields — see [52, 133] for a more detailed discussion.

As clarified in [52, 133], these also ought to be taken into account in the renormalisation of the bare tension. Therefore, the renormalised tension is generically written as

$$\mathcal{T}_{\text{ren}} = \mathcal{T}_{\text{bare}} + \mathcal{E}_{\text{bulk}}(\delta) + \mathcal{E}_{\text{loc}}(\delta) \,. \tag{D.19}$$

The bulk contribution $\mathcal{E}_{\text{bulk}}(\delta)$ takes into account the energy of the bulk fields stored between δ and $r_* > \delta$. Singling out the different contributions, we write

$$\mathcal{E}_{\text{bulk}}(\delta) = \mathcal{E}_{\text{bulk}}^{\text{g}}(\delta) + \mathcal{E}_{\text{bulk}}^{\ell}(\delta) + \mathcal{E}_{\text{bulk}}^{\mathcal{B}}(\delta) \,. \tag{D.20}$$

From (2.29) we recall that

$$\mathcal{E}^{\ell}_{\text{bulk}}(\delta) + \mathcal{E}^{\mathcal{B}}_{\text{bulk}}(\delta) = M_{\text{P}}^2 \int_{\delta}^{r_*} \mathcal{J}_{\mathcal{M}} = e^i M_{\text{P}}^2 [\ell_i(r_*) - \ell_i(\delta)]$$
(D.21)

and, employing the BPS-condition (2.19), we further recognise that

$$\mathcal{E}_{\text{bulk}}^{\ell}(\delta) = \mathcal{E}_{\text{bulk}}^{\mathcal{B}}(\delta) = \frac{1}{2} e^{i} M_{\text{P}}^{2} [\ell_{i}(r_{*}) - \ell_{i}(\delta)].$$
(D.22)

The localised contributions to the renormalisation come from gravity and the dual saxion:

$$\mathcal{E}_{\rm loc}(\delta) = \mathcal{E}_{\rm loc}^{\rm g}(\delta) + \mathcal{E}_{\rm loc}^{\ell}(\delta) \,. \tag{D.23}$$

As noticed in [52], these are related to the bulk contributions appearing in (D.21) as follows:

$$\mathcal{E}_{\rm loc}^{\rm g}(\delta) = -\mathcal{E}_{\rm bulk}^{\rm g}(\delta), \qquad \mathcal{E}_{\rm loc}^{\ell}(\delta) = -2\mathcal{E}_{\rm bulk}^{\ell}(\delta) = -eM_{\rm P}^2[\ell(r_*) - \ell(\delta)]. \tag{D.24}$$

Clearly, inserting (D.20) and (D.23) in (D.19) the non-renormalisation for the tension is recovered.

The non-renormalisation of the string tension has the following implication. If we wish to evaluate the string tension at a reference radius r, then its value is simply given by the bare tension (D.6), as it appears in the action (2.22), evaluated at r:

$$\mathcal{T}(\ell(r)) = M_{\rm P}^2 e^i \ell_i(r) \,, \tag{D.25}$$

without further corrections. In the language introduced above, $\ell_i(r)$ have to take the background values ℓ_i^0 at the distance r from the string. From the solitonic solution (2.13a) we immediately get the 'flow' of the string tension

$$\mathcal{T}(\ell(r)) = \frac{M_{\rm P}^2 e^i \ell_i(r_0)}{1 - \frac{e^i \ell_i(r_0)}{\pi} \log \frac{r}{r_0}},\tag{D.26}$$

where r_0 is a fixed distance from the string.

E EFT instantons

For each instanton charge $\mathbf{m} \in C_{\mathrm{I}}$ appearing in the possible non-perturbatively generated chiral operators (4.3), one can write down BPS (singular) solutions, generalising the D(-1) solution discussed in [134]. These can be described using the Wick-rotated Euclidean version of the bulk terms in (2.21)

$$\frac{1}{2} \int \mathcal{G}^{ij} \left(M_{\rm P}^2 \,\mathrm{d}\ell_i \wedge *\mathrm{d}\ell_j + \frac{1}{M_{\rm P}^2} \mathcal{H}_{3i} \wedge *\mathcal{H}_{3j} \right) \,. \tag{E.1}$$

In Lorentzian signature, the dualisation of the 2-forms \mathcal{B}_{2i} to the dual axions a^i is obtained by relaxing the closure condition $d\mathcal{H}_{3i} = 0$ and introducing the Lagrange multiplier $\int da^i \wedge \mathcal{H}_{3i}$. \mathcal{H}_{3i} . In the Wick-rotated action we then have to add instead $-i \int da^i \wedge \mathcal{H}_{3i}$.

Suppose we want to add to this action a source term $2\pi i m_i t^i(x_0)$, corresponding to a point-like instanton located at the point x_0^{μ} . The relevant terms in the action are then given by

$$\frac{1}{2} \int \mathcal{G}^{ij} \left(M_{\mathrm{P}}^2 \,\mathrm{d}\ell_i \wedge * \mathrm{d}\ell_j + \frac{1}{M_{\mathrm{P}}^2} \mathcal{H}_{3\,i} \wedge * \mathcal{H}_{3\,j} \right) - \mathrm{i} \int \mathrm{d}a^i \wedge \mathcal{H}_{3\,i} - 2\pi \mathrm{i}\, m_i t^i(x_0) \,, \qquad (\mathrm{E.2})$$

so by integrating out a^i one gets the modified Bianchi identity

$$d\mathcal{H}_{3i} = 2\pi m_i \delta_4(x_0), \qquad (E.3)$$

which implies that $\int_{S^3} \mathcal{H}_{3i} = 2\pi m_i$ on any three-sphere S^3 surrounding x_0 . By keeping the boundary term, the action (E.2) reduces to

$$\frac{1}{2} \int \mathcal{G}^{ij} \left(M_{\mathrm{P}}^2 \,\mathrm{d}\ell_i \wedge * \mathrm{d}\ell_j + \frac{1}{M_{\mathrm{P}}^2} \mathcal{H}_{3\,i} \wedge * \mathcal{H}_{3\,j} \right) + 2\pi \, m_i s^i(x_0) - 2\pi \mathrm{i} m_i a_{\infty}^i \,, \qquad (\mathrm{E.4})$$

where a_{∞}^{i} are the expectation values of the axions at infinity. By some simple manipulations, using (E.3) and recalling that $ds^{i} = -\mathcal{G}^{ij}d\ell_{i}$ and $s^{i} = \frac{1}{2}\frac{\partial F}{\partial \ell_{i}}$, with $F = K + 2\ell_{i}s^{i}$ (following the conventions of [42]), we can write the relevant terms in the effective action in BPS form

$$S_{\rm E} = -\frac{1}{2M_{\rm P}^2} \int \mathcal{G}^{ij} \left(\mathcal{H}_{3\,i} + M_{\rm P}^2 \, \ast \, \mathrm{d}\ell_i \right) \wedge \ast \left(\mathcal{H}_{3\,j} + M_{\rm P}^2 \, \ast \, \mathrm{d}\ell_j \right) - 2\pi \mathrm{i} m_i t_{\infty}^i \,. \tag{E.5}$$

It is then clear that configurations satisfying the BPS condition

$$* \mathrm{d}\ell_i = -\frac{1}{M_\mathrm{P}^2} \mathcal{H}_{3\,i}\,,\tag{E.6}$$

extremise the action and do not provide any contribution to the energy-momentum tensor. Hence, (E.6) on a Euclidean flat metric $g_{\mu\nu} = \delta_{\mu\nu}$ gives a complete solution of the gravitational equations of motion. Combining (E.6) and (E.3) one gets

$$d * d\ell_i = -\frac{2\pi}{M_P^2} m_i \delta_4(x_0) , \qquad (E.7)$$

which given the boundary conditions has unique solution:

$$\ell_i(x) = \ell_i^{\infty} + \frac{m_i}{2\pi M_{\rm P}^2 |x - x_0|^2}, \qquad (E.8)$$

with $|x|^2 = \delta_{\mu\nu} x^{\mu} x^{\nu}$. The corresponding profile of \mathcal{H}_{3i} is fixed by (E.6). Furthermore, on these BPS solutions (E.5) reduces to $S_{\rm E}|_{\rm BPS} = -2\pi i \langle \mathbf{m}, \mathbf{t}_{\infty} \rangle$. Hence these instantons will generate non-perturbative effects proportional to $e^{-S_{\rm E}|_{\rm BPS}} = e^{2\pi i \langle \mathbf{m}, \mathbf{t} \rangle}|_{\infty}$, as expected.

Remember that in our picture the instanton charges $\mathbf{m} \in C_I$ generate the dual cone Δ^{\vee} and that the positivity of the EFT string tensions over the entire saxionic cone Δ is equivalent to $\overline{\mathcal{P}} \subset \Delta^{\vee}$, where \mathcal{P} is defined in (4.7). We will assume that \mathcal{P} is a cone, as it is automatically the case if we consider Kähler potentials of the form (3.8). Furthermore, we will assume that this cone is convex.²³ Now, if $\overline{\mathcal{P}}$ is a *strict* subcone of Δ^{\vee} , then we can distinguish two possible classes of flows: either $\mathbf{m} \in \overline{\mathcal{P}}$ or $\mathbf{m} \in \Delta^{\vee} - \overline{\mathcal{P}}$. In the first case the instanton backreaction (E.8) never exits \mathcal{P} (if $\ell_{\infty} \in \mathcal{P}$). Note also that, even though at a distance $|x - x_0| \sim M_{\rm P}^{-1}$ the saxions $\ell^i(x)$ become of order one and enter a strongly coupled region, this happens at distances well below the EFT cutoff length scale $\Lambda^{-1} \gg M_{\rm P}^{-1}$. Hence, the solution remains weakly coupled and trustable in the region in which the EFT makes sense. This motivates us to indentify the instanton lattice subcone

$$\mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}} \equiv \{ \mathbf{m} \in \mathcal{C}_{\mathrm{I}} \cap \overline{\mathcal{P}} \} \quad \subset \quad \mathcal{C}_{\mathrm{I}} \,. \tag{E.9}$$

In the explicit heterotic example discussed in section 4.2, C_{I}^{EFT} is generated by an NS5-brane wrapping X and world-sheet instantons wrapping the nef divisor intersections h and e.

Let us now look at the case where $\mathbf{m} \in C_{\mathrm{I}} - C_{\mathrm{I}}^{\mathrm{EFT}}$. Then the solution (E.8) exits the domain \mathcal{P} as we approach x_0 from infinity. In particular, the smaller we choose the asymptotic values ℓ_i^{∞} , the larger is the radial distance $|\Delta x|_* = |x - x_0|_*$ at which the solution exits the perturbative domain \mathcal{P} . The value of $|\Delta x|_*$ is directly related to the tension of a possible string whose charge \mathbf{e} belongs to $\mathcal{P}^{\vee} - \overline{\Delta}$, or equivalently $\mathbf{e} \in \mathcal{C}_{\mathrm{S}} - \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$. Indeed, for such a charge we would have $\langle \mathbf{e}, \boldsymbol{\ell}^{\infty} \rangle > 0$ while $\langle \mathbf{m}, \mathbf{e} \rangle < 0$ since $\mathbf{m} \notin \mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}}$. This means that at the distance

$$|\Delta x|_*^2 = -\frac{\langle \mathbf{e}, \mathbf{m} \rangle}{2\pi \langle \mathbf{e}, \boldsymbol{\ell}^{\infty} \rangle} \equiv \frac{|\langle \mathbf{e}, \mathbf{m} \rangle|}{2\pi \, \mathcal{T}_{\mathbf{e}}^{\infty}}, \qquad \text{for } \mathbf{e} \notin \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}, \qquad (E.10)$$

the probe tension of the string of charge \mathbf{e} vanishes, and the dual saxions ℓ_i attain the boundary of their domain \mathcal{P} . Notice that such a string $\mathbf{e} \notin \mathcal{C}_{\mathrm{S}}^{\mathrm{EFT}}$ would precisely correspond to a non-EFT, whose RG flow is expected to receive strong non-perturbative corrections generated by BPS instantons of charges $\mathbf{m} \notin \mathcal{C}_{\mathrm{I}}^{\mathrm{EFT}}$.

To sum up, instanton solutions corresponding to $\mathbf{m} \in C_{\mathrm{I}} - C_{\mathrm{I}}^{\mathrm{EFT}}$ break down along their flow. Given the analogy with string solutions corresponding to $\mathbf{e} \in C_{\mathrm{S}} - C_{\mathrm{S}}^{\mathrm{EFT}}$, we see a direct correlation between the existence of non-EFT strings and non-EFT instantons. The latter are the ones relevant along the flows of the formers, and viceversa. This may provide an EFT criterion to distinguish between the available non-perturbative corrections, possibly refining the conjecture of [135].

²³In string models we expect \mathcal{P} to be conical at least in some approximate sense, 'around the tip' of \mathcal{P} . Instead the convexity of \mathcal{P} , while easily realised in explicit models, is not automatic and one can in fact find explicit counterexamples thereof.

F EFT string flows in toroidal orbifolds

In this appendix we check conjectures 1 and 2 for compactifications related to toroidal orbifolds, in which we have better control over the spectra of KK towers. In F.1 we consider EFT string flows on the complex structure sector of heterotic and Type I compactifications, in F.2 we analyse the F-theory model discussed in [136] and in F.3 we investigate in detail the M-theory Joyce's model introduced in section 6.5. For toroidal orbifold EFT string flows in the context of type IIA see subsection 6.4.

F.1 Heterotic/type I complex structure flows in toroidal orbifolds

Let us examine the string flows on the complex structure sector in heterotic and Type I four-dimensional compactifications. For definiteness, we first focus on the heterotic models, considering the simple case of compactification over the toroidal orbifold $X = T^6/(\mathbb{Z}_2 \times \mathbb{Z}'_2)$. We denote the internal complex coordinates $z^a = y^{2a-1} + \tau^a y^{2a}$ (a = 1, 2, 3), with $y^m \simeq y^m + 1$ and the complex structures τ^a , which are identified under the orbifold action as follows:

$$(z^1, z^2, z^3) \simeq (-z^1, -z^2, z^3) \simeq (z^1, -z^2, -z^3).$$
 (F.1)

Below we shall focus on the 'untwisted' sector of moduli space only. However, it is worth noticing that the orbifold allows for the presence of fixed points, which can be resolved resulting in the introduction of a set of twisted blow-up moduli. However, the blow-up moduli may be considered to lie in the bulk of the moduli space and thus are not expected to significantly affect the asymptotic limits of the untwisted sector.

The internal string frame metric which appears in (6.2) is

$$ds_X^2 = \sum_{a=1}^3 \frac{2s^a}{\mathrm{Im}\,\tau^a} dz^a d\bar{z}^{\bar{a}} = \sum_{a=1}^3 \frac{2s^a}{\mathrm{Im}\,\tau^a} \left[(dy^{2a-1} + \mathrm{Re}\,\tau^a \,dy^{2a})^2 + (\mathrm{Im}\,\tau^a)^2 (dy^{2a})^2 \right] \quad (F.2)$$

where we have introduced the Kähler moduli s^a . The six radii (R_1, \ldots, R_6) (measured in string units) can be easily expressed in terms of the Kähler and complex structure moduli $(s^a, \operatorname{Im}\tau^a)$ as

$$R_{2a-1} = \frac{1}{2\pi} \sqrt{\frac{2s^a}{\mathrm{Im}\,\tau^a}} \quad , \quad R_{2a} = \frac{|\tau^a|}{2\pi} \sqrt{\frac{2s^a}{\mathrm{Im}\,\tau^a}} \tag{F.3}$$

The holomorphic three-form and the Kähler form are

$$\Omega = \mathrm{d}z^1 \wedge \mathrm{d}z^2 \wedge \mathrm{d}z^3 \,, \qquad J = s^a \omega_a \tag{F.4}$$

where ω_a integrally quantised untwisted 2-forms are given by

$$\omega_a = \mathrm{d}y^{2a-1} \wedge \mathrm{d}y^{2a} = \frac{\mathrm{i}}{2\mathrm{Im}\,\tau^a}\,\mathrm{d}z^a \wedge \mathrm{d}\bar{z}^a \tag{F.5}$$

In comparison with the more general models examined in section 6.1, here the only non-vanishing intersection number is $\kappa_{123} = 1$.

Let us now pass to examine the four-dimensional $\mathcal{N} = 1$ chiral spectrum. A first set of chiral coordinates is given by the complex structure moduli τ^a , with the identification $\tau^a \simeq \tau^a + 1$. The Kähler moduli s^a introduced in (F.4) combine with the B_2 -axions delivering another set of chiral fields $t^a = a^a + is^a$. Finally, the ten-dimensional dilaton ϕ enters the definition of an additional 4D chiral field, the universal modulus $t^0 = a^0 + is^0$, with

$$s^{0} = e^{-2\phi}V = e^{-2\phi}s^{1}s^{2}s^{3}.$$
 (F.6)

The Kähler potential is given by the sum of (4.23) and (6.1) and takes the form

$$K = -\log s^0 - \sum_{a=1}^3 \log s^a - \sum_{a=1}^3 \log \operatorname{Im} \tau^a \,. \tag{F.7}$$

The dual saxions $\ell_i = (\ell_0, \ell_a, \lambda_a)$, corresponding to the saxions $s^i = (s^0, s^a, \operatorname{Im} \tau^a)$ respectively, can be straightforwardly computed from the definition (2.17):

$$\ell_0 = \frac{1}{2s^0}$$
 , $\ell_a = \frac{1}{2s^a}$, $\lambda_a = \frac{1}{2\text{Im}\,\tau^a}$ (F.8)

Let us then distinguish the corresponding string charges as $\mathbf{e} = (\hat{e}, \vec{e}, \vec{\epsilon}) = (\hat{e}, e_a, \epsilon_a).$

The flow of the saxion s^0 induced by an F1 string was already analyzed in section 6.1. Instead, the flow which drives the saxions s^a towards large values is induced by NS5 strings and can be easily inferred from the general discussion of section 6.1. Indeed, being κ_{123} the only non-vanishing intersection number, the toroidal orbifold case corresponds to the Case 3 of table 3: the string tension scales in the same manner as both the M-theory KK scale and the tension of the lightest membranes.

However, the toroidal case offers the opportunity to examine more explicitly the flow of the complex structure moduli τ^a . The strings inducing such flows are identified with 10D KK monopoles wrapping some internal divisors and have charges $\mathbf{e} = (0, \vec{0}, \vec{\epsilon})$. The corresponding flow (4.9) becomes

$$\operatorname{Im} \tau^a = \operatorname{Im} \tau_0^a + \epsilon^a \sigma \tag{F.9}$$

and drives $\operatorname{Im} \tau^a$ to large values when $\sigma \to \infty$. Thus, along these flows, the string tensions scale as

$$\mathcal{T}_{\rm str} = M_{\rm P}^2 \epsilon^a \lambda_a \sim \frac{M_{\rm P}^2}{\sigma} \to 0.$$
 (F.10)

First, let us consider an elementary flow, say generated by $\vec{\epsilon} = (1, 0, 0)$. (More general flows with $\vec{\epsilon}$ of the form $(\epsilon^1, 0, 0)$, $(0, \epsilon^2, 0)$ and $(0, 0, \epsilon^3)$ lead to identical conclusions.) In this case only τ^1 flows, driving $\text{Im } \tau^1 \to \infty$. Recalling (F.3) we see that $R_1 \sim (\text{Im } \tau)^{-\frac{1}{2}}$ and $R_2 \sim (\text{Im } \tau)^{\frac{1}{2}}$. Hence, the tower scale m_* can be identified with either the lightest KK mode along R_2 or the lightest winding mode along R_1 . The resulting behaviour is qualitatively the same. For instance, by using the KK scale we get:

$$m_*^2 = m_{\rm KK}^2 = \frac{(2\pi)^2 e^{2A}}{R_2^2 l_{\rm s}^2} \sim \frac{M_{\rm P}^2}{\sigma}$$
 (F.11)

where we have employed (6.3). The lightest membranes correspond to NS5-branes wrapping the y^1 -direction and a two-cycle along (y^3, \ldots, y^6) . The corresponding scale (5.8) falls off as

$$E_{\rm mem}^2 \simeq M_{\rm P}^2 e^K \sim \frac{M_{\rm P}^2}{\sigma} \,. \tag{F.12}$$

	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{6}}$
$\mathcal{T}_{\mathrm{F1}}^{1/2}$	KK monopole	
m_*	$m_{ m KK}$	
$E_{\rm mem}$	NS5	
$\mathcal{T}_{ ext{mem}}^{1/3}$		NS5

Table 9. Asymptotic mass scalings along CS flow $\operatorname{Im} \tau^1 \sim \sigma \to \infty$.

	$\sigma^{-\frac{3}{2}}$	$\sigma^{-\frac{1}{2}}$
$\mathcal{T}_{\mathrm{F1}}^{1/2}$		KK monopole
m_*		$m_{ m KK}$
$E_{\rm mem}$	NS5	
$\mathcal{T}_{ m mem}^{1/3}$		NS5

Table 10. Asymptotic mass scalings along CS flow $\text{Im} \tau^a \sim \sigma \to \infty, \forall a$.

Thus, the tension of an elementary string scales in the same way as both m_*^2 and the membrane scale E_{mem}^2 . Then, (5.1) and (5.9) are realised with w = n = 1. These results are collected in table 9.

It is also instructive to consider more general CS flows, along which more saxions Im τ^a are sent to large values. It is easy to realise that the asymptotic behaviour of m_* is always as in (F.11), and then the scaling weight is w = 1 for all these flows. Instead, the scaling of the lightest relevant membrane scales changes. Consider for instance a flows generated by $\epsilon^a > 0$, $\forall a$. The relevant scalings are summarised in table 10, and imply that (5.9) is realised with n = 3.

One can similarly work out the case of flows associated with only two non-vanishing charges ϵ^a , e.g. $\vec{\epsilon} = (1, 1, 0)$. They all realise (5.9) with n = 2.

These findings also hold for the complex structure sector of the Type I compactification over the toroidal orbifold $X = T^6/(\mathbb{Z}_2 \times \mathbb{Z}'_2)$. In fact, as outlined in section 6.2, the complex structure moduli of Type I EFTs are described by the same Kähler potential as those of heterotic EFTs. Therefore, the relations among the mass scales for complex structure flows is the same between the two models.

F.2 F-theory Kähler flows on the resolved base $T^6/(\mathbb{Z}_2)^3$

Here we discuss the F-theory model discussed in [136]. Its base space X can be identified with one of the possible resolutions of $T^6/(\mathbb{Z}_2)^3$, which is the orientifold projection of a $T^6/(\mathbb{Z}_2 \times \mathbb{Z}_2)$ orbifold and can be identified with $\mathbb{P}^1 \times \mathbb{P}^1 \times \mathbb{P}^1$. The elliptic fiber degenerates to a I_0^* Kodaira singularity along 12 planes, each representing one O7 plus 4 D7 branes. These intersect along $4 \times 4 \times 3 = 48$ lines at which the Weierstrass model become non-minimal, which in turn intersect at 64 points, representing O3-planes. The singular lines can be resolved in different ways by blowing up 48 exceptional divisors $E_{1\alpha,2\beta}, E_{2\beta,3\gamma}, E_{3\gamma,1\alpha}$, with $\alpha, \beta, \gamma = 1, \ldots, 4$. The different possible resolutions of the base are related by flips. As described in detail in [136], these flips of X can be regarded as orientifold projections of flops of the Calabi-Yau double cover \hat{X} , connecting the different possible resolutions of the orbifold $T^6/(\mathbb{Z}_2 \times \mathbb{Z}_2)$.

As in [136], one can choose as basis D^a , a = 1, ..., 51, of $H_4(X, \mathbb{Z})$ the effective divisors $(R_1, R_2, R_3, E_{1\alpha,2\beta}, E_{2\beta,3\gamma}, E_{3\gamma,1\alpha})$, where R_1, R_2, R_3 correspond to $3 \mathbb{P}^1 \times \mathbb{P}^1$ movable divisors of the base. Note that the divisors R_1, R_2, R_3 are 'non-elementary' effective divisors. For instance, we can decompose R_1 in the following linearly equivalent way

$$R_1 \simeq D_{1\alpha} + \sum_{\beta} E_{1\alpha,2\beta} + \sum_{\gamma} E_{3\gamma,1\alpha} \qquad \forall \alpha = 1 \dots, 4, \qquad (F.13)$$

where $D_{1\alpha}$ are 4 rigid effective divisors wrapped by 4 stacks of 7-branes, each supporting a pure SO(8) SYM sector. By cyclically permuting the indices of (F.13) one gets similar relations for R_2 and R_3 , in terms of effective divisors $D_{2\beta}$ and $D_{3\gamma}$. One can then take $D_{i\alpha}$ and $E_{i\alpha,j\beta}$ as generators of the (non-simplicial) cone Eff¹(X). If in (6.20) we use the above basis of divisors D^a to identify the set of saxions $s^a = \{s^i, s^{i\alpha,j\beta}\}$, the saxionic cone is defined by

$$\Delta = \begin{cases} s^{1\alpha,2\beta} > 0 & (\text{for any } \alpha,\beta) & \text{and } (1,2,3) \text{ cyclic permutations,} \\ s^1 - \sum_\beta s^{1\alpha,2\beta} - \sum_\gamma s^{3\gamma,1\alpha} > 0 & (\text{for any } \alpha) & \text{and } (1,2,3) \text{ cyclic permutations.} \end{cases}$$
(F.14)

In order to identify the EFT strings, we pick the following dual basis of curves $C_a = (C_i, C_{i\alpha,j\beta})$:

$$C_1 = R_2 R_3$$
, $C_{1\alpha,2\beta} = -E_{1\alpha,2\beta} R_3$ and $(1,2,3)$ cyclic permutations (F.15)

such that $C_a \cdot D^b = \delta^b_a$. Note that $C_{i\alpha,j\beta}$ are not effective. Then, a BPS string corresponding to the charges

$$\mathbf{e} = e^{a} C_{a} = e^{i} C_{i} + e^{1\alpha,2\beta} C_{1\alpha,2\beta} + e^{2\beta,3\gamma} C_{2\beta} + e^{3\gamma,1\alpha} C_{3\gamma,1\alpha} \in \mathrm{Eff}_{1}(X)_{\mathbb{Z}}$$
(F.16)

is an EFT string if **e** belongs to

$$\mathcal{C}_{\rm S}^{\rm EFT} = \begin{cases} e^{1\alpha,2\beta} \ge 0 \quad (\text{for any } \alpha,\beta) \quad \text{and } (1,2,3) \text{ cyclic permutations,} \\ e^1 - \sum_{\beta} e^{1\alpha,2\beta} - \sum_{\gamma} e^{3\gamma,1\alpha} \ge 0 \quad (\text{for any } \alpha) \quad \text{and } (1,2,3) \text{ cyclic permutations.} \end{cases}$$
(F.17)

Note that these comments, and in particular $C_{\rm S}^{\rm EFT}$, are independent of the chosen resolution.

The simplest possible choice of EFT string is given by $e^{i\alpha,j\beta} = 0$, so that $e^i \ge 0$. According to (6.25), these EFT strings correspond to D3-branes wrapping the movable curve $\mathbf{e} = e^i C_i$. In this case the corresponding flow $s^i = s_0^i + e^i \sigma$, $s^{i\alpha,j\beta} \equiv s_0^{i\alpha,j\beta}$ describes a rescaling of the movabile divisors R_i , while the volume of the exceptional divisors $E_{i\alpha,j\beta}$ remains constant. The Kähler form can be expanded in the following way (6.19):

$$J = \sum_{i} v_i[R_i] + \sum_{\alpha,\beta} v_{1\alpha,2\beta}[E_{1\alpha,2\beta}] + \sum_{\beta,\gamma} v_{2\beta,3\gamma}[E_{2\beta,3\gamma}] + \sum_{\gamma,\alpha} v_{3\gamma,1\alpha}[E_{3\gamma,1\alpha}], \qquad (F.18)$$

but intersection numbers and volumes of the exceptional divisors depend on the Kähler cone chamber of $\mathcal{K}(X)_{\text{ext}}$. For simplicity, we restrict to the chamber corresponding to the 'symmetric' resolution [136] of all singularities, in which the intersection numbers are given by

$$\mathcal{I} = R_1 R_2 R_3 + \left(-\sum_{\alpha\beta} E_{1\alpha,2\beta}^2 R_3 - \ldots \right) + \left(2\sum_{\alpha\beta} E_{1\alpha,2\beta}^3 + \ldots \right) + \left[-\frac{1}{2} \sum_{\alpha\beta\gamma} E_{1\alpha,2\beta} (E_{2\beta,3\gamma}^2 + E_{3\gamma,1\alpha}^2) - \ldots \right] + \frac{1}{2} \sum_{\alpha\beta\gamma} E_{1\alpha,2\beta} E_{2\beta,3\gamma} E_{3\gamma,1\alpha}$$
(F.19)

where ... are (1, 2, 3) cyclically permuted terms and the half-integral values is due to the presence of 64 O3-planes. The Kähler cone is defined by the following conditions [136]: $v_i + \sum_{\alpha} v_{i\alpha,j\beta} > 0$, $v_{1\alpha,2\beta} - v_{2\beta,3\gamma} - v_{3\gamma,1\alpha} > 0$ and (1, 2, 3) cyclic permutations, from which we can also extract the generators of $\text{Eff}_1(X)$. Note that the these conditions imply that $v_{i\alpha,j\beta} < 0$, which is compatible with the fact that the curves $C_{i\alpha,j\beta}$ are *anti*-effective. Given the structure of the symmetric resolution, it is consistent to further restrict to the subcone of $\mathcal{K}(X)$ in which $v_{i\alpha,j\beta} \equiv v_{i,j}$ and $s^{i\alpha,j\beta} \equiv s^{i,j}$ for any $\alpha, \beta = 1, \ldots, 4$. The relation between the saxions and the Kähler moduli then reduces to

$$s^{1} = v_{2}v_{3} - 8v_{2,3}^{2},$$

$$s^{2,3} = -v_{1}v_{2,3} + v_{2,3}^{2} - v_{1,2}^{2} - v_{3,1}^{2} - 2v_{2,3}(v_{1,2} + v_{3,1}) + 2v_{1,2}v_{3,1}$$
(F.20)

and (1, 2, 3) cyclic permutations.

We first consider a flow generated by a set of positive charges $e^1, e^2, e^3 > 0$, say $\mathbf{e} = [C_1] + [C_2] + [C_3]$. In this case all the saxions associated to the movable divisors R_i flow as $s^i = s_0^i + \sigma$ with $\forall i$, while the remaining saxions are kept fixed to constant values, $s^{i,j} \equiv s_0^{i,j}$. Asymptotically as $\sigma \to \infty$, it can be seen that along the flow the volumes v_i of the curves C_i scale as $v_i \sim \sqrt{\sigma}$, while the volumes of the curves $C_{i\alpha,j\beta}$ fall off as $|v_{i,j}| \sim \frac{1}{\sqrt{\sigma}}$. Thus, the internal volume scales as $V_X \sim \sigma^{\frac{3}{2}}$. Recalling (6.18), we now can estimate the KK-scale as $m_*^2 \sim \frac{e^{2A}}{l_s^2 R_*^2} \sim \frac{M_P^2}{V_X R_*^2}$, where R_* denotes the Einstein frame radius which grows faster along the flow, namely $R_*^2 \sim v_i \sim \sqrt{\sigma}$. Thus, $m_*^2 \sim M_P^2 \sigma^{-2}$ and from (6.23) one can explicitly check that the tension $\mathcal{T}_{\mathbf{e}} = M_P^2 e^i \ell_i$ of the string generating the flow scales as $\mathcal{T}_{\mathbf{e}} \sim M_P^2 \sigma^{-1}$, as expected. Therefore, Conjecture 2 is realised with scaling weight w = 2.

Consider now the flow corresponding to one vanishing $e^i = 0$, say $\mathbf{e} = [C_1] + [C_2]$. Only the saxions s^1 and s^2 flow as $s^i = s_0^i + \sigma$ for i = 1, 2, while the remaining ones are fixed $s^3 \equiv s_0^3$, $s^{i,j} \equiv s_0^{i,j}$. Only the volume of the curve C_3 is driven to large value, as $v_3 \sim \sigma$, while the volume $v_{1,2}$ falls off as $v_{1,2} \sim \sigma^{-1}$; the volumes of all the other curves are not affected by the flow and we can assume them to be fixed at some finite values. Since the full internal volume scales as $V_X \sim \sigma$ and the radius with maximal growth scales as $R_*^2 \sim \sigma$, we can estimate $m_*^2 \sim M_{\rm P}^2 \sigma^{-2}$ and thus, as above, Conjecture 2 holds with scaling weight w = 2.

Finally, consider an elementary charge, say $\mathbf{e} = [C_1]$. In this case, solely the saxion s^1 flows as $s^1 = s_0^1 + \sigma$, with all the other saxions are fixed at constant values. Asymptotically

along the flow, the volumes of the curves C_2 and C_3 grow as $v_2 \sim v_3 \sim \sqrt{\sigma}$, while the volumes of C_1 , $C_{1,2}$, $C_{3,1}$ shrink as $v_1 \sim v_{1,2} \sim v_{3,1} \sim \frac{1}{\sqrt{\sigma}}$. This implies that the volume scales as $V_X \sim \sqrt{\sigma}$ and the maximal radius as $R_*^2 \sim \sqrt{\sigma}$. Hence $m_*^2 \sim M_{\rm P}^2 \sigma^{-1}$ and we see that Conjecture 2 is now satisfied with scaling weight w = 1.

In the case of some non-vanishing charges $e^{i\alpha,j\beta} > 0$, the volumes of the corresponding exceptional divisors $E_{i\alpha,j\beta}$ grow along the flow. However, from (F.17) it follows that also $e^i, e^j > 0$ and then the volume of the moving divisors R_i, R_j should also increase along the flow. Hence, we expect the above conclusions to be qualitatively valid in these cases as well, but it would be interesting to explicitly test our expectations in all these cases as well as in other models along the same lines.

Let us now investigate the mass scalings along the flows that involve the complex structure moduli. The model exhibits only three complex structure moduli which are packaged within three chiral fields τ^i , i = 1, 2, 3. The identification $\tau^i \simeq \tau^i + 1$ singles out their real parts $\hat{a}^i \equiv \text{Re} \tau^i$ as the axions, while the imaginary parts $\hat{s}^i \equiv \text{Im} \tau^i$ are their saxionic partners. These determines the size of the internal (Einstein frame) radii as

$$R^{2i-1} \simeq \frac{1}{\sqrt{\hat{s}^i}}, \qquad R^{2i} \simeq \sqrt{\hat{s}^i}, \qquad (F.21)$$

up to irrelevant constant, for fixed Kähler moduli and in the limit of negligible blown-up modes.

The complex structure moduli space is then described by the Kähler potential

$$K = -\log \hat{s}^1 \hat{s}^2 \hat{s}^3 \,. \tag{F.22}$$

Therefore, the dual saxions (2.17) are

$$\hat{\ell}_i = \frac{1}{2\hat{s}^i} \,. \tag{F.23}$$

The strings which drive the complex structure flows display the tension

$$\mathcal{T}_{\hat{\mathbf{e}}} = M_{\mathrm{P}}^2 \hat{e}^i \hat{\ell}_i \,. \tag{F.24}$$

Here we have collected the charges as $\hat{\mathbf{e}} = (\hat{e}^1, \hat{e}^2, \hat{e}^3)$.

Consider a string with elementary charge $\hat{\mathbf{e}} = (1, 0, 0)$ (the other elementary flow cases $\hat{\mathbf{e}} = (0, 1, 0)$ and $\hat{\mathbf{e}} = (0, 0, 1)$ can be treated analogously). The string induces a flow that drives \hat{s}^1 to the boundary of the moduli space as $\hat{s}^1 \sim \sigma \to \infty$, while keeping the remaining saxions fixed $\hat{s}^2 = \hat{s}_0^2$, $\hat{s}^3 = \hat{s}_0^3$. The tower scale is set by the lightest KK modes. Such a scale can be estimated by noticing that, along the flow, the radius, among (F.21), that diverge more rapidly is R^2 as $(R^2)^2 \equiv R_*^2 \sim \hat{s}^1 \sim \sigma$. Thus, employing (6.18), we get $m_*^2 \sim \frac{M_P^2}{R_*^2} \sim M_P \sigma^{-1}$. Alternatively, the tower scale is equivalently set by the winding modes, owing to a shrinking internal radius $(R_1)^2 \equiv R_*^2 \sim \frac{1}{\hat{s}^1} \sim \sigma^{-1}$. Thus we recognise that Conjecture 2 here holds with the scaling weight w = 1.

Flows induced by non-elementary string charges $\hat{\mathbf{e}} = (\hat{e}^1, \hat{e}^2, \hat{e}^3)$, with $\hat{e}^i > 0$, also realise Conjecture 2 with the scaling weight w = 1. As above, the tower of states can

be identified with either the KK or winding modes. In fact, along any of these flow it is possible to single out an internal radius among (F.21) which grows as $R_*^2 \sim \sigma$ determining the lightest KK modes, and another which shrinks as $R_*^2 \sim \sigma^{-1}$, which instead determines the lightest winding modes. In both cases the tower scale falls off as $m_*^2 \sim M_{\rm P}^2 \sigma^{-1}$. Being this the same behaviour as the string tension, Conjecture 2 is verified with w = 1.

F.3 M-theory flows on the toroidal orbifold $T^7/(\mathbb{Z}_2)^3$

In this section we delve into the details of the analysis of string flows in M-theory toroidal orbifold models. We consider Joyce's models [119, 123, 124], which are introduced in section 6.5. Specifically, the compactification manifold with G_2 -holonomy is a toroidal orbifold T^7/Γ , and we here choose the orbifold Γ to be $\Gamma = \mathbb{Z}_2 \times \mathbb{Z}_2 \times \mathbb{Z}_2$. As in section 6.5, we parametrise the toroidal coordinates with y_a , $a = 1, \ldots, 7$, with the identification $y_a \simeq y_a + 1$, and we denote the radii of the torii with R_a . The generators of the orbifold $\mathbb{Z}_2 \times \mathbb{Z}_2 \times \mathbb{Z}_2$ act on the coordinates y_a as in (6.83).

The model is characterised by a set of seven saxions s^a , defined by

$$s^a = \int_{C^a} \Phi \,. \tag{F.25}$$

Here Φ is the associative three-form (6.82) that is invariant under the orbifold action (6.83) and C^a is a basis of three-cycles of $H_3(T^7/\Gamma, \mathbb{Z})$. For example, the cycle C^1 can be chosen to span the directions (y_1, y_2, y_3) that are preserved under (6.83), C^2 spans (y_1, y_4, y_5) , and similarly with the remaining. Explicitly, plugging (6.82) in (F.25), we find the following relation among the saxions s^a and the radii R_a :

$$s^{1} = R_{1}R_{2}R_{3}, \qquad s^{2} = R_{1}R_{4}R_{5}, \quad s^{3} = R_{1}R_{6}R_{7} \qquad s^{4} = R_{2}R_{4}R_{6}, s^{5} = R_{2}R_{5}R_{7}, \qquad s^{6} = R_{3}R_{4}R_{7}, \quad s^{7} = R_{3}R_{5}R_{6},$$
(F.26)

which can be inverted as

$$R_{1} = \frac{(s^{1}s^{2}s^{3})^{\frac{1}{3}}}{(s^{4}s^{5}s^{6}s^{7})^{\frac{1}{6}}}, \quad R_{2} = \frac{(s^{1}s^{4}s^{5})^{\frac{1}{3}}}{(s^{2}s^{3}s^{6}s^{7})^{\frac{1}{6}}}, \quad R_{3} = \frac{(s^{1}s^{6}s^{7})^{\frac{1}{3}}}{(s^{2}s^{3}s^{4}s^{5})^{\frac{1}{6}}},$$
$$R_{4} = \frac{(s^{2}s^{4}s^{6})^{\frac{1}{3}}}{(s^{1}s^{3}s^{5}s^{7})^{\frac{1}{6}}}, \quad R_{5} = \frac{(s^{2}s^{5}s^{7})^{\frac{1}{3}}}{(s^{1}s^{3}s^{4}s^{6})^{\frac{1}{6}}}, \quad R_{6} = \frac{(s^{3}s^{4}s^{7})^{\frac{1}{3}}}{(s^{1}s^{2}s^{5}s^{6})^{\frac{1}{6}}},$$
$$R_{7} = \frac{(s^{3}s^{5}s^{6})^{\frac{1}{3}}}{(s^{1}s^{2}s^{4}s^{7})^{\frac{1}{6}}}.$$

Thus, the internal volume is expressed in terms of the saxions as follows:

$$V_X = R_1 R_2 \dots R_7 = (s^1 s^2 \dots s^7)^{\frac{1}{3}}.$$
 (F.28)

The Kähler potential acquires the simple form

$$K = -\log(s^1 s^2 \dots s^7), \qquad (F.29)$$

and the dual saxions (2.17) are

$$\ell_a = \frac{1}{2s^a} \,. \tag{F.30}$$

In 4D, strings originate from M5 branes which wrap cycles in $H_4(T^7/\Gamma, \mathbb{Z})$. By choosing a basis of four-cycles \tilde{C}_a dual to C^a , the tension of a string that stem from an M5 brane wrapped over the cycle $\tilde{C} = e^a \tilde{C}_a$ can be computed as follows

$$\mathcal{T}_{\mathbf{e}} = \frac{2\pi}{l_M^4} e^{2A} \int_{\tilde{C}} *\Phi = M_{\mathrm{P}}^2 e_a \ell^a \,. \tag{F.31}$$

In the last step we have employed (6.75) and then re-expressed the tension in terms of the dual saxions (F.30). We will also collect $\mathbf{e} = (e_a)$.

In the following, in order to investigate the validity of Conjecture 2 and compute the scaling weight w for various string flows, we need to specify the scale m_* at which the 4D EFT breaks down and that has to be related to the string tension (F.31). The natural candidates are the lightest among KK, whose masses are set by

$$m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{R_*^2 V_X} \,,$$
 (F.32)

up to irrelevant constants. Here R_* generically denotes any of the radii R_a . The EFT breaking scale m_* is then set by any of the scales (F.32) which displays the fastest fall off along the given string flow.

For completeness, in the flows examined below we will also relate the mass scale (5.8) set by the lightest membranes. In 4D, membranes can originate from either M2 branes spanning three external spacetime directions or M5 branes wrapped on linear combinations of three-cycles C^a . In either case, the scale (5.8) is

$$E_{\rm mem} \sim e^{\frac{K}{2}} M_{\rm P} = \frac{M_{\rm P}}{\sqrt{s^1 s^2 \dots s^7}} \,.$$
 (F.33)

Let us now examine some relevant string flows and how Conjecture 2 is realised.

 $\mathbf{e} = (1, 0, 0, 0, 0, 0, 0)$ Elementary flows are generated by strings with a single non-vanishing, positive charge e^a . Such a string can be obtained from an M5 brane wrapped over any of the four-cycles \tilde{C}_a . As an example, we may consider the flow generated by the string with charge $e^1 > 0$, while $e^a = 0 \quad \forall a \neq 1$. The flow generated by this string drives only the saxion s^1 to infinite distance, that is $s^1 \sim e^1 \sigma \to \infty$. The string generating the flow becomes tensionless with a fall off $\mathcal{T}_{\text{str}} \sim M_P^2 \sigma^{-1}$. As can be easily seen from (F.27), along such a flow $R_1, R_2, R_3 \sim \sigma^{\frac{1}{3}}$, while $R_4, R_5, R_6, R_7 \sim \sigma^{-\frac{1}{6}}$. Thus, the lightest modes among those in (F.32) are the KK modes associated to $R_* = R_1, R_2, R_3$. Their masses scale as

$$m_*^2 = m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{\sigma} \,. \tag{F.34}$$

Therefore, Conjecture 2 is realised with the scaling weight w = 1. It can be easily checked that the scale (F.33) set by lightest membranes displays an analogous scaling along this elementary flow. These findings are summarised in table 11.

	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{6}}$
$\mathcal{T}_{ m str}^{1/2}$	M5	
m_*	$m_{ m KK}$	
$E_{\rm mem}$	M2, M5	
$\mathcal{T}_{ m mem}^{1/3}$		M2, M5

Table 11. Mass scalings along the flow generated by an elementary string obtained from an M5 wrapped over the four-cycle \tilde{C}_a .

	σ^{-1}	$\sigma^{-\frac{1}{2}}$	$\sigma^{-\frac{1}{3}}$
$\mathcal{T}_{ m str}^{1/2}$		M5	
m_*	$m_{ m KK}$		
$E_{\rm mem}$	M2, M5		
$\mathcal{T}_{ m mem}^{1/3}$			M2, M5

Table 12. Mass scalings along the flows generated by the string with charges (1, 1, 0, 0, 0, 0, 0).

	$\sigma^{-\frac{3}{2}}$	$\sigma^{-\frac{1}{2}}$
$\mathcal{T}_{ m str}^{1/2}$		M5
m_*	$m_{ m KK}$	
$E_{\rm mem}$	M2, M5	
$\mathcal{T}_{ m mem}^{1/3}$		M2, M5

Table 13. Mass scalings along the flows generated by the string with charges (1, 1, 1, 0, 0, 0, 0).

 $\mathbf{e} = (1, 1, 0, 0, 0, 0, 0)$ A non-elementary string with only non-null charges $e^1 = e^2 = 1$ induces the flow $s^1 \sim s^2 \sim \sigma \to \infty$, while the other saxions remain fixed. The tower scale m_* is set by lightest KK modes, associated to $R_* = R_1$ in (F.32). These scale as

$$m_*^2 = m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{\sigma^2},$$
 (F.35)

realising Conjecture 2 with scaling weight w = 2. These results are collected in table 12.

 $\mathbf{e} = (1, 1, 1, 0, 0, 0, 0)$ Consider a non-elementary string with only non-null charges $e^1 = e^2 = e^3 = 1$. This string induces a flow that drives $s^1 \sim s^2 \sim s^3 \to \infty$. The tower scale m_* is here identified with the KK modes associated to $R_* = R_1$ and their masses fall off as

$$m_*^2 = m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{\sigma^3}$$
. (F.36)

Thus, Conjecture 2 is realised with scaling weight w = 3. The relevant scales are summarised in the table 13.

	$\sigma^{-\frac{3}{2}}$	σ^{-1}	$\sigma^{-\frac{1}{2}}$
$\mathcal{T}_{ m str}^{1/2}$			M5
m_*		$m_{\rm KK}$	
$E_{\rm mem}$	M2, M5		
$\mathcal{T}_{ m mem}^{1/3}$			M2, M5

Table 14. Mass scalings along the flows generated by the string with charges (1, 1, 0, 1, 0, 0, 0).

	$\sigma^{-\frac{7}{2}}$	$\sigma^{-\frac{3}{2}}$	$\sigma^{-\frac{7}{6}}$	$\sigma^{-\frac{1}{2}}$
$\mathcal{T}_{ m str}^{1/2}$				M5
m_*		$m_{\rm KK}$		
$E_{\rm mem}$	M2			
$\mathcal{T}_{ m mem}^{1/3}$			M2	

Table 15. Mass scalings along the flows generated by the string with charges (1, 1, 0, 1, 0, 0, 0).

 $\mathbf{e} = (1, 1, 0, 1, 0, 0, 0)$ Consider non-elementary strings with only non-null charges $e^1 = e^2 = e^4 = 1$. Since $s^1 \sim s^2 \sim s^4 \sim \sigma \to \infty$, the tower scale m_* can be either fixed by the KK scale (for $R_* = R_1, R_2, R_4$). These scale as

$$m_*^2 = m_{\rm KK}^2.$$
 (F.37)

Conjecture 2 is realised with scaling weight w = 2. The behavior of the relevant scales along such a flow is summarised in the table 14.

 $\mathbf{e} = (1, 1, 1, 1, 1, 1)$ As last example, consider non-elementary strings with all non-null charges. For example, choose $e^a = 1 \forall a$, resulting in a flow that drives all the saxions to infinite distance $s^a \sim \sigma \to \infty$. The tower scale m_* is fixed by the KK scale (for R_* given by any of the radii in (F.27)), which falls off as

$$m_*^2 = m_{\rm KK}^2 \simeq \frac{M_{\rm P}^2}{\sigma^3}$$
. (F.38)

Thus, as in the latter two cases above, Conjecture 2 is realised with scaling weight w = 3. However, in this case, the membrane cutoff scale $\mathcal{T}_{\text{mem}}^{1/3}$ falls off faster than the string scale $\mathcal{T}_{\text{str}}^{1/2}$:

$$\mathcal{T}_{\rm mem}^{1/3} \sim \frac{M_{\rm P}}{\sigma^{\frac{7}{6}}} \sim M_{\rm P} \left(\frac{\mathcal{T}_{\rm str}^{1/2}}{M_{\rm P}}\right)^{\frac{1}{3}}$$
 (F.39)

Indeed, from the Type IIA perspective, this flow corresponds to a large string coupling limit $g_s \gg 1$ and one needs to employ the full M-theory description. The relevant scales for such a flow are summarised in table 15.

The analysis above can be similarly carried for other non-elementary charges, leading to the classification of flows as in table 7.

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- H. Ooguri and C. Vafa, On the Geometry of the String Landscape and the Swampland, Nucl. Phys. B 766 (2007) 21 [hep-th/0605264] [INSPIRE].
- [2] C. Vafa, The string landscape and the swampland, hep-th/0509212 [INSPIRE].
- [3] T.D. Brennan, F. Carta and C. Vafa, The String Landscape, the Swampland, and the Missing Corner, PoS TASI2017 (2017) 015 [arXiv:1711.00864] [INSPIRE].
- [4] E. Palti, The Swampland: Introduction and Review, Fortsch. Phys. 67 (2019) 1900037
 [arXiv:1903.06239] [INSPIRE].
- [5] M. van Beest, J. Calderón-Infante, D. Mirfendereski and I. Valenzuela, Lectures on the Swampland Program in String Compactifications, arXiv:2102.01111 [INSPIRE].
- [6] T.W. Grimm, E. Palti and I. Valenzuela, Infinite Distances in Field Space and Massless Towers of States, JHEP 08 (2018) 143 [arXiv:1802.08264] [INSPIRE].
- [7] S.-J. Lee, W. Lerche and T. Weigand, Tensionless Strings and the Weak Gravity Conjecture, JHEP 10 (2018) 164 [arXiv:1808.05958] [INSPIRE].
- [8] T.W. Grimm, C. Li and E. Palti, Infinite Distance Networks in Field Space and Charge Orbits, JHEP 03 (2019) 016 [arXiv:1811.02571] [INSPIRE].
- [9] P. Corvilain, T.W. Grimm and I. Valenzuela, The Swampland Distance Conjecture for Kähler moduli, JHEP 08 (2019) 075 [arXiv:1812.07548] [INSPIRE].
- [10] A. Joshi and A. Klemm, Swampland Distance Conjecture for One-Parameter Calabi-Yau Threefolds, JHEP 08 (2019) 086 [arXiv:1903.00596] [INSPIRE].
- F. Marchesano and M. Wiesner, Instantons and infinite distances, JHEP 08 (2019) 088 [arXiv:1904.04848] [INSPIRE].
- [12] A. Font, A. Herráez and L.E. Ibáñez, The Swampland Distance Conjecture and Towers of Tensionless Branes, JHEP 08 (2019) 044 [arXiv:1904.05379] [INSPIRE].
- [13] T.W. Grimm and D. Van De Heisteeg, Infinite Distances and the Axion Weak Gravity Conjecture, JHEP 03 (2020) 020 [arXiv:1905.00901] [INSPIRE].
- [14] S.-J. Lee, W. Lerche and T. Weigand, Emergent Strings, Duality and Weak Coupling Limits for Two-Form Fields, arXiv:1904.06344 [INSPIRE].
- [15] S.-J. Lee, W. Lerche and T. Weigand, *Emergent Strings from Infinite Distance Limits*, arXiv:1910.01135 [INSPIRE].
- [16] F. Baume, F. Marchesano and M. Wiesner, Instanton Corrections and Emergent Strings, JHEP 04 (2020) 174 [arXiv:1912.02218] [INSPIRE].
- [17] N. Gendler and I. Valenzuela, Merging the weak gravity and distance conjectures using BPS extremal black holes, JHEP 01 (2021) 176 [arXiv:2004.10768] [INSPIRE].
- [18] F. Xu, On TCS G₂ manifolds and 4D emergent strings, JHEP **10** (2020) 045 [arXiv:2006.02350] [INSPIRE].

- [19] T.W. Grimm, Moduli Space Holography and the Finiteness of Flux Vacua, arXiv:2010.15838 [INSPIRE].
- [20] D. Klaewer, S.-J. Lee, T. Weigand and M. Wiesner, Quantum corrections in 4d N = 1 infinite distance limits and the weak gravity conjecture, JHEP 03 (2021) 252
 [arXiv:2011.00024] [INSPIRE].
- [21] F. Baume and J. Calderón Infante, Tackling the SDC in AdS with CFTs, JHEP 08 (2021)
 057 [arXiv:2011.03583] [INSPIRE].
- [22] B. Bastian, T.W. Grimm and D. van de Heisteeg, Weak gravity bounds in asymptotic string compactifications, JHEP 06 (2021) 162 [arXiv:2011.08854] [INSPIRE].
- [23] E. Perlmutter, L. Rastelli, C. Vafa and I. Valenzuela, A CFT Distance Conjecture, arXiv:2011.10040 [INSPIRE].
- [24] J. Calderón-Infante, A.M. Uranga and I. Valenzuela, The Convex Hull Swampland Distance Conjecture and Bounds on Non-geodesics, JHEP 03 (2021) 299 [arXiv:2012.00034]
 [INSPIRE].
- [25] D. Klaewer and E. Palti, Super-Planckian Spatial Field Variations and Quantum Gravity, JHEP 01 (2017) 088 [arXiv:1610.00010] [INSPIRE].
- [26] G. Buratti, J. Calderón and A.M. Uranga, Transplanckian axion monodromy!?, JHEP 05 (2019) 176 [arXiv:1812.05016] [INSPIRE].
- [27] P. Draper and S. Farkas, Gravitational Instabilities and Censorship of Large Scalar Field Excursions, JHEP 05 (2019) 158 [arXiv:1901.00515] [INSPIRE].
- [28] H. Geng, S. Grieninger and A. Karch, Entropy, Entanglement and Swampland Bounds in DS/dS, JHEP 06 (2019) 105 [arXiv:1904.02170] [INSPIRE].
- [29] P. Draper and S. Farkas, Transplanckian Censorship and the Local Swampland Distance Conjecture, JHEP 01 (2020) 133 [arXiv:1910.04804] [INSPIRE].
- [30] H. Geng, Distance Conjecture and De-Sitter Quantum Gravity, Phys. Lett. B 803 (2020) 135327 [arXiv:1910.03594] [INSPIRE].
- [31] Q. Bonnefoy, L. Ciambelli, D. Lüst and S. Lüst, Infinite Black Hole Entropies at Infinite Distances and Tower of States, Nucl. Phys. B 958 (2020) 115112 [arXiv:1912.07453] [INSPIRE].
- [32] S. Lanza, F. Marchesano, L. Martucci and I. Valenzuela, Swampland Conjectures for Strings and Membranes, JHEP 02 (2021) 006 [arXiv:2006.15154] [INSPIRE].
- [33] B.R. Greene, A.D. Shapere, C. Vafa and S.-T. Yau, Stringy Cosmic Strings and Noncompact Calabi-Yau Manifolds, Nucl. Phys. B 337 (1990) 1 [INSPIRE].
- [34] C. Vafa, Evidence for F-theory, Nucl. Phys. B 469 (1996) 403 [hep-th/9602022] [INSPIRE].
- [35] T. Banks and L.J. Dixon, Constraints on String Vacua with Space-Time Supersymmetry, Nucl. Phys. B 307 (1988) 93 [INSPIRE].
- [36] T. Banks and N. Seiberg, Symmetries and Strings in Field Theory and Gravity, Phys. Rev. D 83 (2011) 084019 [arXiv:1011.5120] [INSPIRE].
- [37] M. Reece, Photon Masses in the Landscape and the Swampland, JHEP 07 (2019) 181
 [arXiv:1808.09966] [INSPIRE].
- [38] B. Michel, E. Mintun, J. Polchinski, A. Puhm and P. Saad, *Remarks on brane and antibrane dynamics*, JHEP 09 (2015) 021 [arXiv:1412.5702] [INSPIRE].

- [39] J. Polchinski, Brane/antibrane dynamics and KKLT stability, arXiv:1509.05710 [INSPIRE].
- [40] S. Katz, H.-C. Kim, H.-C. Tarazi and C. Vafa, Swampland Constraints on 5d $\mathcal{N} = 1$ Supergravity, JHEP 07 (2020) 080 [arXiv:2004.14401] [INSPIRE].
- [41] E. Cremmer, S. Ferrara, L. Girardello and A. Van Proeyen, Yang-Mills Theories with Local Supersymmetry: Lagrangian, Transformation Laws and SuperHiggs Effect, Nucl. Phys. B 212 (1983) 413 [INSPIRE].
- [42] S. Lanza, F. Marchesano, L. Martucci and D. Sorokin, How many fluxes fit in an EFT?, JHEP 10 (2019) 110 [arXiv:1907.11256] [INSPIRE].
- [43] S. Lanza, Exploring the Landscape of effective field theories, Ph.D. thesis, Padua University (2019), arXiv:1912.08935 [INSPIRE].
- [44] A. Hebecker, P. Henkenjohann and L.T. Witkowski, What is the Magnetic Weak Gravity Conjecture for Axions?, Fortsch. Phys. 65 (2017) 1700011 [arXiv:1701.06553] [INSPIRE].
- [45] B. Heidenreich, J. McNamara, M. Montero, M. Reece, T. Rudelius and I. Valenzuela, Chern-Weil Global Symmetries and How Quantum Gravity Avoids Them, arXiv:2012.00009 [INSPIRE].
- [46] G. Dvali, Three-form gauging of axion symmetries and gravity, hep-th/0507215 [INSPIRE].
- [47] N. Kaloper and L. Sorbo, A Natural Framework for Chaotic Inflation, Phys. Rev. Lett. 102 (2009) 121301 [arXiv:0811.1989] [INSPIRE].
- [48] F. Marchesano, G. Shiu and A.M. Uranga, F-term Axion Monodromy Inflation, JHEP 09 (2014) 184 [arXiv:1404.3040] [INSPIRE].
- [49] T.W. Grimm, C. Li and I. Valenzuela, Asymptotic Flux Compactifications and the Swampland, JHEP 06 (2020) 009 [Erratum ibid. 01 (2021) 007] [arXiv:1910.09549]
 [INSPIRE].
- [50] M.J. Dolan, P. Draper, J. Kozaczuk and H. Patel, Transplanckian Censorship and Global Cosmic Strings, JHEP 04 (2017) 133 [arXiv:1701.05572] [INSPIRE].
- [51] N. Arkani-Hamed, L. Motl, A. Nicolis and C. Vafa, The string landscape, black holes and gravity as the weakest force, JHEP 06 (2007) 060 [hep-th/0601001] [INSPIRE].
- [52] A. Buonanno and T. Damour, Effective action and tension renormalization for cosmic and fundamental strings, Phys. Lett. B 432 (1998) 51 [hep-th/9803025] [INSPIRE].
- [53] J. Polchinski, Monopoles, duality, and string theory, Int. J. Mod. Phys. A 19S1 (2004) 145 [hep-th/0304042] [INSPIRE].
- [54] D. Harlow, Wormholes, Emergent Gauge Fields, and the Weak Gravity Conjecture, JHEP 01 (2016) 122 [arXiv:1510.07911] [INSPIRE].
- [55] B. Heidenreich, M. Reece and T. Rudelius, The Weak Gravity Conjecture and Emergence from an Ultraviolet Cutoff, Eur. Phys. J. C 78 (2018) 337 [arXiv:1712.01868] [INSPIRE].
- [56] B. Heidenreich, M. Reece and T. Rudelius, Emergence of Weak Coupling at Large Distance in Quantum Gravity, Phys. Rev. Lett. 121 (2018) 051601 [arXiv:1802.08698] [INSPIRE].
- [57] P. Candelas, A. Font, S.H. Katz and D.R. Morrison, Mirror symmetry for two parameter models. 2, Nucl. Phys. B 429 (1994) 626 [hep-th/9403187] [INSPIRE].
- [58] S. Boucksom, J.-P. Demailly, M. Păun and T. Peternell, The pseudo-effective cone of a compact Kähler manifold and varieties of negative Kodaira dimension, J. Algebraic Geom. 22 (2013) 201.

- [59] P. Hořava and E. Witten, Heterotic and type-I string dynamics from eleven-dimensions, Nucl. Phys. B 460 (1996) 506 [hep-th/9510209] [INSPIRE].
- [60] P. Hořava and E. Witten, Eleven-dimensional supergravity on a manifold with boundary, Nucl. Phys. B 475 (1996) 94 [hep-th/9603142] [INSPIRE].
- [61] P. Wilson, The kähler cone on calabi-yau threefolds, Invent. Math. 107 (1992) 561.
- [62] D.R. Morrison, Compactifications of moduli spaces inspired by mirror symmetry, in Journées de géométrie algébrique d'Orsay — Juillet 1992, no. 218 in Astérisque, pp. 243–271, Société mathématique de France, (1993).
- [63] A. Grassi and D.R.O. Morrison, Automorphisms and the kähler cone of certain calabi-yau manifolds, Duke Math. J. 71 (1992) 831.
- [64] D.R. Morrison, Beyond the Kahler cone, in Proceedings of the Hirzebruch 65 conference on algebraic geometry, Ramat Gan 1993, (1994), pp. 361–376.
- [65] Y. Kawamata, On the cone of divisors of Calabi-Yau fiber spaces, Int. J. Math. 8.5 (1997) 665 [alg-geom/9701006].
- [66] B. Skauli, The cone conjecture for some calabi-yau varieties, MSc Thesis, University of Oslo, Norway (2017).
- [67] C.R. Brodie, A. Constantin, A. Lukas and F. Ruehle, Swampland conjectures and infinite flop chains, Phys. Rev. D 104 (2021) 046008 [arXiv:2104.03325] [INSPIRE].
- [68] E. Looijenga, New compactifications of locally symmetric varieties, in Proceedings of the 1984 Vancouver Conference in Algebraic Geometry, American Mathematical Society, (1986), pp. 341–364.
- [69] T.W. Grimm, F. Ruehle and D. van de Heisteeg, Classifying Calabi-Yau Threefolds Using Infinite Distance Limits, Commun. Math. Phys. 382 (2021) 239 [arXiv:1910.02963] [INSPIRE].
- [70] N. Arkani-Hamed, S. Dimopoulos and S. Kachru, Predictive landscapes and new physics at a TeV, hep-th/0501082 [INSPIRE].
- [71] J. Distler and U. Varadarajan, Random polynomials and the friendly landscape, hep-th/0507090 [INSPIRE].
- [72] S. Dimopoulos, S. Kachru, J. McGreevy and J.G. Wacker, N-flation, JCAP 08 (2008) 003 [hep-th/0507205] [INSPIRE].
- [73] G. Dvali, Black Holes and Large N Species Solution to the Hierarchy Problem, Fortsch. Phys. 58 (2010) 528 [arXiv:0706.2050] [INSPIRE].
- [74] G. Dvali and M. Redi, Black Hole Bound on the Number of Species and Quantum Gravity at LHC, Phys. Rev. D 77 (2008) 045027 [arXiv:0710.4344] [INSPIRE].
- [75] T. Trenner and P.M.H. Wilson, Asymptotic curvature of moduli spaces for calabi-yau threefolds, arXiv:0902.4611.
- [76] S.-J. Lee, W. Lerche and T. Weigand, A Stringy Test of the Scalar Weak Gravity Conjecture, Nucl. Phys. B 938 (2019) 321 [arXiv:1810.05169] [INSPIRE].
- [77] B. Heidenreich, J. McNamara, M. Montero, M. Reece, T. Rudelius and I. Valenzuela, Non-invertible Global Symmetries and Completeness of the Spetrum, to appear.
- [78] J. McNamara and C. Vafa, Cobordism Classes and the Swampland, arXiv:1909.10355 [INSPIRE].

- [79] M. Dierigl and J.J. Heckman, Swampland cobordism conjecture and non-Abelian duality groups, Phys. Rev. D 103 (2021) 066006 [arXiv:2012.00013] [INSPIRE].
- [80] F. Denef, Les Houches Lectures on Constructing String Vacua, Les Houches 87 (2008) 483 [arXiv:0803.1194] [INSPIRE].
- [81] K. Oguiso, On algebraic fiber space structures on a calabi-yau 3-fold, with an appendix by noboru nakayama, Int. J. Math. 4 (1993) 439.
- [82] P.M.H. Wilson, The existence of elliptic fibre space structures on calabi-yau threefolds, II, Math. Proc. Cambridge Phil. Soc. 123 (1998) 259.
- [83] A. Strominger, S.-T. Yau and E. Zaslow, Mirror symmetry is T duality, Nucl. Phys. B 479 (1996) 243 [hep-th/9606040] [INSPIRE].
- [84] T.W. Grimm, The N = 1 effective action of F-theory compactifications, Nucl. Phys. B 845 (2011) 48 [arXiv:1008.4133] [INSPIRE].
- [85] S.B. Giddings, S. Kachru and J. Polchinski, *Hierarchies from fluxes in string compactifications*, *Phys. Rev. D* 66 (2002) 106006 [hep-th/0105097] [INSPIRE].
- [86] G. Shiu, G. Torroba, B. Underwood and M.R. Douglas, Dynamics of Warped Flux Compactifications, JHEP 06 (2008) 024 [arXiv:0803.3068] [INSPIRE].
- [87] A.R. Frey, G. Torroba, B. Underwood and M.R. Douglas, The Universal Kähler Modulus in Warped Compactifications, JHEP 01 (2009) 036 [arXiv:0810.5768] [INSPIRE].
- [88] L. Martucci, On moduli and effective theory of N = 1 warped flux compactifications, JHEP 05 (2009) 027 [arXiv:0902.4031] [INSPIRE].
- [89] L. Martucci, Warping the Kähler potential of F-theory/ IIB flux compactifications, JHEP 03 (2015) 067 [arXiv:1411.2623] [INSPIRE].
- [90] L. Martucci, Warped Kähler potentials and fluxes, JHEP 01 (2017) 056 [arXiv:1610.02403] [INSPIRE].
- [91] S. Boucksom, J.-P. Demailly, M. Paun and T. Peternell, The pseudo-effective cone of a compact kähler manifold and varieties of negative kodaira dimension, J. Alg. Geom. 22 (2013) 201.
- [92] J. Xiao, Positivity in Kähler geometry, Theses, Université de Fudan Shanghai, Chine, May, (2016).
- [93] J. Fu and J. Xiao, Relations between the kähler cone and the balanced cone of a kähler manifold, Adv. Math. 263 (2014) 230.
- [94] E. Witten, Phase transitions in M-theory and F-theory, Nucl. Phys. B 471 (1996) 195 [hep-th/9603150] [INSPIRE].
- [95] P. Mayr, Mirror symmetry, N = 1 superpotentials and tensionless strings on Calabi-Yau four folds, Nucl. Phys. B 494 (1997) 489 [hep-th/9610162] [INSPIRE].
- [96] Y. Hu and S. Keel, Mori dream spaces and GIT, Michigan Math. J. 48 (2000) 331.
- [97] C. Birkar, P. Cascini, C.D. Hacon and J. McKernan, Existence of minimal models for varieties of log general type, J. Am. Math. Soc. 23 (2010) 405.
- [98] B. Lehmann and J. Xiao, Convexity and zariski decomposition structure, Geom. Funct. Anal. 26 (2016) 1135.
- [99] Y. Kawamata, Crepart blowing-up of 3-dimensional canonical singularities and its application to degenerations of surfaces, Annals Math. 127 (1988) 93.

- [100] S.-J. Lee, W. Lerche and T. Weigand, Modular Fluxes, Elliptic Genera, and Weak Gravity Conjectures in Four Dimensions, JHEP 08 (2019) 104 [arXiv:1901.08065] [INSPIRE].
- [101] A. Collinucci, F. Denef and M. Esole, D-brane Deconstructions in IIB Orientifolds, JHEP 02 (2009) 005 [arXiv:0805.1573] [INSPIRE].
- [102] T.W. Grimm and J. Louis, The effective action of type IIA Calabi-Yau orientifolds, Nucl. Phys. B 718 (2005) 153 [hep-th/0412277] [INSPIRE].
- [103] N.J. Hitchin, The Geometry of Three-Forms in Six Dimensions, J. Diff. Geom. 55 (2000) 547 [math/0010054] [INSPIRE].
- [104] B.S. Acharya, M. Aganagic, K. Hori and C. Vafa, Orientifolds, mirror symmetry and superpotentials, hep-th/0202208 [INSPIRE].
- [105] I. Garcia-Etxebarria, F. Marchesano and A.M. Uranga, Non-perturbative F-terms across lines of BPS stability, JHEP 07 (2008) 028 [arXiv:0805.0713] [INSPIRE].
- [106] J. de Boer, F. Denef, S. El-Showk, I. Messamah and D. Van den Bleeken, *Black hole bound* states in $AdS_3 \times S^2$, *JHEP* **11** (2008) 050 [arXiv:0802.2257] [INSPIRE].
- [107] M. Cvetič, G. Shiu and A.M. Uranga, Chiral four-dimensional N = 1 supersymmetric type 2A orientifolds from intersecting D6 branes, Nucl. Phys. B 615 (2001) 3 [hep-th/0107166]
 [INSPIRE].
- [108] R. Blumenhagen, M. Cvetič, P. Langacker and G. Shiu, Toward realistic intersecting D-brane models, Ann. Rev. Nucl. Part. Sci. 55 (2005) 71 [hep-th/0502005] [INSPIRE].
- [109] R. Blumenhagen, B. Körs, D. Lüst and S. Stieberger, Four-dimensional String Compactifications with D-branes, Orientifolds and Fluxes, Phys. Rept. 445 (2007) 1 [hep-th/0610327] [INSPIRE].
- [110] F. Marchesano, Progress in D-brane model building, Fortsch. Phys. 55 (2007) 491 [hep-th/0702094] [INSPIRE].
- [111] A. Font, L.E. Ibáñez and F. Marchesano, Coisotropic D8-branes and model-building, JHEP 09 (2006) 080 [hep-th/0607219] [INSPIRE].
- [112] D. Cremades, L.E. Ibáñez and F. Marchesano, SUSY quivers, intersecting branes and the modest hierarchy problem, JHEP 07 (2002) 009 [hep-th/0201205] [INSPIRE].
- [113] G. Lawlor, The angle criterion, Invent. Math. 95 (1989) 437.
- [114] M.R. Douglas, Topics in D geometry, Class. Quant. Grav. 17 (2000) 1057
 [hep-th/9910170] [INSPIRE].
- [115] C. Beasley and E. Witten, A note on fluxes and superpotentials in M-theory compactifications on manifolds of G₂ holonomy, JHEP 07 (2002) 046 [hep-th/0203061]
 [INSPIRE].
- [116] B.S. Acharya and S. Gukov, M theory and singularities of exceptional holonomy manifolds, Phys. Rept. 392 (2004) 121 [hep-th/0409191] [INSPIRE].
- [117] J. Halverson and D.R. Morrison, On gauge enhancement and singular limits in G₂ compactifications of M-theory, JHEP 04 (2016) 100 [arXiv:1507.05965] [INSPIRE].
- [118] G.W. Moore and E. Witten, Selfduality, Ramond-Ramond fields, and k-theory, JHEP 05 (2000) 032 [hep-th/9912279] [INSPIRE].
- [119] D.D. Joyce, Compact riemannian 7-manifolds with holonomy g_2 . I, J. Diff. Geom. 43 (1996) 291.

- [120] R. Bryand and S. Salamon, On the construction of some complete metrices with expectional holonomy, Duke Math. J. 58 (1989) 829 [INSPIRE].
- [121] G.W. Gibbons, D.N. Page and C.N. Pope, Einstein Metrics on S³ R³ and R⁴ Bundles, Commun. Math. Phys. 127 (1990) 529 [INSPIRE].
- [122] M. Atiyah and E. Witten, M theory dynamics on a manifold of G₂ holonomy, Adv. Theor. Math. Phys. 6 (2003) 1 [hep-th/0107177] [INSPIRE].
- [123] D.D. Joyce, Compact riemannian 7-manifolds with holonomy g₂. II, J. Diff. Geom. 43 (1996) 329.
- [124] D.D. Joyce, Compact manifolds with special holonomy, Oxford University Press on Demand, (2000).
- [125] A. Lukas and S. Morris, Moduli Kähler potential for M-theory on a G₂ manifold, Phys. Rev. D 69 (2004) 066003 [hep-th/0305078] [INSPIRE].
- [126] A.B. Barrett and A. Lukas, Classification and Moduli Kähler potentials of G₂ manifolds, Phys. Rev. D 71 (2005) 046004 [hep-th/0411071] [INSPIRE].
- [127] W.D. Goldberger and M.B. Wise, Renormalization group flows for brane couplings, Phys. Rev. D 65 (2002) 025011 [hep-th/0104170] [INSPIRE].
- [128] T.W. Grimm, J. Monnee and D. Van De Heisteeg, Bulk Reconstruction in Moduli Space Holography, arXiv:2103.12746 [INSPIRE].
- [129] R. Lazarsfeld, Positivity in Algebraic Geometry I, Springer-Verlag Berlin Heidelberg, (2004), [DOI].
- [130] A. Dabholkar, G.W. Gibbons, J.A. Harvey and F. Ruiz Ruiz, Superstrings and Solitons, Nucl. Phys. B 340 (1990) 33 [INSPIRE].
- [131] S. Weinberg, Gravitation and Cosmology: Principles and Applications of the General Theory of Relativity, John Wiley and Sons, New York, U.S.A. (1972).
- [132] P.A. Dirac, Classical theory of radiating electrons, Proc. Roy. Soc. Lond. A 167 (1938) 148.
- [133] U. Cannella and R. Sturani, Classical energy momentum tensor renormalisation via effective field theory methods, Gen. Rel. Grav. 42 (2010) 2491 [arXiv:0808.4034] [INSPIRE].
- [134] G.W. Gibbons, M.B. Green and M.J. Perry, Instantons and seven-branes in type IIB superstring theory, Phys. Lett. B 370 (1996) 37 [hep-th/9511080] [INSPIRE].
- [135] E. Palti, C. Vafa and T. Weigand, Supersymmetric Protection and the Swampland, JHEP 06 (2020) 168 [arXiv:2003.10452] [INSPIRE].
- [136] F. Denef, M.R. Douglas, B. Florea, A. Grassi and S. Kachru, Fixing all moduli in a simple F-theory compactification, Adv. Theor. Math. Phys. 9 (2005) 861 [hep-th/0503124]
 [INSPIRE].